A SINGLE TRANSISTOR UNITY POWER FACTOR RECTIFIER

A THESIS SUBMITTED TO THE GRADUATE SCHOOL OF NATURAL AND APPLIED SCIENCES OF MIDDLE EAST TECHNICAL UNIVERSITY

BY

MURAT TUNÇ

IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF MASTER OF SCIENCE IN ELECTRICAL AND ELECTRONICS ENGINEERING

JANUARY 2007

Approval of the Graduate School of Natural and Applied Sciences

Prof. Dr. Canan ÖZGEN Director

I certify that this thesis satisfies all the requirements as a thesis for the degree of Master of Science.

Prof. Dr. İsmet ERKMEN Head of the Department

This is to certify that we have read this thesis and that in our opinion it is fully adequate, in scope and quality, as a thesis for the degree of Master of Science.

Prof. Dr. Yıldırım ÜÇTUĞ Co-Supervisor Prof. Dr. H. Bülent ERTAN Supervisor

Examining Committee Members

Prof. Dr. Muammer ERMİŞ (M.E.T.U. EE)	
Prof. Dr. H. Bülent ERTAN (M.E.T.U. EE)	
Prof. Dr. Yıldırım ÜÇTUĞ (M.E.T.U. EE)	
Prof. Dr. Kemal LEBLEBİCİOĞLU (M.E.T.U. EE)	
Doç. Dr. M. Timur AYDEMİR (GAZI UNV. EE)	

I hereby declare that all information in this document has been obtained and presented in accordance with academic rules and ethical conduct. I also declare that, as required by these rules and conduct, I have fully cited and referenced all material and results that are not original to this work.

Name, Last name : Murat Tunç

Signature :

ABSTRACT

A NEW SINGLE TRANSISTOR UNITY POWER FACTOR RECTIFIER

Tunç, Murat

M.Sc., Department of Electrical and Electronics Engineering Supervisor : Prof. Dr. Bülent ERTAN Co-Supervisor: Prof. Dr. Yıldırım ÜÇTUĞ

January 2007, 257 pages

This thesis analyzes unity power factor rectifiers since this type of rectifiers use energy as efficient as possible.

Throughout the thesis, some unity power factor rectifier topologies are investigated and some of them selected to investigate in detail. Afterwards, a new single transistor unity power factor rectifier topology is proposed, simulated, implemented and compared with one of the selected unity power factor rectifier topology on the basis of efficiency, total harmonic distortion, input current ripple and output voltage ripple.

Keywords: rectifier, unity power factor, single transistor unity power factor rectifier

ÖΖ

YENİ TEK TRANSİSTÖRLÜ BİRİMSEL GÜÇ FAKTÖRLÜ DOĞRULTMAÇ

Tunç, Murat Yüksek Lisans, Elektrik ve Elektronik Mühendisliği Bölümü Danışman : Prof. Dr. Yıldırım ÜÇTUĞ Yardımcı Danışman: Prof. Dr. Bülent ERTAN

Ocak 2007, 257 sayfa

Bu tez birimsel güç faktörlü doğrultmaçları, bu tip doğrultmaçlar elektriksel enerjiyi oldukça efektif kullandıkları için incelemektedir.

Tez boyunca bazı birimsel güç faktörlü doğrultmaçlar incelenmiştir ve bunlardan bazıları daha detaylı incelenmek için seçilmiştir. Daha sonra yeni bir birimsel güç faktörlü doğrultmaç devresi önerilmiştir, simüle edilmiştir, gerçekleştirilmiştir ve daha önce seçilen bir birimsel güç faktörlü doğrultmaç devresi ile verimlilik, toplam harmonik bozulma, giriş akımı dalgalanması ve çıkış voltajı dalgalanması baz alınarak kıyaslanmıştır.

Anahtar kelimeler : doğrultmaç, birimsel güç faktörü , tek transistörlü birimsel güç faktörlü doğrultmaç

To My Parents

ACKNOWLEDGMENTS

I wish to express my deepest gratitude to my supervisor Prof. Dr. Bülent Ertan and co-supervisor Prof. Dr. Yıldırım Üçtuğ for their valuable advices and guidance and encouragements.

I would like to thanks Prof. Dr. Işık Çadırcı for her guidance about one rectifier topology.

I offer sincere thanks to Tolga Çamlıkaya, Erdal Bizkevelci for their patience to my questions.

I thanks to my friends (Tolga İnan, Serkan Şedele, Cüneyt Karacan, Levent Burak Yalçiner, Çağlar Özyurt, Akın Acar, Arif Yılmaz, Barış Kolacan)

Finally, I would like to thanks to my family for their support and guidance through all stages of my life.

TABLE OF CONTENTS

ABSTRACT	iv
ÖZ	v
ACKNOWLEDGMENTS	vii
TABLE OF CONTENTS	viii
LIST OF TABLES	xiii
LIST OF FIGURES	xv
CHAPTER	
1. INTRODUCTION	1
1.1 Introduction	1
1.2 Outline	2
2. UNITY POWER FACTOR RECTIFIERS	4
2.1 Introduction	4
2.2 Single Phase Unity Power Factor Rectifiers	5
2.2.1 Single Phase Half-Bridge Boost Topology	5
2.2.2 Front-End Rectifier Followed By A Boost Converter	9
2.2.3 Low Conduction Loss AC/DC UPF Rectifier	12
2.3 Three Phase Unity Power Factor Rectifiers	14
2.3.1 Single Transistor Three-Phase Multiresonant ZCS HPF Rectifier	15
2.3.2 12-Pulse Voltage Source Rectifier With PWM-SHE	19
2.3.3 Single Controlled Switch Three Phase Rectifier With Unity Power Fa	ctor
and Sinusoidal Input Current	23
3. SELECTION OF UPF RECTIFIER TOPOLOGY	27
3.1 Introduction	27
3.2 Calculations Of Single Phase Unity Power Factor Rectifiers	28
3.2.1 Calculations Of Unity Power Factor Converter Using Half-Bridge Bo	ost

Topology	28
3.2.2 Calculations Of Front-End Rectifier Followed By A Boost Converter	31
3.2.3 Calculations Of Low Conduction Loss AC/DC UPF Rectifier	32
3.3 Calculations Of Three Phase Unity Power Factor Rectifiers	34
3.3.1 Calculations Of Single Transistor Three-Phase Multiresonant ZCS HPF	
Rectifier	34
3.3.2 Calculations Of 12-Pulse Voltage Source Rectifier With PWM-SHE	39
3.3.3 Calculations Of Single Controlled Switch Three Phase Rectifier With	
Unity Power Factor And Sinusoidal Input Current	41
4. PROPOSED SINGLE PHASE SINGLE SWITCHING TRANSISTOR UPF	
RECTIFIER TOPOLOGY	45
4.1 Introduction	45
4.2 Proposed Single Phase Single Switching Transistor UPF Configuration	46
4.2.1 Comparison Of The Proposed Rectifier With Front-End Rectifier Follow	ed
By A Boost Converter And Low Conduction Loss AC/DC UPF Rectifier	49
4.2.2 Analysis Of The Proposed UPF Rectifier Analytically	55
4.3 Result	57
5. CONTROL TECHNIQUES FOR UNITY POWER FACTOR RECTIFIERS	59
5.1 Introduction	59
5.2 Popular PFC Control Techniques	60
5.2.1 Peak Current Control	63
5.2.2 Average Current Control	64
5.2.3 Hysteresis Control	66
5.2.4 Borderline Control	67
5.2.5 Discontinuous Current PWM Control	68
6. SIMULATIONS OF PROPOSED CONVERTER AND CLASSICAL UNITY	
POWER FACTOR RECTIFIER	70
6.1 Introduction	70
6.2 Choice Of The Simulation Software And The Simulation Models	72
6.3 Component Selection	79
6.4 Proposed and Classical Unity Power Factor Rectifiers' Comparison Via	
Simulations	83
6.4.1 Classical And Proposed UPF Rectifier Simulation With 3 kW Purely	

Resistive Load And Their Performance Comparison	
6.4.1.1 Classical UPF Rectifier Simulation With 3 kW Purely Rest	istive Load
	89
6.4.1.2 Proposed Converter's Simulation With 3 kW Purely Resist	ive Load 91
6.4.1.3 Evaluation Of Simulation Results Of Classical And Propos	sed UPF
Rectifier With 3 kW Purely Resistive Load	
6.4.2 Classical UPF Rectifier And Proposed Rectifier's Simulation V	Vith 3 kW
Inductive-Resistive Load And Their Performance Comparison	
6.4.2.1 Classical Unity Power Factor Rectifier's Simulation With I	nductive-
Resistive Load	
6.4.2.1.1 Classical Unity Power Factor Rectifier's Simulation W	ith
Inductive-Resistive Load (100uH)	
6.4.2.1.2 Classical Unity Power Factor Rectifier's Simulation W	ith
Inductive-Resistive Load (10uH)	
6.4.2.2 Proposed Converter's Simulation With Inductive-Resistive	Load 103
6.4.2.2.1 Proposed Converter's Simulation With Inductive-Resis	tive Load
(100 uH)	
6.4.2.2.2 Proposed Converter's Simulation With Inductive-Resis	tive Load
(10 uH)	
6.4.2.3 Evaluation Of Simulation Results Of Classical And Propos	sed UPF
Rectifier With 3 kW Inductive-Resistive Loads (100 uH and 10 uF	H) 107
6.4.3 Classical Unity Power Factor Rectifier's Simulation With 3 kW	/ Resistive
Load By Taking Reference Voltage From A Transformer Connected	To Mains
Supply	111
6.4.4 Classical Unity Power Factor And Proposed Rectifier's Simula	tion With 3
kW Purely Resistive Load In Dynamic Conditions And Their Comp	arison 112
6.4.4.1 Classical Unity Power Factor Rectifier's Simulation With 3	8 kW Purely
Resistive Load In Dynamic Conditions	
6.4.4.2 Proposed Rectifier's Simulation With 3 kW Purely Resistiv	ve Load In
Dynamic Conditions	114
5 Conclusion	115
HARDWARE DESIGN IMPLEMENTATION AND RESULTS	
1 Introduction	

7.2 Hardware Design And Implementation	9
7.2.1 Inductor Design	0
7.2.2 Snubber Design	1
7.2.2.1 Overvoltage Snubber Design	1
7.2.2.2 R-C Snubber Design	4
7.2.3 Current Sensing Circuit	7
7.2.4 Low Pass Filter and Mains Supply Feedback Circuit	8
7.2.5 Control Circuit	1
7.2.6 Power Supply Block14	4
7.2.7 Proposed Topology Block	5
7.3 Proposed Rectifier And Classical UPF Rectifier's Schematic Diagram 14	6
7.4 Experimental Results And Comparison With Simulations In Same Conditions	
	9
7.4.1 Proposed Rectifier's Experimental Results With 1 kW Resistive Load And	
Comparison With Simulation In Same Conditions	0
7.4.2 Proposed Rectifier's Experimental Results With 2 kW Resistive Load And	
Comparison With Simulations In Same Conditions15	6
7.5 Conclusion	1
8. CONCLUSION	4
8.1 General	4
8.2 Experimental And Simulation Results	5
8.3 Conclusion And Future Work17	0
REFERENCES	2
APPENDICES	
APPENDIX A: UC3854 Controlled Power Factor Correction Circuit Design 17	7
APPENDIX B: Datasheets Of The Components	7
APPENDIX C: IGBT And Ultrafast Diode Selection Tables	5
APPENDIX D: Classical UPF Rectifier And Proposed Rectifier's Waveforms	
Obtained Via Simulations	7
D.1 Waveforms of Classical Unity Power Factor Rectifier And Proposed Rectifier	•
With 3 kW Purely Resistive Load	7
D.1.1 Waveforms of Classical Unity Power Factor Rectifier With 3 kW Purely	
Resistive Load	7

LIST OF TABLES

TABLES

Table 3.1 Comparison of the rectifiers
Table 5.1 Current waveforms in popular current control techniques
Table 6.1 Results of classical unity power factor rectifier's simulation made by using
PSIM 6.1 with the UC3854B IC model in figure 6.1, figure 6.2 at steady state
with 3 kW purely resistive load and comparison of the simulations' speed 77
Table 6.2 Results of classical unity power factor rectifier's simulation made by using
MATLAB-Simulink R14, PSIM 6.1 and Simulink co-simulation at steady state
with 3 kW purely resistive load and comparison of the simulations' speed 78
Table 6.3 Ultrafast Diode Selection Criteria. 82
Table 6.4 Ultrafast diode selection table
Table 6.5 Bridge diode selection table. 83
Table 6.6 Performance comparison table between proposed rectifier and classical
unity power factor rectifier with 3 kW purely resistive load (simulated)
Table 6.7 Proposed and classical UPF rectifier's input current harmonics' number
and magnitudes up to 50^{th} harmonic with 3 kW purely resitive load at steady
state
Table 6.8 Performance comparison table between proposed rectifier and classical
unity power factor rectifier with two 3 kW inductive-resistive (100 uH and 10
uH) load108
Table 6.9 Proposed and classical UPF rectifier's input current harmonics' number
and magnitudes up to 50^{th} harmonic with 3 kW inductive-resistive load (100 uH)
at steady state
Table 6.10 Proposed and classical UPF rectifier's input current harmonics' number
and magnitudes up to 50 th harmonic with 3 kW inductive-resistive load (10 uH)
at steady state

Table 6.11 Classical UPF rectifier performance table when its reference voltage is
taken by using a transformer
Table 6.12 Performance comparison table between proposed rectifier and classical
unity power factor rectifier with 3 kW purely resistive load
Table 6.13 Performance comparison table between proposed rectifier and classical
unity power factor rectifier with serially connected 64.5 Ω resistor and 100 uH
inductor load117
Table 7.1 Required core area product, core number, core and total inductor losses for
the 500 uH inductor by different B_{max} values
Table 7.2 Approximate power dissipation values in watts for a given temperature rise
and core number
Table 7.3 Proposed rectifier's input current harmonics up to 50 th harmonic measured
by using HIOKI 3194 Motor/Harmonic Hi Tester, 997th and 999th harmonic
obtained by DSO6014A scope and the input current harmonics up to 50 th
harmonic and the harmonics had higher magnitudes near the 50 kHz switching
frequency obtained by PSIM 6.1 simulation program with 1 kW load at steady
state
Table 7.4 Proposed rectifier's input current harmonics up to 50 th harmonic measured
by using HIOKI 3194 Motor/Harmonic Hi Tester, 999th harmonic obtained by
DSO6014A scope and its harmonics up to 50 th harmonic, harmonics that have
higher magnitudes near at switching speed obtained by using PSIM 6.1 with 2
kW load at steady state
Table 7.5 Simulation and implemented rectifier circuit results with different loads.
Table C.1 IGBT selection table
Table C.2 Ultrafast diode selection table. 226

LIST OF FIGURES

FIGURES

Figure 2.1 Half-bridge boost PFC circuit
Figure 2.2 Current waveforms in HCC technique, (a) Over one line cycle, (b) Over
one switching cycle7
Figure 2.3 The two modes of operation of the half-bridge circuit, (a) Mode 1:
positive inductor-current-slope mode, (b) Mode 2: negative inductor-current-
slope mode7
Figure 2.4 Block schematic representation of a typical closed-loop system for PFC8
Figure 2.5 Closed-loop control system with imbalance control
Figure 2.6 Front-end rectifier followed by a boost converter9
Figure 2.7 Inductor current waveform
Figure 2.8 Single phase, low conduction loss AC/DC UPF rectifier with input power
factor correction (a) Power circuit (b) its controller block diagram12
Figure 2.9 (a) Three-phase multiresonant ZCS cell. (b) The input-side resonant-tank
capacitor-voltage waveform together with the input-current waveform. (c) The
enlarged waveform of V _{Cr1} during one switching period16
Figure 2.10 Basic single transistor three-phase multiresonant ZCS HPF rectifier
circuit and its ideal waveforms
Figure 2.11 The simplified single-input and single-output model of the single-
transistor three-phase multiresonant ZCS HPF rectifier
Figure 2.12 Main circuit of the voltage source converter drive system20
Figure 2.13 Phasor diagram of front-end voltage source converter
Figure 2.14 PWM-SHE switching pattern for 11 th and 13 th harmonic elimination
(modulation index M=0.9). (a) Switch gating signal. (b) Line-to-line voltage22
Figure 2.15 Topology of the converter
Figure 3.1 The normalized input characteristic of the single-transistor three-phase

multiresonant ZCS HPF rectifier. The normalized transistor off time α is
expressed in radians
Figure 3.2 Normalized voltage stress of switch S ₁
Figure 3.3 Normalized current stress of switch S1
Figure 3.4 Normalized voltage stress of output diode
Figure 4.1 Proposed single phase single switching transistor UPF rectifier
Figure 4.2 Equivalent circuit when switching transistor on and mains supply's
voltage phase is between 0 and 180 ⁰ 47
Figure 4.3 Equivalent circuit when switching transistor is off and mains supply's
voltage phase is between 0 and 180 ⁰ 47
Figure 4.4 Equivalent circuit when mains supply's voltage phase is between 180 and
360 ⁰ , switching transistor is on
Figure 4.5 Equivalent circuit when mains supply's voltage phase is between 180 and
360 ⁰ and switching transistor is off
Figure 4.6 Front-end rectifier followed by a boost converter topology
Figure 4.7 Front-end rectifier followed by a boost converter's equivalent circuit when
the switching transistor is on (a) mains supply's voltage phase is between 0 and
180° , (b) mains supply's voltage phase is between 180 and 360°
Figure 4.8 Front-end rectifier followed by a boost converter's equivalent circuit when
the switching transistor is off (a) mains supply's voltage phase is between 0 and
180° , (b) mains supply's voltage phase is between 180 and 360°
Figure 4.9 Low conduction loss AC/DC UPF rectifier topology
Figure 4.10 Low conduction loss AC/DC UPF rectifier's equivalent circuit when the
switching transistor is on (a) mains supply's voltage phase is between 0 and
180° , (b) mains supply's voltage phase is between 180 and 360°
Figure 4.11 Low conduction loss AC/DC UPF rectifier's equivalent circuit when the
switching transistor is off (a) mains supply's voltage phase is between 0 and
180° , (b) mains supply's voltage phase is between 180 and 360°
Figure 5.1 Principle scheme of a boost PFC
Figure 5.2 Current Waveforms in "Discontinuous Conduction Mode"
, "Continuous Conduction Mode", "Boundary Conduction Mode"
Figure 5.3 Peak Current Control Scheme
Figure 5.4 Average Current Control Scheme

Figure 5.5 Hysteresis Control Scheme	. 66
Figure 5.6 Borderline Control Scheme.	. 67
Figure 5.7 Discontinuous current PWM control scheme	. 68
Figure 6.1 UC3854B model created using discrete components for PSIM 6.1	
simulation program	. 73
Figure 6.2 UC3854B model in PSIM	. 74
Figure 6.3 UC3854B model drawn using discrete blocks in Simulink R14	. 75
Figure 6.4 IGBT's voltage and current waveform while input voltage is 187 V rms	.80
Figure 6.5 Ultrafast diode current and voltage waveform when V_{in} is equal to 187	V
rms.	. 80
Figure 6.6 Figure 6.5's zoomed version.	. 81
Figure 6.7 Classical unity power factor rectifier and its control circuit's block	
diagram	. 86
Figure 6.8 Proposed rectifier topology and its control block diagram	. 87
Figure 6.9 Classical unity power rectifier's input voltage and current waveform with	ith
3 kW purely resistive load at steady state.	. 89
Figure 6.10 Classical unity power rectifier's output voltage waveform with 3 kW	
purely resistive load at steady state.	. 90
Figure 6.11 Proposed rectifier's input voltage and current waveform with 3 kW	
purely resistive load at steady state.	. 92
Figure 6.12 Proposed rectifier's output voltage waveform with 3 kW purely resisti	ive
load at steady state	. 92
Figure 6.13 Proposed rectifier's efficiency graphs obtained with measurements in	
laboratory and simulation with different purely resistive loads at steady state	. 96
Figure 6.14 Proposed rectifier's input power factor graphs obtained with	
measurements in laboratory and simulation with different purely resistive load	ls
at steady state	. 97
Figure 6.15 THD graphs of proposed rectifier's input current obtained with	
measurements in laboratory and simulation with different purely resistive load	ls
at steady state	. 97
Figure 6.16 Classical unity power rectifier's input voltage and current waveform	
with 3 kW inductive-resistive load (100 uH) at steady state	100
Figure 6.17 Classical unity power rectifier's output voltage waveform with 3 kW	

inductive-resistive load (100 uH) at steady state	. 100
Figure 6.18 Classical unity power rectifier's input voltage and current waveform	l
with 3 kW inductive-resistive load (10 uH) at steady state	. 102
Figure 6.19 Classical unity power rectifier's output voltage waveform with 3 kW	1
inductive-resistive load (10 uH) at steady state	. 102
Figure 6.20 Proposed rectifier's input voltage and current waveform with 3 kW	
inductive-resistive load (100 uH) at steady state	. 104
Figure 6.21 Proposed rectifier's output voltage waveform with 3 kW inductive-	
resistive load (100 uH) at steady state	. 105
Figure 6.22 Proposed rectifier's input voltage and current waveform with 3 kW	
inductive-resistive load (10 uH) at steady state	. 106
Figure 6.23 Proposed rectifier's output voltage waveform with 3 kW inductive-	
resistive load (10 uH) at steady state	. 107
Figure 6.24 (a) Classical unity power factor rectifier's output voltage and input	
current waveform when 3 kW purely resistive load is connected, it is	
disconnected between 1 s. and 1.05 s. and connected again, (b) its zoomed	
version.	. 113
Figure 6.25 (a) Proposed rectifier's output voltage and input current waveform w	vhen
3 kW purely resistive load is connected, it is disconnected between 1 s. and 2	1.05
s. and connected again, (b) its zoomed version	. 114
Figure 7.1 Implemented proposed rectifier's block diagram.	. 119
Figure 7.2 B-H curve for Metglas amorphous metal cores in kilogauss and oerste	ed.
	. 124
Figure 7.3 C core dimensions.	. 127
Figure 7.4 (a) Overvoltage snubber, (b) its equivalent circuit during transistor tur	rn-
off	. 132
Figure 7.5 Overvoltage snubber in implemented circuit	. 133
Figure 7.6 R-C snubber connected to a transistor.	. 135
Figure 7.7 R-C snubber optimal design curves	. 135
Figure 7.8 R-C snubber circuit	. 136
Figure 7.9 (a) LEM current sensor module, (b) current sensing circuit with the	
current sensor.	. 137
Figure 7.10 Circuit of mains supply's sample block in implemented circuit	. 139

Figure 7.11 Low pass filter in proposed rectifier circuit
Figure 7.12 Low pass filter in classical unity power factor rectifier
Figure 7.13 An integrator
Figure 7.14 Error amplifier with compensator
Figure 7.15 Current error amplifier in implemented circuit
Figure 7.16 Voltage error amplifier in implemented circuit
Figure 7.17 Power supply which produces +12 V, -12 V, +18 V DC voltage 145
6.7 and 6.8
Figure 7.18 The proposed and classical unity power factor rectifier PCB schematic
not including +12 V, -12 V, +18 V supply voltages and reference voltages,
VVAC and VVF1147
Figure 7.19 Printed circuit board schematic shown +12 V, -12 V, +18 V voltage
supply and reference voltages, VVAC and VVF1 proportional to the mains
supply in rectified form
Figure 7.20 Proposed rectifier's input current and input voltage waveform with 1 kW
nearly purely resistive load
Figure 7.21 Proposed rectifier's input current and input voltage waveform at 1 kW
nearly purely resistive load
Figure 7.22 FFT of proposed rectifier's input current when 1 kW purely resistive
load is connected to the rectifier's output and 100 MHz DSO6014A scope's
scale, offset, span, center frequency and sample rate are 20 dB/division, -48 dB,
100 kHz, 50 kHz and 200 kSa/s respectively
Figure 7.23 Proposed rectifier's input current and input voltage waveform obtained
using PSIM 6.1 simulation program at steady state when 1 kW purely resistive
load is connected
Figure 7.24 Proposed rectifier's output voltage waveform obtained using PSIM 6.1
simulation program at steady state when 1 kW purely resistive load is connected.
Figure 7.25 (a) FFT waveform of proposed rectifier's input current obtained using
PSIM 6.1 simulation program between 0 and 50500 Hz at steady state when 1
kW purely resistive load is connected, (b) the FFT waveform between 47 and 55
kHz
Figure 7.26 Proposed rectifier's input current and input voltage waveform with 2 kW

nearly purely resistive load
Figure 7.27 Proposed rectifier's input current and input voltage waveform with 2 kW
nearly purely resistive load157
Figure 7.28 FFT of proposed rectifier's input current when 2 kW purely resistive
load is connected to the rectifier's output and 100 MHz DSO6014A scope's
scale, offset, span, center frequency and sample rate are 10 dB/division, -20 dB,
100 kHz, 50 kHz and 200 kSa/s respectively158
Figure 7.29 Proposed rectifier's input current and input voltage waveform obtained
using PSIM 6.1 simulation program at steady state when 2 kW purely resistive
load is connected
Figure 7.30 Proposed rectifier's output voltage waveform obtained by using PSIM
6.1 simulation program at steady state when 2 kW purely resistive load is
connected
Figure 7.31 (a) FFT waveform of proposed rectifier's input current obtained by using
PSIM 6.1 simulation program between 0 and 50500 Hz at steady state when 2
kW purely resistive load is connected, (b) the FFT waveform between 47 and 55
kHz
Figure 8.1 Proposed UPF rectifier's implemented and simulation circuits'
performance diagrams with 1 kW purely resistive load at steady state
Figure 8.2 Proposed UPF rectifier's implemented and simulation circuits'
performance diagrams with 2 kW purely resistive load at steady state
Figure 8.3 Classical and proposed UPF rectifier's performance diagrams obtained via
simulation with 3 kW purely resistive load at steady state
Figure 8.4 Classical and proposed UPF rectifier's performance diagram with 3 kW
inductive-resistive load (100 uH)
Figure D.1 Classical unity power factor rectifier's input power and power factor at
purely resistive 3 kW load, (a) its input power, (b) its power factor graph 227
Figure D.2 (a) FFT waveform of classical unity power factor rectifier's input current
at 3 kW purely resistive load at steady state, (b) its zoomed version
Figure D.3 THD of classical UPF rectifier's input current at 3 kW purely resistive
load at steady state
Figure D.4 (a) IGBT's current and voltage waveform of classical UPF rectifier with
3 kW purely resistive load at steady state, (b) their zoomed version

Figure D.5 (a) One of the bridge diodes, D1's current and voltage waveform of
classical UPF rectifier with 3 kW purely resistive load at steady state, (b) their
zoomed version
Figure D.6 (a) Boost diode, D5's current and voltage waveform of classical UPF
rectifier with 3 kW purely resistive load at steady state, (b) their zoomed version.
231
Figure D.7 Proposed rectifier's input power and power factor graph at purely
resistive 3 kW load, (a) its input power, (b) its power factor graph232
Figure D.8 (a) FFT waveform of proposed rectifier's input current at 3 kW purely
resistive load at steady state, (b) its zoomed version
Figure D.9 THD of proposed rectifier's input current at 3 kW purely resistive load at
steady state
Figure D.10 (a) IGBT's current and voltage waveform of proposed rectifier with 3
kW purely resistive load at steady state, (b) their zoomed versions
Figure D.11 (a) One of the boost diodes, D3's current and voltage waveform of
proposed UPF rectifier with 3 kW purely resistive load at steady state, (b) their
zoomed versions
Figure D.12 (a) One of the bridge diodes, D1's current and voltage waveform of
proposed UPF rectifier with 3 kW purely resistive load at steady state, (b) their
zoomed versions
Figure D.13 (a) One of the bridge diodes, D7's current and voltage waveform of
proposed UPF rectifier with 3 kW purely resistive load at steady state, (b) their
zoomed versions
Figure D.14 Classical unity power factor rectifier's input power and power factor at
inductive-resistive 3 kW load (100 uH), (a) its input power, (b) its power factor.
Figure D.15 (a) FFT waveform of classical unity power factor rectifier's input
current at 3 kW inductive-resistive load (100 uH) at steady state, (b) its zoomed
version
Figure D.16 THD of classical UPF rectifier's input current at 3 kW inductive-
resistive load (100 uH) at steady state
Figure D.17 (a) IGBT's current and voltage waveform of classical UPF rectifier with
3 kW inductive-resistive load (100 uH) at steady state, (b) their zoomed version.

Figure D.18 (a) One of the bridge diodes, D1's current and voltage waveform of
classical UPF rectifier with 3 kW inductive-resistive load (100 uH) at steady
state, (b) their zoomed version
Figure D.19 (a) Boost diode, D5's current and voltage waveform of classical UPF
rectifier with 3 kW inductive-resistive load (100 uH) at steady state, (b) their
zoomed version
Figure D.20 Classical unity power factor rectifier's input power and power factor at
inductive-resistive 3 kW load (10 uH), (a) its input power, (b) its power factor.
Figure D.21 (a) FFT waveform of classical unity power factor rectifier's input
current at 3 kW inductive-resistive load (10 uH) at steady state, (b) its zoomed
version
Figure D.22 THD of classical UPF rectifier's input current at 3 kW inductive-
resistive load (10 uH) at steady state
Figure D.23 Proposed rectifier's input power and power factor at inductive-resistive 3
kW load (100 uH), (a) its input power, (b) its power factor245
Figure D.24 (a) FFT waveform of proposed rectifier's input current at 3 kW
inductive-resistive load (100 uH) at steady state, (b) its zoomed version 246
Figure D.25 THD of proposed rectifier's input current at 3 kW inductive-resistive
load (100 uH) at steady state
Figure D.26 (a) IGBT's current and voltage waveform of proposed rectifier with 3
kW inductive-resistive load (100 uH) at steady state, (b) their zoomed versions.
Figure D.27 (a) One of the boost diodes, D3's current and voltage waveform of
proposed UPF rectifier with 3 kW inductive-resistive load (100 uH) at steady
state, (b) their zoomed versions
Figure D.28 (a) One of the bridge diodes, D1's current and voltage waveform of
proposed UPF rectifier with 3 kW inductive-resistive load (100 uH) at steady
state, (b) their zoomed versions
Figure D.29 One of the bridge diodes, D7's current and voltage waveform of
proposed UPF rectifier with 3 kW inductive-resistive load (100 uH) at steady
state

CHAPTER I

INTRODUCTION

1.1 Introduction

Electrical power consumers such as motor drives, controlled or uncontrolled rectifiers are major pollutant of the ac power distribution network. As the number of devices used in the ac distribution network increase, pollution of the ac power distribution increases due to their non-linear electrical characteristics. Detrimental effects of the distortion in ac power distribution network is known. In addition to the pollution on the distribution network, usage of electrical energy as efficiently as possible is another research area in power electronics. To decrease pollution in ac power distribution network and use electrical energy as efficient as possible, power factor correction techniques are often used.

The objective of the use of power factor correction circuits is to make input voltage and input current of a rectifier or a power consuming electrical device in phase so that the power factor of the device as seen by the ac power distribution network is unity or close to unity. These type of rectifiers are called unity power factor rectifiers.

The basic classification of the unity power factor rectifiers in this work is single phase or three phase unity power factor rectifiers on the basis of whether the rectifier is supplied from three phase or single phase mains supply. According to the relative magnitude of rectifier's input voltage and output voltage , there is the classification of buck, boost and buck-boost type of rectifier. Unity power factor rectifiers can also be classified as those using single switching transistor or more than one switching transistor.

The basic control modes of unity power factor rectifiers are current mode control and voltage mode control. With respect to the operation mode of the power stage, unity power rectifier (UPF rectifier) may be divided in to three categories: Discontinuous Conduction Mode (DCM), Critical Conduction Mode (CrCM), and Continuous Conduction Mode (CCM). CCM mode unity power factor rectifiers are preferred due to their low conducted noise, low conduction losses in the semiconductor switches and inductors, and low inductor core losses at high power applications. Peak current control, average current control, hysteresis control are generally used to control CCM operated rectifiers while the UPF rectifier used borderline control operates at the boundary between CCM and DCM.

The aim of the thesis is to investigate unity power factor rectifiers and looking for a new unity power factor rectifier topology with possible performance improvement. The application area is a device supplying 3 kW power to the load, from a dc bus of 400 and 550 volt. The unity power factor rectifier's output voltage is desired to be constant. For this purpose, first the literature is studied and some of the existing topologies are studied.

1.2 Outline

In chapter 2, unity power factor rectifiers are categorized as single phase and three phase unity power factor rectifiers. 6 UPF rectifier topologies are selected and investigated in detail.

Chapter 3 is assigned to calculations of the ratings of main semiconductors in the topologies on the basis of given specifications. Then, they are compared with each other and the findings are summarized in a table. One of them is selected for further study.

In chapter 4, a new unity power factor rectifier topology is developed and its working principle is explained. Analytical expressions for the new topology is given also in this chapter, and its performance is compared with the topology selected in chapter 3.

Control techniques for power factor correction converters are given in chapter 5. The control techniques are explained briefly in this chapter.

Chapter 6 is assigned to simulations of the selected unity power factor rectifier in chapter 3 and the new single switching transistor unity power factor rectifier topology. These rectifiers' simulation results are given in this chapter.

In chapter 7, the proposed new single switching transistor unity power factor rectifier and the rectifier selected in chapter 3 are implemented on a single printed circuit board (PCB), and the proposed rectifier's performance is measured in a laboratory by using 1 kW and 2 kW purely resistive loads. The proposed rectifier is simulated in nearly same condition with the implemented rectifier. After that, the simulation results and the measured results are compared in this chapter.

Chapter 8 is conclusion chapter and it summarizes all of the thesis briefly.

CHAPTER II

UNITY POWER FACTOR RECTIFIERS

2.1 Introduction

There are lots of study, papers and documents upon unity power factor rectifiers. To investigate UPF (Unity Power Factor) rectifiers, some documents are read and the topologies in this documents are classified in this chapter. Some of them are selected to investigate in detail.

Unity power factor rectifiers can be classified as single phase unity power factor rectifiers and three phase unity power factor rectifiers considering mains supply type; buck, boost, and buck-boost type unity power factor rectifiers according to the rectifier's relative input and output voltage magnitudes; and also UPF rectifiers used single switching power transistor or multi switching power transistor. There are so many papers about UPF rectifiers. 32 papers were investigated to find cheap and best solution. Reference 2, 18, 26 express single phase single power switch boost type rectifier. In reference 1, 3, 12, 14, 19, 20, 22, 29, 30 single phase boost type, in reference 7, 27 single phase buck type and in reference 25, 31, 32 single phase buck-boost type rectifiers are investigated. These rectifiers have switching power transistor boost type rectifier, but reference 4 buck type rectifier. Topologies in reference 7, 8, 9, 10, 11, 13, 15, 17, 21 are three phase boost type ones. Rectifiers in reference 16, 24, 28 are three phase buck type UPF rectifiers, and these use more

than one switching power transistor as well as ones in reference 5, 8, 9, 10, 11, 13, 15, 17, 21. Among these 32 papers, reference 1-6 are selected. The selected papers and the rectifier topologies in these papers are summarized in the following sections in this chapter. The most suitable and cheapest rectifier is tried to be picked out, and calculations of the rectifiers' main electrical component ratings are given in chapter 3.

2.2 Single Phase Unity Power Factor Rectifiers

In this section, the selected papers in reference [1], [2], [3] are summarized and the rectifier topologies, whose mains supply is single phase supplies in these papers are investigated. The rectifier topologies are a single phase half-bridge boost topology, well known front-end rectifier followed by a boost converter and a low conduction loss AC/DC UPF rectifier topologies.

2.2.1 Single Phase Half-Bridge Boost Topology

This topology is boost type rectifier had a half controlled full bridge configuration except the lack of two diodes reduced the conduction losses and its working and control principles are given in this section [1].

In this topology, at any time there is only one semiconductor on-state voltage drop. Hence conduction loss in this rectifier topology is very low, so it is expected that the rectifier operates at high operating efficiencies.

For proper boost operation, the rectifier's output voltage will be at least twice the value of the peak input voltage. For example, when its mains supply's voltage is 230 V rms, the rectifier's output voltage will have to be greater than 760 V. This means that high-voltage rated semiconductors and passive components are needed in this topology due to the high voltage at the output side of the rectifier. However, when its mains supply's voltage is e.g. 130 Vrms, the rectifier's output voltage will have to be greater than only 370V. This topology is good option for mains supply had nominal 110 V rms.

Figure 2.1 shows the rectifier topology. It is essential that voltage across the capacitor C1 and C2 greater than the rectifier's instantaneous input voltage for the control action to be effective throughout the line cycle. In order to make its input current and input voltage in phase, fixed-band hysteresis current control technique can be used.



Figure 2.1 Half-bridge boost PFC circuit.

In the fixed-band hysteresis current control technique, by appropriately switching the transistors S1 and S2, the input current i_L is tried to be kept in a Δi current band about the current reference $i_{L,ref}$ figure 2.3 (a) and (b) show the two modes of the circuit operation, named mode 1 and mode 2. In mode 1, at the time the inductor current hits the lower bound (LB) at time t_0 , the switching transistor S1 is turned on, and whenever i_L greater than zero S1 conducts, else D1 conducts. The voltage across the output capacitor C1 is greater than the supply voltage peak either S1 or D1 conducts, so the operation is boost operation, and the inductor current slope will be positive. In mode 2, the inductor current hits the upper bound (UB) at time t_1 , the switching transistor S2 is turned on, and depending on the i_L polarity, S2 or the diode D2 conducts. Also, the voltage across the output capacitor, C2 is greater than the mains supply's voltage peak in mode 2. However, the inductor current slope will

be negative in this mode. At time t_2 , the inductor current hits the lower bound. In this way, one switching cycle is completed. The switching frequency is much more higher than the mains supply frequency, so the inductor current tracks the reference current $i_{L,ref}$ easily.



Figure 2.2 Current waveforms in HCC technique, (a) Over one line cycle, (b) Over one switching cycle.



Figure 2.3 The two modes of operation of the half-bridge circuit, (a) Mode 1: positive inductor-current-slope mode, (b) Mode 2: negative inductor-current-slope mode.



Figure 2.4 Block schematic representation of a typical closed-loop system for PFC.

The rectifier's output voltage is the sum of the voltages on the capacitor C1 and C2, so the rectifier's output voltage must be greater than twice its input peak voltage. In order to regulate the rectifier's output voltage (V_o) to a desired level, closed loop control system is applied to the rectifier and it is shown in figure 2.4.



Figure 2.5 Closed-loop control system with imbalance control.

Main drawback of the rectifier topology is imbalance of the capacitor voltages. However, changing the closed loop control system as in figure 2.5 can eliminate this imbalance. Error voltage, ΔV is introduced due to voltage difference over the capacitors C1 and C2. The voltage difference over the capacitors should be eliminated. Feeding ΔV through gain block (K_u) and adding its output i_{L,ref}(t), new reference current i'_{L,ref} is obtained. In this way, voltage difference over the output capacitors are forced to be zero.

$$\dot{i}_{L,ref} = \dot{i}_{L,ref} + K_u \Delta V \tag{2.1}$$

The new block schematic is shown in figure 2.5 and input current tracks $i'_{L,ref}(t)$ instead of $i_{L,ref}(t)$ when the new closed-loop control system is used.

2.2.2 Front-End Rectifier Followed By A Boost Converter

Front-end rectifier followed by a boost converter is a well-known unity power factor rectifier. The rectifier topology and its working principle are explained briefly in this section. Average current control technique expressed in detail in chapter 5 is used as its control technique here [2].



Figure 2.6 Front-end rectifier followed by a boost converter.

In figure 2.6, the rectifier topology with active power factor correction is shown. Sinusoidal input voltage is converted to DC voltage with ripple via the diode bridge rectifier. Boost inductor, Ls begins to store energy when the IGBT is on. In this time, the output capacitor C supplies necessary voltage and current to the load. Stored energy in the boost inductor and the energy from the mains supply is given to the load and the output capacitor when IGBT is off. A power factor corrector integrated circuit (PFC IC) such as UC3854 measures the output voltage, rectified input current, it processes these signals, gives PWM signal with correct duty cycle to a power semiconductor switch such as an IGBT's gate to achieve sinusoidal input side, regulated and correct output voltage at the output side is achieved. In order to turn on and off the IGBT properly, an IGBT driver externally can be used for giving appropriate voltage to the IGBT's gate while some PFC ICs such as UC3854 has an IGBT driver internally.

The power factor corrector integrated circuit (PFC IC) in figure 2.6 is an IC such as UC3854 whose datasheet is given in Appendix B. It operates the rectifier in continuous conduction mode and uses the average current control technique. In continuous conduction mode, the rectifier's input current follows the input voltage waveform, so the input voltage and input current are nearly in phase. This means that nearly unity power factor is achieved at the input side of the rectifier.

The IGBT in figure 2.6 is switched in fixed frequency but the duty cyle of the signal is varying due to the nature of average current control technique.

Amount of high frequency ripple current in the input side of the rectifier is determined by the value of the boost inductor, Ls for the continuous conduction mode of the operation. The ripple current, ΔI must not exceed the twice the input peak current value to become inductor current continuous.

Increasing the value of the boost inductor, ripple current can be made arbitrarily small. As the ripple current on the input current is low, injected high frequency noise to the supply line becomes low. The boost inductor's value can be calculated using the equation 2.2. In this equation, $V_{in(min)}$ is the minimum input peak voltage. D stands for duty cyle of the PWM signal given to the switching transistor's gate. The input sine wave peak current, I_{peak} can be calculated by using the equation 2.3.

$$L = V_{in(\min)} D / (f_s \Delta I)$$
(2.2)

, where $D = (V_o - V_{in(\min)})/V_o$ and $V_{in(\min)} = \sqrt{2}V_{in(rms)\min}$.

$$I_{peak} = \sqrt{2}I_{in(rms)\max} + \Delta I / 2$$
(2.3)

, where $I_{in(rms)\max} = P_{in(\max)} / V_{in(rms)\min}$.



Figure 2.7 Inductor current waveform.

Current variations in i_L in continuous conduction mode for the average current control technique is shown in figure 2.7. When switching transistor is on, boost inductor's current ramps upwards from its initial value. When switching transistor is off, inductor current, i_L ramps down and stored energy in the inductor is given to the load and the output capacitor.

The output capacitance can be calculated by using equation 2.4. Derivation of the equation is based on the energy transfer capability of the capacitor from V_o to V_{min} .

$$C_{o} = 2P_{out}\Delta t / (V_{o}^{2} - V_{o\min}^{2})$$
(2.4)

As shown from the equation 2.4, the rectifier's output voltage ripple is decreased as the output capacitor's value is increased. Also, larger value of the output capacitor means longer hold-up time (Δt). When hold-up time is high, the designed rectifier can tolerate supply interruptions with short durations at the expense of bulky and costly capacitor banks.

2.2.3 Low Conduction Loss AC/DC UPF Rectifier

A novel unity-power-factor controller for single-phase PWM rectifier is shown in figure 2.8. This rectifier has controllable DC link voltage, high efficiency, high power factor, and simple power circuits. Using the "feedback linearization concept", the current controller to obtain sinusoidal input current in phase with input voltage of the rectifier is derived. Two active switches with reverse recovery diodes and two diodes are used for ac-to-dc voltage conversion. The rectifier topology, its working principle and controller are given in reference [3] and they are summarized in this section.





In this topology, boost diode in figure 2.6 is eliminated to avoid the conduction loss of the diode and boost inductor on DC side of the rectifier in figure 2.6 is located to the AC side of the rectifier to reduce the EMI (Electro Magnetic Interference). This rectifier has a simple power circuit compared with the one in figure 2.6. However, this topology requires more complicated control circuit than the one in figure 2.6.

The converter in figure 2.8(a) is controlled by turning on and off the switching transistor simultaneously. There is an assumption that current of the inductor in figure 2.8(a) is continuous. When the switching transistors are turned on simultaneously, the inductor current increases through one of the switches and the other antiparallel diode. In this way, the inductor stores magnetic energy. Diodes, D1 and D2 prevent electrical energy stored in the output capacitor to flow to the AC side of the rectifier since the diodes are reverse biased, and the output capacitor, C provides the load current when the switching transistors are turned on simultaneously. Then,

$$V_s - L\frac{di_s}{dt} = 0 \tag{2.5}$$

, where i_s and V_s are the source current and voltage respectively. L is the inductance of the inductor located on the AC side of the rectifier. When the switching transistors are turned off simultaneously, magnetic energy stored in the inductor and electrical energy from the mains supply given to the load and output capacitor through one diode, D1 or D2 and one parallel diode of the two switching transistors, depending on the direction of input current, i_s . Then,

$$V_s - L\frac{di_s}{dt} - V_d \operatorname{sgn}(i_s) = 0$$
(2.6)

In this equation, V_d is the converter's output voltage, and $sgn(i_s)$ denotes the sign of the rectifier's input current. Since it is a unity power factor rectifier, $sgn(V_s)$ is equal to $sgn(i_s)$. Depending on the duty ratio, D of the PWM signal given the switching transistors' base and direction of the current, average inductor voltage over one switching period, Ts gives the following source current variation, Δi_s :

$$V_{s}D + [V_{s} - V_{d} \operatorname{sgn}(i_{s})](1 - D) = L \frac{\Delta i_{s}}{T_{s}}$$
(2.7)

Hence, the duty ratio, D is equal to:
$$D = D_n + D_c \tag{2.8}$$

, where $D_n = 1 - \frac{V_s |\sin wt|}{V_d}$, $D_c = \frac{L\Delta |i_s|}{V_d T_s}$. In equation 2.8, w and V_s are angular

frequency and its maximum input voltage respectively. The proportional and integral (PI) current controller can be used so that the source current tracks the current command, and duty cyle. PI is expressed by equation 2.9.

$$\mathbf{D}_{c} = k_{pc} i_{e} + k_{ic} \int i_{e} dt \tag{2.9}$$

In this equation, k_{ic} and k_{pc} are the integral and proportional gains respectively. The current error, i_e is given in the equation 2.10.

$$i_e = \left| i_s^* \right| - \left| i_s \right| \tag{2.10}$$

The rectifier's controller block diagram is given in figure 2.8(b). The duty cycle, D_c is necessary to make its input current in phase with the input voltage. The rectifier is originally a nonlinear dynamic system; but with the addition of the nominal duty cycle, D_n to the rectifier, the relation between the input D_c and output $|i_s|$ becomes a first-order linear dynamic system. Thus, the addition of the nominal duty cycle to the duty cycle, D_c relaxes the burden of PI controller and this improves the input current waveform and makes controllability of the rectifier easier.

2.3 Three Phase Unity Power Factor Rectifiers

In this section, the rectifiers, which have three phase mains supply, namely, single transistor three phase multiresonant high power factor (HPF) rectifier, pulse voltage source rectifier with PWM-SHE (Pulse Width Modulation- Specific Harmonic Elimination) and single controlled switch three phase rectifier with unity power factor and sinusoidal input current are expressed in detail in reference [4], [5], [6]. The papers in these references are summarized and the rectifier topologies are expressed briefly here.

2.3.1 Single Transistor Three-Phase Multiresonant ZCS HPF Rectifier

A single transistor three-phase multiresonant ZCS high power factor rectifier topology and its working principle are given in reference [4]. These are summarized in this section.

The unity power factor rectifier is a new family of a three-phase single switch high power factor (HPF) multiresonant buck type rectifier. It has continuous input and output current. The three phase multiresonant ZCS (Zero Current Switching) cell and resonant capacitor, C_{r1} 's voltage shapes are shown in figure 2.9. The transistor operates with ZCS and the diodes operate with ZVS (Zero Voltage Switching) by using a multiresonant scheme. These single transistor multiresonant rectifiers draw higher quality of input current waveform at nearly unity power factor, have lower stresses than the quasi-resonant rectifiers and PWM boost type rectifiers, which operate in discontinuous conduction mode (DCM), since the portions of periods, B and C in a switching interval shown in figure 2.9(c) are shorter than those of other type of topologies. More linear input characteristic is produced due to the dominance of the period, A and the voltage, V_{Cr1} is proportional to the input current in this time interval. Since this rectifier is buck type rectifier, the rectifier's output voltage lower than the rectifier's input peak voltage. This type of rectifiers has low stresses on the semiconductor devices and wide load range.

 L_a , L_b , L_c and L_f are the filter inductors had small switching-frequency current ripples. The resonant capacitors, C_{r1} , C_{r2} , C_{r3} 's average voltage during a switching period are equal to the input voltages, Φ_a , Φ_b and Φ_c respectively at steady state. Furthermore, their peak voltages and the input currents, i_a , i_b , i_c are proportional to each other. Since the switching frequency is much higher than input line frequency, the input current waveshape follows the input voltage waveshape easily. High power factor and low harmonic input current content are results of this phenomenon.

It is sufficient to consider a 30^{0} interval of the AC input line waveform by assuming the three phase line voltage is well balanced and symetric to make analysis of the three phase multiresonant buck type rectifier shown in figure 2.9(a). Also, it is assumed that the switching frequency of the rectifier is much higher than the input



Figure 2.9 (a) Three-phase multiresonant ZCS cell. (b) The input-side resonant-tank capacitor-voltage waveform together with the input-current waveform. (c) The enlarged waveform of V_{Cr1} during one switching period.

line frequency. Analysis of rectifier in 30^0 interval, where $i_a > 0 > i_c > i_b$, is described here[4]:

- Interval 1:to≤t≤t1, All switches are off except D_d: In this interval, each tank capacitor, C_{r1} –C_{r3} charges up linearly at a rate proportional to its respective line current. This will continue until switch S₁ is turned on. During this interval, resonant-tank inductor, L_r supplies the output load current. When switch S₁ turns on, the input line-to-line voltage, V_{ab} is maximum and this forces D₁ and D₅ to conduct.
- 2) Interval 2: t1≤t≤t2, D₁, D₅, D_d, and S₁ are on: In this interval, the capacitor voltage, V_{Cr3} continues to increase while the other two capacitor voltages ring with resonant-tank inductor, L_r until the tank inductor current increases to zero from the negative output current. Then, the diode, D_d turns off, initiating the next interval.
- 3) Interval 3: t2≤t≤t3, D₁, D₅, and S₁ are on: In this interval, the capacitor voltage, V_{Cr3} continues to increase, while the other two capacitor voltages ring with resonant-tank inductor, L_r. This will continue until V_{Cr3} becomes equal to

 V_{Cr2} . Then, diode, D_6 also conducts. Depending on the magnitudes of i_b and i_c , the order of interval 2 and 3 may be reversed. Interval 3 occurs before interval 2 when currents i_b and i_c are similar in magnitude.

- 4) Interval 4: $t3 \le t \le t4$, D₁, D₅, D₆ and S₁ are on: In this interval, the resonanttank capacitors, C_{r1}- C_{r3}, parallel capacitor, C_d and resonant-tank inductor, L_r form a resonant-tank circuit. This interval ends when the resonant-tank capacitor's voltages, V_{Cr1} - V_{Cr3} discharge to zero.
- 5) Interval 5: t4≤t≤t5, All switches are on except D_d: In this interval, the parallel capacitor, C_d and resonant-tank inductor, L_r ring until the tank inductor current decreases to the negative load current. At this point, the input bridge rectifiers become reverse biased. Hence, the switch current becomes zero.
- 6) Interval 6: t5≤t≤t6, All switches are off: Interval 6 is actually a subset of Interval 1 as shown in the waveforms of figure 2.10. The capacitor, C_d's voltage linearly decreases until it reaches zero voltage and the diode, D_d turns on. During the input diode bridge are reverse biased, the complete load current is supplied by parallel capacitor C_d. Therefore, the switch, S₁ can be turned off with ZCS as shown in figure 2.10.

The system behaviour in the entire line period can be investigated by the extension of the interval $[0, \pi/6]$ due to the three-phase input voltages' symmetry. One operating point at the time $\pi/2$ is chosen to simplify the three phase input and single ended output circuit shown in figure 2.10. At the time $\pi/2$, the phase voltage, V_{an} has its peak value. The other phase voltages, V_{bn}, V_{cn} are equal in magnitude to one half of V_{an} and both are negative. At this condition, capacitors C_{r2} and C_{r3} shown in figure 2.10 are charged and discharged exactly in the same manner. Thus, the capacitors C_{r2} and C_{r3} are considered as parallel connected capacitors. At this operating point, the phase current, ian has its peak value. Also, ibn and icn are both one half of the negative phase current, ian. Hence, current source Ig as shown in figure 2.11 can be placed instead of the input voltage sources and the input filter inductors. I_g in figure 2.11 is the peak phase current, $i_{an-peak}$. The effective capacitor, C_x shown in figure 2.11 can be put instead of the input side resonant capacitors, $C_{r1} - C_{r3}$. C_x is equal to the series connection of C_{r1} and parallel-connected C_{r2} and C_{r3}. The three phase input diode bridge in figure 2.10 is represented by the diodes D_1 and D_2 in figure 2.11. The current source, I in figure 2.11 is placed instead of the output filter

inductor in figure 2.10. Simplified single input and single ended circuit diagram is obtained and shown in figure 2.11.



Figure 2.10 Basic single transistor three-phase multiresonant ZCS HPF rectifier circuit and its ideal waveforms.

This simplified single input, single ended output model and the relations between the actual three phase input circuit are given approximately as follows:

- 1) $C_x = C_r \times 2/3$, where input-resonant capacitors $C_{r1} C_{r3}$ have the same values and are represented as C_r .
- 2) $I_g = peak phase current, i_{an-peak}$.
- 3) $V_g = 3/2$ times the peak-phase voltage, $V_{an-peak}$, where V_g is also the same as the average voltage of V_{Cx} during one switching period.
- 4) Three-phase input power $P_{in} = V_g \times I_g = 3/2 \times V_{an-peak} \times i_{an-peak}$.



Figure 2.11 The simplified single-input and single-output model of the singletransistor three-phase multiresonant ZCS HPF rectifier.

The converter equations are normalized with respect to the DC output voltage and they are given in the following equations:

Base voltage =
$$V$$
 (2.11)

Base current,
$$I_o = V/R_o$$
 (2.12)

Base impedance,
$$R_o = \sqrt{\frac{L_r}{C_x}}$$
 (2.13)

Base frequency,
$$f_o = \frac{1}{2\Pi\sqrt{L_r C_x}}$$
 (2.14)

V in equation 2.11 and 2.12 is the rectifier's output voltage.

The normalized values of the input and output quantities can be given as $M_g = V_g/V$, $J_g = I_g R_o/V$, $J = I R_o/V = M_g J_g$, $\alpha = 2\pi f_o(t_1-t_0)$, $\beta = 2\pi f_o(t_5-t_1)$, and $\gamma = (\alpha+\beta)/2$ if it is assumed that the rectifier is an ideal loss-free system. The switch, S_1 's off time is t_1-t_0 and its on time is t_5-t_1 as shown in figure 2.10.

2.3.2 12-Pulse Voltage Source Rectifier With PWM-SHE

12-pulse voltage source rectifier with PWM-SHE is explained in detail in reference [5]. It is summarized briefly in this section.

AC drive system used with medium to high voltage systems is shown in

figure 2.12. Two identical voltage source rectifiers connected in series are used as active front-end rectifier. The rectifier shown in figure 2.12 is a 12-pulse rectifier and its lowest order input current harmonics are 11^{th} and 13^{th} harmonics with a balanced input voltage source. The input dual transformer in the rectifier eliminates the input current harmonics in the order of $6k \pm 1$, where k = 1,3,5,...(2n-1). Hence, the input dual transformer eliminates 5^{th} , 7^{th} , 17^{th} and 19^{th} harmonics. To eliminate all the harmonics lower than 23^{rd} , only 11^{th} and 13^{th} harmonics are required to be eliminated by the PWM-SHE (PWM-Specific Harmonic Elimination) method. Hence, a quite low switching frequency can be obtained and it is an advantage. The minimum switching frequency can be 350 Hz for this topology, but 750 Hz for the single converter to eliminate all the harmonics with the order lower than 23^{rd} is necessary.

Dynamic and steady state voltage-sharing problem seen for the series devices in a single bridge topology are avoided in this topology. This high-power mediumvoltage PWM AC/DC voltage source rectifier has the ability to deliver nearly sinusoidal current from mains supply at unity power factor while phase controlled high-power medium-voltage rectifiers don't have this ability.



Figure 2.12 Main circuit of the voltage source converter drive system.

The AC and DC systems are interfaced by the rectifier using bi-directional current and unidirectional voltage blocking switches in figure 2.12. The rectifier exhibits voltage source characteristics at the DC terminal (DC link capacitors) and it exhibits current source characteristics at the AC supply terminal. On the basis of minimum ripple requirement of the DC bus voltage, V_{dr} , the DC link capacitor should be chosen. The relationship of switching frequency, F_s , DC voltage ripple, V_{dr} , input line current, I_m , and modulation index, M are given in equation 2.15.

$$C = \frac{I_m}{4F_s V_{dr}} \left[\cos(\frac{\Pi M}{4}) - \cos(\frac{\Pi M}{2}) \right]$$
(2.15)

In order to maintain nearly unity power factor at its input side, there are various control strategies such as hysteresis current control, phase and amplitude control. However, their performance is not same. One different current control strategy is suitable for PWM-SHE (Pulse Width Modulation-Specific Harmonic Elimination) method and it is applied here. According to this control method, the rectifier's AC input voltage is made equal to the AC line voltage. The resulting power factor is greater than 0.98 with the assumption that the link reactor is less than 0.4 p.u. Figure 2.13 shows the corresponding phasor diagram and V_{rec} is the rectifier input voltage in this figure.



Figure 2.13 Phasor diagram of front-end voltage source converter.

In order to evaluate the performance of different kinds of PWM patterns, there are various criteria. Pulse width modulation-specific harmonic elimination method has lower total harmonic distortion (THD), harmonic loss factor (HLF), and second order distortion factor (DF2) when it is compared with other control strategies under similar conditions. Hence, the most effective high performance results can be obtained by using pulse width modulation-specific harmonic elimination patterns.

All of angles of pulse width modulation-specific harmonic elimination can be precalculated and stored in memory so that the unexpected narrow switching pulses produced randomly by other online PWM patterns disappear. In order to prevent to destroy the switching devices, too narrow pulses (1-100µs) are avoided to apply the gate of the switching devices since they may violate the restrictions on minimum on time and turn off delay of switching devices.



Figure 2.14 PWM-SHE switching pattern for 11th and 13th harmonic elimination (modulation index M=0.9). (a) Switch gating signal. (b) Line-to-line voltage.

There are well known advantages of pulse width modulation-specific harmonic elimination (PWM-SHE) patterns. Since over-modulation is possible, high voltage gain can be obtained. The ripple on the DC link current is small due to the high quality output current and voltage, so reduction in the size of the DC link filter components can be achieved. Moreover, elimination of the lower order harmonics prevents harmonic interference such as resonance with external line filtering networks.

The switching pattern, which especially eliminates the 11th and 13th harmonics, is just common two-level line-to-neutral PWM pattern, and it has quarter and half wave symmetries. Fourier coefficients of the switching pattern are given in equation 2.16 and 2.17.

$$a_n = \frac{4}{n\Pi} \left[-1 - 2\sum_{k=1}^{N} (-1)^k \cos(n\alpha_k) \right]$$
(2.16)

 $b_n = 0 \tag{2.17}$

The switching pattern can be obtained by solving three nonlinear functions ($a_{11}=0$, $a_{13}=0$, and $a_1=M$), and it is shown in figure 2.14.

2.3.3 Single Controlled Switch Three Phase Rectifier With Unity Power Factor and Sinusoidal Input Current

A single controlled switch three phase rectifier with unity power factor and sinusoidal input current and its working principle are expressed in detail in reference [6]. They are summarized in this section.

Single controlled switch three phase rectifier is possible with low harmonic rectification, without active control of the line currents, and without the use of bulky low-frequency passive elements. The three phase rectifier topology, the voltage waveshape of the input capacitor, C_{sa} , and the input current, i_{sa} is shown in figure 2.15.

The rectifier has two stages. Three phase diode bridge with the capacitors,

 C_{sa} , C_{sb} , C_{sc} and inductors, L_{sa} , L_{sb} , L_{sc} connected to there phase mains supply is one of the two stages. The other stage is the active output stage consisting of the boost type single ended converter.



Figure 2.15 Topology of the converter.

Discontinuous current control technique with variable frequency is employed to the active switching device of the output stage. In this rectifier topology, only one boost inductor, L_B is used. The rectifier has a pulsating input voltage during each switching period. The pulsating input peak voltage, V_{Csa} is proportional to the input line current, i_{sa} and this provides an average of line current nearly sinusoidal and approximately proportional to the phase voltage, V_{an} . The input inductors, L_{sa} , L_{sb} , L_{sc} filter the input line currents, i_{sa} , i_{sb} and i_{sc} respectively.

In order to obtain a high quality input current, the input inductors are forced to operate in discontinious current mode in this high frequency converter. The input capacitors C_{sa} , C_{sb} , and C_{sc} , which are chosen to be sufficiently small to operate the

circuit in discontinuous voltage mode are used in the boost rectifier shown in figure 2.15 to obtain low harmonic rectification.

Operation mode of the rectifier includes energy transfer from capacitors C_{sa} , C_{sb} , and C_{sc} to the inductor L_B . Switch Q is turned on and the capacitors are discharged by the resonating switch current to achieve this energy transfer. All bridge diodes are conducting when the capacitors voltages are reduced to zero. The magnetic energy stored in the inductor, L_B is transferred to the load and the output capacitor through the diode, D_B when the switching transistor, Q is turned off. The input capacitors, C_{sa} , C_{sb} , and C_{sc} are charged linearly by the currents, i_{sa} , i_{sb} and i_{sc} respectively when the diode, D_B turns off until the switching transistor Q is turned on again.

The system behaviour with suitable phase shifts for the entire fundamental period can be analysed in a 60° interval. The interval which corresponds to the condition: $V_{sa}>0>V_{sb}>V_{sc}$ is considered here. It is assumed that the mains supply's phase voltages are sinusoidal, balanced, and approximately constant within a pulse period. Operation modes of the rectifier within the 60° interval is as the following:

- Mode 1: t0<t<t1, diodes D1, D6 and switch Q are on. At t=t₀, switch Q turns on; during this interval, the capacitor voltage V_{Csc} continues to increase, while the other capacitors, C_{sa}, C_{sb} and the inductor, L_B resonate until the capacitor voltage, V_{Csb} reaches to the capacitor voltage, V_{Csc}. The diode, D2 then also conducts during the next interval.
- Mode 2: t1<t<t2, diodes D1, D2, D6 and switch Q are on. The three capacitors, C1, C2, C3 and the inductor L_B resonate until all the capacitors are discharged by a pulsed current.
- 3. Mode 3: t2<t<t3, all diodes are on, switch Q is off. This mode starts when the voltages across the capacitors are equal to zero and the switch, Q is then turned off and energy is transferred from the boost inductor to the load through diode D_B . All diodes in this rectifier conduct. This mode ends when the boost-inductor current becomes equal to value of the line current, i_{sa} .
- 4. Mode 4: t3<t<t4, diodes D1, D6 and D_B are on. The capacitors, C_{sa}, C_{sb} and the inductor, L_B resonate. The capacitor, C_{sc} is charged linearly by its line current. The duration of this mode is negligible. The diodes, D1, D6 and D_B are conducting during this interval. This mode ends when the boost inductor

current becomes zero.

5. Mode 5: t4<t<t5, all switches are off. The input capacitors, C_{sa}, C_{sb}, C_{sc} are charged linearly to their peak values by currents i_{sa}, i_{sb} and i_{sc}. This mode ends when the switch Q turns on.

CHAPTER III

SELECTION OF UPF RECTIFIER TOPOLOGY

3.1 Introduction

Some UPF (Unity Power Factor) rectifier topologies were investigated in previous chapter. One of the topologies should be selected to simulate. Cheapest, simplest and best suitable rectifier among the ones investigated in previous chapter is tried to be picked out in this chapter. Transistor ratings, value of inductors and capacitors of the selected rectifier topologies in previous chapter are calculated and number of all critical components are given. At the end of the chapter, these values and ratings are given in a table. The calculations of the unity power factor rectifiers are made on the basis of the following specifications:

- Rectifiers' output power = 3 kW
- Switching frequency, $f_{sw} = 20$ kHz (it may be different)
- Rectifiers' output voltage = $530 \pm 2 \% V$
- Mains supply's voltage = 220V rms if single phase rectifier
- Mains supply's voltage = 380V (line to line) rms if three phase rectifier
- Mains supply's frequency = 50 Hz
- Input current ripple, $\Delta I = 5$ % peak to peak current
- Rectifier efficiency, $\eta \ge 95$ % at full load
- Input power factor, $pf \approx 1$

The rectifiers' efficiency is equal to or greater than 95 % according to above specifications. Then, their maximum input power is calculated as 3158 VA by using

the equation 3.1. In calculation of the component ratings, the rectifiers' output power is taken as 3158 VA.

$$Efficiency = \frac{Power(output)}{Power(input)}$$
(3.1)

Diodes and switching power transistors' current and voltage ratings are taken as two times what is calculated in the following sections to prevent possible failure of the semiconductor devices except in section 3.3.1 since necessary safety margins are taken in the calculations in section 3.3.1.

3.2 Calculations Of Single Phase Unity Power Factor Rectifiers

In this section, single phase half-bridge boost rectifier, front-end rectifier followed by a boost converter and low conduction loss AC/DC UPF rectifier's switching transistors' voltage and current ratings are calculated according to the specifications given in section 3.1. Values of power inductors and capacitors used in these rectifiers, which have single phase AC mains supply are also calculated and number of critical components are given.

3.2.1 Calculations Of Unity Power Factor Converter Using Half-Bridge Boost Topology

Single phase half-bridge boost topology is investigated before in section 2.2.1. In this section, its switching transistors and their free-wheeling diodes' voltage and current ratings, value of its boost inductor and power capacitors are calculated according to the specifications given in section 3.1. These are listed at the end of this section together with number of the switching transistors and boost inductor and power capacitors.

Using the specifications in section 3.1 and equation 2.3 describing that the boost inductor's peak current is sum of the high frequency ripple current and the sum

of the power frequency line current [2], the rectifier's maximum input rms current, $I_{in(rms)max}$, its boost inductor's ripple current, ΔI , and peak current, I_{peak} are calculated as 16.887 A, 1.2247 A and 24.494 A respectively.

Since this rectifier topology's output voltage is greater or equal to two times its peak input voltage, modulation index, M is equal or greater to 2 [1]. Modulation index formula is given in equation 3.2.

$$M = \frac{V_o}{V_p} \ge 2 \tag{3.2}$$

, where $V_p = \sqrt{2}V_{in(rms)}$. The voltage, V_p is equal to 311.12 V when the rectifier's input voltage is equal to 220 V rms. The rectifier's output voltage becomes its minimum output voltage when the modulation index is taken as 2. Its minimum output voltage, V_{omin} is calculated as 622.25 V when the voltage, V_p and the modulation index, M are taken as 311.12 V and 2 respectively. Hence, the output voltage specification in section 3.1 can't be achieved in this topology. The rectifier's peak output voltage is sum of its DC output voltage and output voltage ripple. Its peak voltage is found as 634.695 V by taking the output voltage ripple as ± 2 % of the pure DC output voltage. The modulation index is found as 2.04 when the rectifier's output voltage is at its peak by putting the peak output voltage to the equation 3.2 and taking V_p voltage as 311.12 V. This value of modulation index is used in the following calculations in this section.

The maximum switching frequency, f_{smax} is given in equation 3.3 and its derivation is given in reference 1.

$$f_{s\max} = \frac{V_o}{4L\Delta i} \tag{3.3}$$

The boost inductance in this rectifier topology is calculated as 6.47 mH by putting the values of ΔI and V_o in to equation 3.3 and taking f_{smax} as 20 kHz. The minimum switching frequency is given by the equation 3.4 and it occurs when the mains supply's voltage is at its peak value [1].

$$f_{s\min} = 4f_{s\max}\left(\alpha + \frac{1}{M}\right)\left((1-\alpha) - \frac{1}{M}\right)$$
(3.4)

Using the equation 3.4 and taking α (average duty cycle of switching transistor, S2 in figure 2.1) as 0.5, the minimum switching frequency, f_{smin} is calculated as 776.619

Hz. The output current, I_o given to the rectifier's load is found as nearly 4.73 A by dividing its 3 kW output power to its 634.695 V output voltage. The overall output voltage ripple expression is given by equation 3.5 [1].

$$\delta V_o = \frac{I_o}{2\Pi f_s C} \left[2M(1 - 2\alpha)\cos\theta - \sin 2\theta \right]$$
(3.5)

, where $\alpha \ge \alpha_{\min} = \frac{1}{M}$. *C*, f_s , α , and *M* are one of its output capacitances, switching frequency of the switching transistors, average duty cycle of switching transistor, S2 ($(1 - \alpha)$) is the average duty cycle of the transistor S1) in figure 2.1, and modulation index respectively. θ is equal to *wt* and *w* is the angular frequency of the rectifier's input voltage. Minimum capacitance value of the rectifier is calculated using the equation 3.5. Minimum capacitance of the capacitor, C1 or C2 in figure 2.1 is calculated as 204 uF by taking δV_{omax} , θ , α , and f_s as 12.445V (+2 % ripple voltage at the output), -36.375⁰ (because output voltage ripple has its peak value at $\theta = -36.75^{\circ}$), α_{\min} (it is equal to 0.49 at modulation index, M = 2.04) and 20 kHz maximum switching frequency.

There are the derivations of the following equations 3.6, 3.7, 3.8, 3.9 in reference [1].

$$I_{s1,avg} = I_o \left(\frac{2(1-\alpha)M}{\Pi} - 0.5\right)$$
(3.6)

$$I_{s2,avg} = I_o (\frac{2\alpha M}{\Pi} - 0.5)$$
(3.7)

$$I_{D1,avg} = I_o \left(\frac{2(1-\alpha)M}{\Pi} + 0.5\right)$$
(3.8)

$$I_{D2,avg} = I_o \left(\frac{2\alpha M}{\Pi} + 0.5\right)$$
(3.9)

 $I_{s1,avg}$, $I_{s2,avg}$, $I_{D1,avg}$, $I_{D2,avg}$ are calculated as 0.7075 A, 0.7075 A, 5.43 A, 5.43 A by using the equations 3.6, 3.7, 3.8, 3.9 and taking α , M, I_o as 0.5 (since average ratings of the diodes and switches are calculated here), 2.04, 4.73 A respectively. Voltage stresses of the switches, S1, S2, the diodes, D1, and D2 in figure 2.1 are same as the rectifier's peak output voltage, which is equal to 634.695 V. The necessary diode, switching transistor, inductor and capacitor ratings, their number, and an expression about the rectifier's control circuit are listed as following:

Needed:

- 1. 2 IGBT (1269 V, 1.414A)
- 2. 2 reverse recovery diode (1269 V, 10.86 A)
- 3. 1 inductor (L=6.47 mH)
- 4. 2 capacitor (Each $C \ge 204 uF$)
- 5. UPF IC or special control circuit to switch the IGBTs.

3.2.2 Calculations Of Front-End Rectifier Followed By A Boost Converter

Front-end rectifier followed by a boost converter topology is expressed before in section 2.2.2. In this section, its boost inductance and output capacitance and switching transistor's voltage and current rating are calculated according to the specifications given in section 3.1. They are listed at the end of this section.

Its peak input current and input current ripple are calculated as 24.494 A and 1.2247 A respectively by using the equations 2.2 and 2.3 and the specifications given in section 3.1. Its minimum input peak voltage, $V_{in(min)peak}$ is calculated as 264.46 V since its minimum input rms voltage, $V_{in(min)rms}$ is equal to 187 V rms according to the specifications. Putting the values of the minimum input peak voltage, $V_{in(min)peak}$ and the input current ripple, ΔI in to equation 2.2 and taking switching frequency as 20 kHz, value of the boost inductor, Ls is calculated as 5.4 mH.

Its output capacitance is calculated as 1124.2 uF by taking hold-up time, Δt , worst-case output voltage drop, the rectifier's output power as 10 ms, 10 % of its mean output voltage, 3 kW and using the equation 2.4.

The switching transistor's current stress is equal to the rectifier's peak input current, and it is 24.494A. Its voltage stress is same as the rectifier's output voltage. Hence, the transistor's voltage stress is 530 V.

The bridge diodes' voltage stresses in this rectifier is same as its input peak voltage and it is 311 V. These diodes' current stresses are equal to the rectifier input peak current.

The boost diode in the rectifier has same voltage and current stresses as the switching transistor's ones and it should be fast diode. Main components of the

rectifier, value of the output capacitor and boost inductor, the switching transistor and the diodes current and voltage ratings, and which type of IC can control the rectifier are summarized as the following:

Needed:

- 1. Four bridge diode (622 V, 49A)
- 2. One fast boost diode (1060V, 49A).
- 3. One IGBT with reverse recovery diode (1060V, 49A)
- 4. One capacitor (1124.2uF)
- 5. One inductor (5.4mH)
- 6. UPF (unity power factor) IC.

3.2.3 Calculations Of Low Conduction Loss AC/DC UPF Rectifier

In this section, equations related with the boost inductance and output capacitance in low conduction loss AC/DC UPF rectifier expressed before in section 2.2.3 are derived. The rectifier's boost inductance, output capacitance, power semiconductor diodes and switching transistor's voltage and current ratings are calculated according to the specifications given in section 3.1.

In order to calculate the output capacitance and boost inductance, L in figure 2.8(a), equations related with the inductance and capacitance are derived as the following:

1st stage: When switching transistors, S1 and S2 are on:

Using the following inductor voltage equation, and integrating the two sides of the equation, equation 3.10 is obtained.

$$V_{in} = L \frac{di}{dt} \Longrightarrow L di = V_{in} dt$$

$$\int_{0}^{\Delta I} L di = V_{in} T_{1}$$

$$L \Delta I = V_{in} T_{1} \implies T_{1} = \frac{L \Delta I}{V_{in}}$$
(3.10)

2nd stage: When switching transistors, S1 and S2 are off,

$$L\frac{di}{dt} = V_{in} - V_O$$

Two sides of the equation is integrated as the following,

$$\int_{\Delta I}^{0} Ldi = \int_{T_1}^{T_S} (V_{in} - V_O) dt$$

Calculating the above equation, equation 3.11 is derived.

$$-L\Delta I = (V_{in} - V_O)(T_S - T_1)$$
(3.11)

In order to remove T_1 from the equation 3.11, T_1 in equation 3.10 is put in to equation 3.11.

$$-L\Delta I = (V_{in} - V_O)(T_S - \frac{L\Delta I}{V_{in}})$$

$$\Rightarrow L = V_{in(\min)} \frac{(V_O - V_{in(\min)})}{V_O} / (f_S \Delta I)$$
(3.12)

Equation 3.12 expresses the boost inductance, L in terms of the rectifier's output voltage, V_o , minimum peak input voltage, $V_{in(min)}$, the switching frequency, f_s and the input ripple current, ΔI .

Derivation of the output capacitance value using the energy transfer capability of a capacitor is given below:

$$E = \frac{CV^2}{2}$$
, this is the well-known energy formula of a capacitor.

C is the output capacitance and V stands for the voltage over C.

$$E_1 = \frac{CV_o^2}{2}$$
, where V_o is the desired output voltage of the UPF rectifier.
 $E_2 = \frac{CV_{o\min}^2}{2}$, where $V_{o\min}$ is the rectifier's permissible output voltage.

Using the energy difference between E_1 and E_2 ,

$$E_1 - E_2 = P_{out} \Delta t$$

$$C = \frac{2P_{out} \Delta t}{V_0^2 - V_{O\min}^2}$$
(3.13)

In equation 3.13, P_{out} and Δt are the rectifier's output power and hold-up time. Equations 3.12 and 3.13 are equal to equations 2.2 and 2.4 respectively. Thus, the boost inductance and output capacitance in figure 2.8(a) are calculated and same

results are found as in section 3.2.2.

The transistors, S1 and S2, and the diodes, D1 and D2 have voltage stresses equal to the rectifier's output voltage. The two transistors and the diodes' current stresses are equal to the rectifier's input peak current. Therefore, their voltage and current stresses are 530 V, 24.494 A. The two transistors' reverse recovery diodes have the same voltage and current stresses as the transistors.

Needed:

- 1. Two diode (1060V, 49 A)
- 2. Two IGBT with reverse recovery diodes (1060V, 49 A)
- 3. One capacitor (1124.2uF)
- 4. One inductor (5.4mH)
- 5. Complex control circuit or PFC IC to switch IGBTs.

3.3 Calculations Of Three Phase Unity Power Factor Rectifiers

Switching transistors' voltage and current ratings, value of power inductors and capacitors, and number of critical components in the single transistor three phase multiresonant ZCS HPF rectifier, 12-pulse voltage source rectifier with PWM-SHE and single controlled switch three phase rectifier with unity power factor and sinusoidal input current, whose mains supply is 3 phase AC supply are calculated according to the specifications in section 3.1 and given in this section.

3.3.1 Calculations Of Single Transistor Three-Phase Multiresonant ZCS HPF Rectifier

It is a buck type rectifier so it does not give necessary output voltage 530V. Hence input voltage $3 \Phi AC 380V(1-1,rms)$ and output voltage 400V DC is taken for calculations in this configuration. Output power is taken as 3 kW like for other configurations.

In calculations, simplified single input and single output model of the single

transistor three phase multiresonant zero voltage switching high power factor rectifier in figure 2.11 is used. All the equations and the figures below used in calculations are taken from reference 4.

Effective input voltage is given by the equation 3.14 and equation 3.15 expresses the normalized input voltage equation.

$$V_g = \frac{3}{2} V_{an-peak} \tag{3.14}$$

$$M_g = \frac{V_g}{V} \tag{3.15}$$

, where V is the rectifier's output voltage in figure 2.9(a). Effective input voltage for this rectifier is calculated as 465 V by using equation 3.14. Normalized input voltage is found as 1.1625 by putting 400 V as V and 465 V as V_g in to equation 3.15.



Figure 3.1 The normalized input characteristic of the single-transistor three-phase multiresonant ZCS HPF rectifier. The normalized transistor off time α is expressed in radians.

Maximum normalized peak input current, J_g is calculated as 1.3 by using the graphic in figure 3.1 at $M_g = 1.1625$ for 3 kW rectifier. Normalized frequency, F is calculated as 1.22 by using the equation 3.16 and taking α as 2 (α is calculated as 2 using the figure 3.1 when M_g is equal to 1.1625 and J_g is taken as 1.3).

Normalized frequency,
$$F = \frac{2\Pi}{\alpha + \Pi}$$
 (3.16)

The rectifier's output current, I_{out} is calculated as 7.5 A since its output power is 3 kW, and its output voltage is 400V. Characteristic impedance equation is given in the following equation. The characteristic impedance is calculated as 80.6 Ω putting the necessary values calculated above in to equation 3.17.

$$R_o = M_g J_g \frac{V}{I} \tag{3.17}$$

Base frequency, f_o is calculated as 16.39 kHz by taking switching frequency, f_s as 20 kHz and putting the value of normalized frequency, F in to equation 3.18.

$$f_o = \frac{f_s}{F} \tag{3.18}$$

$$R_o = \sqrt{\frac{L_r}{C_x}} \tag{3.19}$$

, where C_x is effective capacitance, and L_r is resonant tank inductor in figure 2.11. Another base frequency equation is given in equation 3.20.

$$f_o = \frac{1}{2\Pi\sqrt{L_r C_x}} \tag{3.20}$$

Values of C_x and L_r are calculated as 0.12 uF and 0.78 mH respectively by using equation 3.19 and 3.20.

$$C_x = \frac{2}{3}C_r \tag{3.21}$$

, where C_r is the capacitance of one of the input side resonant tank capacitors C_{r1} , C_{r2} , and C_{r3} . C_x and C_r are related with each other as in the equation 3.21. Using the equation 3.21, C_r is calculated as 0.18uF.



Figure 3.2 Normalized voltage stress of switch S₁.

Normalized voltage stress is calculated as 3 for this rectifier at $M_g = 1.1625$ and $J_g = 0.7$ by using the figure 3.2. J_g is taken as 0.7 at $M_g = 1.1625$ since the switch's normalized voltage stress is at its peak when J_g is equal to 0.7.

$$V_{s1} = M_{s1} \times V \tag{3.22}$$

, where M_{s1} is normalized voltage stress and V_{s1} is the switch, S1's peak voltage.

The switch, S1's voltage stress with safety margin is calculated as 1200 V by using equation 3.22.

The switch, S1's normalized maximum current stress is observed as 3.75 from figure 3.3 when M_g and J_g are equal to 1.1625 and 0.8 respectively.

Peak switch current is given by the equation 3.23.

$$I_{s1} = J_{s1} \times \frac{V}{R_o} \tag{3.23}$$

, where J_{s1} is normalized peak switch current. The peak switch current is calculated as 18.6 A by using the equation 3.23.

The output diode, D_d 's normalized maximum voltage stress is observed as 3 from figure 3.4 when M_g and J_g are equal to 1.1625 and 0.4 respectively. Its peak voltage stress is expressed by equation 3.24.



Figure 3.3 Normalized current stress of switch S1.



Figure 3.4 Normalized voltage stress of output diode.

$$V_{Dd} = M_{Dd} \times V \tag{3.24}$$

, where V_{Dd} stands for the diode's peak voltage stress and M_{Dd} is its normalized voltage stress. Its current stress is equal to the rectifier's output current (7.5 A). Input diodes, D1, D2, D3, D4, D5, D6 in figure 2.9(a) have the same rating as the switching transistor.

Needed:

- 1. Five inductor ($L_r = 0.78$ mH critical)
- 2. Five capacitor ($C_r = 0.12$ uF critical)
- 3. One output diode (1200 V, 7.5 A), 6 input diode
- 4. One IGBT or power transistor (1200 V, 18.6 A)
- 5. Microcontroller or an special IC to calculate and switch the IGBT at proper times.

3.3.2 Calculations Of 12-Pulse Voltage Source Rectifier With PWM-SHE

12-Pulse voltage source rectifier with PWM-SHE is explained before in section 2.3.2. Its output capacitances, input inductances and switching transistors' current and voltage stresses are calculated by using the specifications given in section 3.1 in this section.

Switching angles, α_1 , α_{11} and α_{13} are calculated as 31.41° , 5.2611° and 2.5667° respectively by using equation 2.16, 2.17, and taking a_{11} , a_{13} , a_1 , M as 0, 0, 0,9, 0,9 as in reference [5].

 e_1 and e_2 are two of three phase voltages of the upper part of the rectifier and they are expressed in equation 3.25 and equation 3.26 respectively.

$$e_1 = 190\sqrt{2}\cos wt$$
 (3.25)

$$e_2 = 190\sqrt{2}(\cos(wt - 120)) \tag{3.26}$$

$$(L_1 + L_2)\frac{di}{dt} + e_1 + u_C - e_2 = 0$$
(3.27)

Equation 3.27 is the node voltage equation of the rectifier's upper part in the converter in figure 2.12. In this equation, u_C stands for the rectifier's upper part's output voltage and L_1 and L_2 are two of the three inductors in the upper part of the

rectifier.

$$i_1 = I\cos wt \tag{3.28}$$

, where i_1 is the current flowing through the inductor, L_1 and i_2 is the current flowing through the inductor, L_2 and it is 120^0 phase delayed version of the current, i_1 . I is the magnitude of i_1 or i_2 . I is calculated as 3.9176 A by dividing its output power to i_1 in rms and multiplying by $\sqrt{2}$.

Assuming its input ripple current can be 5 % peak to peak, ΔI is calculated as 0.20089A. f_s can be as small as 350 Hz in this configuration [5].

$$\Delta wt = 5.2611^{\circ} - 2.5667^{\circ} = 2.6944^{\circ}$$
$$\Rightarrow \Delta wt = 2\Pi f \Delta t = 2.6944^{\circ} \Rightarrow \Delta t = 0.003309 \text{ s}$$

Equation 3.29 is derived by investigating the rectifier when Δt is equal to 0.003309 s and putting the e₁ in equation 3.25 and the e₂ in equation 3.26 in to equation 3.27 and splitting the currents i₁ and i₂ in to two parts, one with ripple and the other without ripple.

$$L_{1}\left(\frac{di_{1}}{dt} + \frac{\Delta I}{\Delta t}\right) + L_{2}\left(\frac{di_{2}}{dt} + \frac{\Delta I}{\Delta t}\right) + 190\sqrt{2}\left[\cos(wt - 120) - \cos wt\right] + u_{c} = 0 \quad (3.29)$$

 L_1 and L_2 are calculated as 128.337 mH by using the calculated values of Δt and ΔI when wt is equal to 120^0 . When switching frequency is taken as 20 kHz, each of these inductances will be nearly 2.25 mH. In these calculations, u_C is taken as 275 V since the rectifier's output voltage is 550 V and the rectifier has two part and their output voltage is assumed as 1/2 times the rectifier's output voltage.

$$C = \frac{I_m}{4f_s V_{dr}} \left[\cos(\frac{\Pi M}{4}) - \cos(\frac{\Pi M}{2}) \right]$$
(3.30)

The relationship between its output capacitance, C, input line current, I_m , switching frequency, f_s , modulation index, M and output voltage ripple, V_{dr} is given by the equation 3.30, and total output capacitance is calculated as 1.6 uF (modulation index, M = 0.667, I_m = 3.9176 A, switching frequency, f_s = 20 kHz, 2 % of its output voltage = output voltage ripple = 11 V) but its total output capacitance is calculated as 93 uF when its switching frequency, f_s is equal to 350 Hz. Since there are two capacitors in series at its output side, each of the two capacitances is 3.2 uF but 186 uF when its switching frequency, f_s is equal to 350 Hz.

IGBT voltage stress is equal to half of its output voltage. IGBT current stress is equal to $I + \Delta I$ (4.12A).

Needed

- 1. Three winding transformer
- 2. 12 IGBT (550V, 8.24A)
- 3. 6 inductor (each 2.25 mH)
- 4. 2 capacitor (3.2 uF)
- 5. Microcontroller (and EEPROM) or DSP IC is required to calculate the necessary switching angles.

3.3.3 Calculations Of Single Controlled Switch Three Phase Rectifier With Unity Power Factor And Sinusoidal Input Current

In this section, main power semiconductors' current and voltage stresses and the boost inductance and main capacitances in the rectifier expressed in section 2.3.3 are calculated according to the specifications given in section 3.1.

Normalized value of the boost inductor, I_{LBn} , normalized switching frequency, F_{sn} , normalized output resistance of the rectifier, R_{outn} can be selected as 4, 0.954 and 50 respectively by considering silicon utilization, voltage stress across the semiconductor devices and the cost and size of reactive elements [6].

Normalized base current, base impedance, base voltage, base frequency are given in equation 3.31, 3.32, 3.33, 3.34 [6].

Base current,
$$I_b = V_b / Z_b$$
 (3.31)

Base impedance,
$$Z_b = \frac{\sqrt{L_B}}{\sqrt{C_e}}$$
 (3.32)

, where L_B is the boost inductance and C_e is the equivalent capacitance of the capacitors, C_{sa} , C_{sb} , and C_{sc} in figure 2.15 when the capacitors are connected parallel.

Base voltage,
$$V_b = V_{out}$$
 (3.33)

Base frequency,
$$F_b = \frac{1}{2\Pi\sqrt{L_B C_e}}$$
 (3.34)

Taking switching frequency, fs as 20 kHz and the rectifier's output voltage, V_{out} as 550 V and using the equations 3.35 and 3.36 given in reference [6], the boost

inductor, L_B and equivalent input capacitance, C_e are calculated as 15.3 uH and 3.76 uF respectively.

$$L_B = \frac{R_{out}F_{sn}}{2\Pi R_{outn}f_s}$$
(3.35)

, where R_{out} is the rectifier's output resistance or load resistance.

$$C_e = \frac{R_{outn}F_{sn}}{2\Pi R_{out}f_s}$$
(3.36)

$$C_e = \frac{2}{3}C_{sa} \tag{3.37}$$

Putting the calculated value of the equivalent input capacitance, C_e in to the equation 3.37 given in reference [6], the input capacitance, C_{sa} is calculated as 5.65 uF. Since the input capacitances, C_{sb} and C_{sc} have same capacitance with C_{sa} , their values are also 5.65 uF. In this topology, equation 3.13 can be used for the calculation of the output capacitor. The output capacitance is calculated as 1044 uF by taking hold-up time, output power, output voltage and worst-case minimum output voltage as 10 ms, 3 kW, 550 V, 495 V respectively and using equation 3.13.

The switching transistor's voltage stress is calculated as 550V by ignoring voltage drop of D_B . Its current stress is same as one of the rectifier's input peak phase current, which is equal to 3.38 A. Current and voltage stresses of the power diodes in figure 2.15 are same as those of the IGBT.

Needed:

- 1. 1 IGBT (1100 V, 6.76 A)
- 2. 4 capacitor (three of them is 5.65 uF and one is 1044 uF)
- 3. 1 boost inductor (15.3 uH) and 3 input inductor
- 4. 7 power diode (1100 V, 6.76 A)
- 5. Unity power factor integrated circuit is needed.

3.4 Results of Calculations

Critical electrical components in the rectifier topologies investigated in chapter 2 are calculated in this chapter. The calculation results are summarized in

		Critical		
TOPOLOGY	TRANSISTOR	L	С	COMMENT
Single Phase Half- Bridge Boost Topology	2 IGBT (1269, 1.414A)	6.47 mH	2 capacitor (each 204 uF)	UPF IC or specific IC to drive IGBTs. Output voltage should be 2 times input voltage amplitude
Front-End Rectifier Followed By A Boost Converter	1 IGBT (1060, 49A)	5.4 mH	1 capacitor (1124.2 uF)	This topology is very cheap and easy to implement and also it satisfies all necessary specification. This configuration needs UPF IC
Low Conduction Loss AC/DC UPF Rectifier	2 IGBT (1060, 49A)	5.4mH	1 capacitor (1124.2uF)	Conduction loss is lower than front-end rectifier followed by a boost converter
Single Transistor Three Phase Multiresonant ZCS HPF Rectifier	1 IGBT (1200V, 18.6 A)	0.78mH	No comment about its output capacitor in reference [4]	Microcontroller or special IC needed to calculate and switch the IGBT at proper times
12-Pulse Voltage Source Rectifier with PWM- SHE	12 switching transistor (550V, 8.24A)	6 inductor (each 2.25 mH)	2 capacitor (each 3.2 uF)	DSP or microcontroller is needed to switch the IGBTs using SHE technique
Single Controlled Switch Three Phase Rectifier with Unity Power Factor and Sinusoidal Input Current	1 IGBT (1060V 6 76 A)	15 3 uH	1044uF	UPF IC is needed

 Table 3.1 Comparison of the rectifiers.

table 3.1.

Single transistor three phase multiresonant ZCS (zero current switching) HPF (high power factor) rectifier scheme uses single switching transistor, but it needs microcontroller or special integrated circuit to switch the IGBT at proper times.

In 12-pulse voltage source rectifier with pulse width modulation-specific harmonic elimination method, the switching frequency can be as low as 350 Hz. However, it uses 12 switching transistor such as IGBT or GTO. Microcontroller or DSP integrated circuit is needed to control the switching transistors. It is suitable for high power applications (e.g. 100 kVA).

Single controlled switch three phase rectifier with unity power factor and sinusoidal current uses a single switching transistor, and a PFC integrated circuit is enough to control the switching transistor.

Single phase mains supply is much more available than three phase mains

supply. Hence a single phase unity power factor rectifier topology is more convenient to design a 3 kW unity power factor rectifier although a 3 kW three phase unity power factor rectifier can also be designed.

Single phase half-bridge boost topology requires two switching power transistor. Output voltage can not be smaller than two times of the mains supply's voltage amplitude. In Turkey, single phase mains supply's voltage amplitude is 311 V $(220 \times \sqrt{2})$ and two times of this value is equal to 622 V. Its required output voltage should be between 400 and 550 V, so single phase half-bridge boost topology mentioned in chapter 3 is not selected to simulate.

In front-end rectifier followed by a boost converter topology, one transistor is used for switching purposes, but diode, D in figure 2.6 increases conduction losses.

Low conduction loss AC/DC UPF rectifier uses two switching transistor, but there is no diode bridge. It has relatively complex control circuit given in figure 2.8 comparing the front-end rectifier followed by a boost converter's control circuit, but it can be controlled with a PFC integrated circuit as the front-end rectifier followed by a boost converter. However, application of PFC integrated circuit with the low conduction loss AC/DC UPF rectifier topology is not tested. This is beyond the scope of the thesis.

Front-end rectifier followed by a boost converter topology is selected to simulate and compare with a proposed single transistor unity power factor rectifier expressed in chapter 4 since the front-end rectifier followed by a boost converter topology's mains supply is single phase supply and it is much more available than three phase mains supply, it has simple control circuit using only one PFC integrated circuit, and it can give necessary output voltage and power.

CHAPTER 4

PROPOSED SINGLE PHASE SINGLE SWITCHING TRANSISTOR UPF RECTIFIER TOPOLOGY

4.1 Introduction

Front-end rectifier followed by a boost converter configuration in figure 2.6 uses single switching transistor and it has simple control circuit, but its conduction loss higher than low conduction loss AC/DC unity power factor rectifer in figure 2.8 while it uses two switching transistors. On the other hand, proposed single phase single switching transistor unity power factor rectifier has simple control circuit and lower conduction loss than front-end rectifier followed by a boost converter and the low conduction loss AC/DC unity power factor rectifier in figure 2.8.

The theory behind proposed single switching transistor low conduction loss unity power factor rectifier is that one switching power transistor can be used instead of two switching power transistor in the low conduction loss AC/DC unity power factor rectifier since the two transistors are turned on an off at the same time. In addition, the proposed single switching transistor low conduction loss unity power factor rectifier should be controlled with a standart unity power factor rectifier integrated circuit.

In this chapter, a new single switching transistor unity power factor topology with low conduction loss is proposed. Its operation principle and analytic analysis are given in the following sub-chapters in this chapter.

4.2 Proposed Single Phase Single Switching Transistor UPF Configuration

A single phase, single switching transistor unity power factor rectifier topology is proposed and its operation principle is given in this sub-chapter. It has one single phase diode bridge, two boost inductors, two ultrafast diodes, one output capacitor and single switching transistor. Actually, one boost inductor can be used instead of two boost inductor since these two boost inductors are serially connected to each other with respective to the mains supply. Two boost inductors whose total inductance is same as the boost inductor are used for making the topology symmetric with respective to mains supply. The proposed topology is shown in figure 4.1.



Figure 4.1 Proposed single phase single switching transistor UPF rectifier.

Operation principle of proposed low conduction loss UPF rectifier configuration in figure 4.1 described below:

 1^{st} step: When switching transistor is on and mains supply's voltage phase is between 0 and 180° , current flows through inductor L1, D5, transistor, diode D4 and inductor L2. Since transistor is on, it is seen that these inductors are connected to supply directly and they begin to store energy. Topology looks like in figure 4.2 at this time.



Figure 4.2 Equivalent circuit when switching transistor on and mains supply's voltage phase is between 0 and 180° .

 2^{nd} step: Switching transistor is off and mains supply's voltage phase between 0 and 180^{0} , current i flows through L1, D1, C1, D4, L2 and the mains supply. In this way, inductor L1, L2 start to give stored energy to load and capacitor C1.



Figure 4.3 Equivalent circuit when switching transistor is off and mains supply's voltage phase is between 0 and 180° .

 3^{rd} step: When mains supply's voltage phase is between 180 and 360^0 and the switching transistor is on, current flows through L2, D6, the switching transistor, D3, L1, and the inductors store energy. However, current direction and inductor voltage polarity is reversed. The topology looks like as in figure 4.4 at this time.



Figure 4.4 Equivalent circuit when mains supply's voltage phase is between 180 and 360° , switching transistor is on.

4th step: When mains supply's voltage phase is between 180 and 360⁰ and the switching transistor is off, current flows through D2, C1, D3 and L1. The topology looks like in figure 4.5 now.



Figure 4.5 Equivalent circuit when mains supply's voltage phase is between 180 and 360° and switching transistor is off.

The boost inductors, L1 and L2, store electrical energy in 1^{st} and 3^{rd} step while they destore and give the electrical energy to the output capacitor, C1 and the load in 2^{nd} and 4^{th} step. In this way, the proposed topology achieves boost operation.

4.2.1 Comparison Of The Proposed Rectifier With Front-End Rectifier Followed By A Boost Converter And Low Conduction Loss AC/DC UPF Rectifier

Proposed unity power factor rectifier is compared with the front-end rectifier followed by a boost converter in figure 2.6 and the low conduction loss AC/DC unity power factor rectifier in figure 2.8 on the basis of efficiency, number of semiconductors. Comparisons are made roughly here. The proposed rectifier and front-end rectifier followed by a boost converter are compared in detail in chapter 6 by making simulations.

Proposed rectifier topology in figure 4.1 is better than front end rectifier followed by a boost converter in figure 2.6 with respect to following characteristics:

- 1. In the proposed topology, the boost diode, D in figure 2.6 is eliminated. In front-end rectifier followed by a boost converter, two bridge diode is on when the transistor is on as in the proposed topology. Two bridge diode and the boost diode, D in figure 2.6 is on in the front- end rectifier followed by a boost converter while only one bridge diode and one boost diode, D1 or D2 is on according to the mains supply's polarity when the transistor is off. Hence, proposed rectifier's conduction loss is less than the front-end rectifier followed by a boost converter topology is redrawn in figure 4.6. Its equivalent circuit when the transistor is on and off is shown in figure 4.7 and figure 4.8.
- 2. In the proposed rectifier topology, inductor core resetting is achieved naturally due to reversing of input current direction at every half cycle of the input voltage, so free-wheeling diode is not necessary and it can not be used with parallel to boost inductors, L1 and L2. In front-end rectifier followed by a boost converter, one free wheeling diode has to be used and connected
parallel to boost inductor, L_S and the boost diode, D in figure 2.6.

Although proposed topology has the mentioned advantages, the bridge diodes, D3, D4, D5, D6 in proposed topology have higher voltage stresses than the bridge diodes in front-end rectifier followed by a boost converter since the boost inductors L1 and L2 are connected between mains supply and the bridge diodes in proposed topology. The bridge diodes D3, D4, D5, D6 in proposed topology will have voltage stresses same with switching transistor in figure 4.1 since the diode bridge is connected parallel with the transistor.



Figure 4.6 Front-end rectifier followed by a boost converter topology.



Figure 4.7 Front-end rectifier followed by a boost converter's equivalent circuit when the switching transistor is on (a) mains supply's voltage phase is between 0 and 180° , (b) mains supply's voltage phase is between 180 and 360° .



Figure 4.8 Front-end rectifier followed by a boost converter's equivalent circuit when the switching transistor is off (a) mains supply's voltage phase is between 0 and 180° , (b) mains supply's voltage phase is between 180 and 360° .

Performance of the proposed rectifier topology and low conduction loss AC/DC UPF rectifier in figure 2.8 are compared, their advantages and disadvantages are listed:

 Proposed rectifier scheme uses only one switching transistor while other one uses two switching transistor (the three transistor ratings are nearly equal). Proposed UPF rectifier has one diode bridge but the other one don't have diode bridge, it uses two ultrafast diodes and two switching transistors' freewheeling diode as diode bridge. 2. In the rectifier in figure 2.8(a), two switching transistor turned on and off simultaneously. When the transistors are on, input current flows through one transistor such as an IGBT and the second transistor's freewheeling diode. When the transistor is on in proposed topology, input current flows through two bridge diodes and one transistor. At this time, the rectifier in figure 2.8 has low conduction loss but the difference is small since freewheeling diodes have relatively higher voltage drop than bridge diodes in general. On the other hand, the transistors in figure 2.8 are off, input current flows through the boost diode, D1 or D2 and one of the transistor's free-wheeling diode in low conduction loss AC/DC unity power factor rectifier while the transistor in figure 4.1 is off, the current flows through boost diode D1 or D2 and bridge diode D4 or D3 depending on the mains supply's polarity in proposed topology. Hence, proposed topology has lower conduction loss than the ones in figure 2.8 since the transistors' freewheeling diode have generally higher conduction voltage drop than the bridge diode's conduction voltage drop.

The rectifier topology in figure 2.8 is redrawn in figure 4.9 and its equivalent circuits when the transistors are on and off are shown in figure 4.10 and 4.11.



Figure 4.9 Low conduction loss AC/DC UPF rectifier topology.



Figure 4.10 Low conduction loss AC/DC UPF rectifier's equivalent circuit when the switching transistor is on (a) mains supply's voltage phase is between 0 and 180° , (b) mains supply's voltage phase is between 180 and 360° .



Figure 4.11 Low conduction loss AC/DC UPF rectifier's equivalent circuit when the switching transistor is off (a) mains supply's voltage phase is between 0 and 180⁰, (b) mains supply's voltage phase is between 180 and 360⁰.

In order to understand whether proposed unity power factor rectifier topology or the one in figure 2.8 has lower conduction loss, the rectifier topology specifications and diodes and switching transistors used in implementation have to be known. In unity power factor rectifier implementation part in this thesis, HGTG20N60A4D IGBT, RURG5060 ultra fast diode, VBO30-12N07 diode bridge are used as switching transistor, boost diode and single-phase diode bridge respectively. Also, the boost rectifier implemented has nominal 220 V rms input voltage, 440 V output voltage and 3 kW output power. These specifications are given in chapter 6. Equation 2.2 is applicable to front-end rectifier followed by a boost converter in figure 2.6, low conduction loss AC/DC unity power factor rectifier in figure 2.8 and proposed rectifier topology since they are all boost rectifier topologies and their operation principle is similar. Average duty ratio can be calculated using equation 2.2 and putting rectified input voltage's average value instead of input minimum voltage to this equation as input voltage, and it is 0.55. The rectifiers' efficiencies are assumed as 100 % to simplify the semiconductors' conduction loss calculations. The rectifiers input current can be calculated using equation 4.1.

$$I_{in(rms)} = \frac{P_{in}}{V_{in(rms)}} \tag{4.1}$$

, where $V_{in(rms)}$ is the rectifiers' input voltages, and P_{in} is their input voltage. Then, the rectifiers' input currents are nearly 13.64 A rms. HGTG20N60A4D IGBT and its free-wheeling diode have 1.6 V, 1.8 V conduction voltage drop while their junction temperatures and currents are 125 °C, 13.64 A. RURG5060 ultra fast diode has 0.9 V conduction voltage drop when its junction temperatures and currents are 100 °C, 13.64 A. One diode in VBO30-12N07 diode bridge has 1 V conduction voltage drop when its current is 13.64 A. Proposed rectifier topology and the other ones' semiconductor conduction loss can be calculated using the equation 4.2 when the transistors are on, the equation 4.3 when they are off.

$$Conduction loss = Semiconductors voltagedrop \times rmscurrent \times D$$
(4.2)

$$Conductionloss = Semiconductorsvoltagedrop \times rmscurrent \times (1 - D)$$
(4.3)

The rectifier in figure 2.8 has nearly 25.5 watts when the transistors are on, 16.57 watts while the transistors are off, and so total semiconductor conduction losses are 42.07 watts. Proposed rectifier has 27 watts semiconductor conduction losses when the switching transistor is on and 11.66 watts when it is off. Hence, its total

semiconductor conduction losses are 38.66 watts.

Proposed unity power factor rectifier topology has lower semiconductor conduction losses than the ones in low conduction loss AC/DC unity power factor rectifier in figure 2.8 when HGTG20N60A4D IGBT and RURG5060 ultrafast diode are used as power semiconductors in these topologies. IGBTs' free-wheeling diodes have to be very fast and their conduction voltage drop has generally a higher value than bridge diodes and boost diodes which are also ultrafast diode but not needed to be so fast as much as IGBTs' free-wheeling diodes. Low conduction loss AC/DC unity power factor rectifier in figure 2.8 uses IGBTs' freewheeling diodes to conduct the current, i in figure 4.10 and 4.11. Therefore, proposed rectifier has higher efficiency than the one in figure 2.8 considering the semiconductor conduction losses.

4.2.2 Analysis Of The Proposed UPF Rectifier Analytically

Proposed unity power factor rectifier topology's inductance and capacitance values are investigated analytically in this section. Expressions to its output capacitance and boost inductance value are derived.

Basic relationship formula of the inductor is

$$V_{LS} = L\frac{di}{dt} \tag{4.4}$$

The voltage over the inductor in figure 4.1 equal to $V_{LS} = V_{in} - V_o$.

1st stage: When the power transistor is on and $0 \le V_{in} < 180^{\circ}$,

$$V_{in} = L \frac{di}{dt} \Rightarrow L di = V_{in} dt$$

Equation 4.5 is obtained by integrating two sides of the above equation.

$$\int_{0}^{\Delta I} Ldi = V_{in}T_{1}$$

$$L\Delta I = V_{in}T_{1} \implies T_{1} = \frac{L\Delta I}{V_{in}}$$
(4.5)

2nd stage: When the power transistor is off, $0 \le V_{in} < 180^{\circ}$,

$$L\frac{di}{dt} = V_{in} - V_O$$

Integrating two sides of the above equation, equation 4.6 is derived.

$$\int_{\Delta I}^{0} L di = \int_{T_1}^{T_s} (V_{in} - V_O) dt$$

- $L \Delta I = (V_{in} - V_O) (T_s - T_1)$ (4.6)

In order to remove T_1 from the equations, T_1 in equation 4.5 is placed in to equation 4.6. Then, equation 4.7 is derived.

$$-L\Delta I = (V_{in} - V_O)(T_S - \frac{L\Delta I}{V_{in}})$$

$$\Rightarrow L = V_{in(\min)} \frac{(V_O - V_{in(\min)})}{V_O} / (f_S \Delta I)$$
(4.7)

Equation 4.7 gives the proposed rectifier's boost inductance value. L1 and L2 in figure 4.1 have an inductance value, which is half of the inductance value calculated using equation 4.7 since L1 and L2 are serially connected to each other.

Using "capacitor energy equation" given below, derivation of equation 4.8 is made.

$$E = \frac{CV^2}{2}$$

In above equation, E is the energy stored in C and V is the voltage over C.

$$E_1 = \frac{CV_o^2}{2}$$
, where V_o is the rectifier's desired output voltage.
 $E_2 = \frac{CV_{O\min}^2}{2}$, where $V_{O\min}$ is the rectifier's permissible minimum output

voltage. Taking difference of E_1 and E_2 , the following equation is obtained.

$$E_1 - E_2 = P_{out} \Delta t$$

In this equation, P_{out} is the rectifier's output power and Δt is hold-up time. Using above last three equations, equation 4.8 is derived.

$$C = \frac{2P_{ound}\Delta t}{V_0^2 - V_{O\min}^2}$$
(4.8)

Longer hold-up time, Δt chosen in a design can tolerate supply interruptions with short duration at the expense of bulky and costly capacitor banks. A

compromise can be made between these factors [2].

Equation 4.7 and 4.8 are same with equation 2.2 and 2.4, so the output capacitances and boost inductances in the proposed rectifier and the popular frontend rectifier followed by a boost converter can be calculated with these equations. Also, they are applicable to the low conduction loss AC/DC unity power factor rectifier in figure 2.8.

4.3 Result

Proposed unity power factor rectifier topology is investigated analytically, and expressions given its boost inductance and output capacitance values are derived. These expressions are directly applicable to the front-end rectifier followed by a boost converter in figure 2.6 and the low conduction loss AC/DC unity power factor rectifier in figure 2.8.

The low conduction loss AC/DC unity power factor rectifier uses two switching transistors while the proposed rectifier and the front-end rectifier followed by a boost converter use only one switching transistor. However, number of diodes used in the low conduction loss AC/DC unity power factor rectifier is less than the two unity power factor rectifier topologies. The proposed rectifier uses two boost diode but one conducts in mains supply's one half cyle, the other one in the other half cycle. On the other hand, the front-end rectifier followed by a boost converter needs free-wheeling diode parallely connected to boost inductor and boost diode to achieve the boost inductor's core resetting. Thus, total power diode number including bridge diode, boost diode and free-wheeling diode is same in these two rectifier topologies. Bridge diodes' voltage stresses in proposed topology are higher than the front-end rectifier followed by a boost converter.

Proposed topology has lower semiconductor conduction losses than the other two rectifier topologies. It is chosen to implement since it has single switching power transistor as the front-end rectifier followed by a boost converter and has higher efficiency considering the semiconductor conduction losses. Front-end rectifier followed by a boost converter and proposed topology are selected to compare with each other since they have single switching transistor and front-end rectifier followed by a boost converter is more popular topology than the one in figure 2.8, so they are investigated in detail by making simulations in chapter 6.

The classical unity power factor rectifier, called front-end rectifier followed by a boost converter and proposed rectifier's advantages and disadvantages are summarized as the following by comparing with each other:

1. Classical UPF rectifier:

Advantages:

- Requires only one ultrafast boost diode.
- Needs a diode bridge had lower voltage stress.

Disadvantages:

- Has higher conduction losses.
- Needs a free-wheeling diode parallely connected to boost inductor and boost diode to achieve the boost inductor's core resetting.

2. Proposed UPF rectifier:

Advantages:

- Has lower conduction losses.
- The free-wheeling diode is not needed.

Disadvantages:

- Needs two ultrafast boost diodes.
- Needs a diode bridge had higher voltage stress.

CHAPTER 5

CONTROL TECHNIQUES FOR UNITY POWER FACTOR RECTIFIERS

5.1 Introduction

Standard bridge rectifier followed by output capacitor result in low efficiency, higher total harmonic distortion and higher reactive power and lower active power with different output loads that can be resistive, inductive, capacitive, and switching loads such as an inverter. In order to develope interface systems that improves the power factor of standard electronic loads, and improve the quality of the currents, there are many efforts being done and various control techniques.

The boost topology shown in figure 5.1 together with power factor corrector controller is the most popular topology in PFC (power factor corrector) applications. The diode bridge rectifier is used for AC/DC conversion, and the controller operates to properly shape the input current i_g with respect to its current reference. The output capacitor absorbs the input power pulsation and allows small output voltage ripple. The boost topology can operate with different control techniques allowing low distorted input current waveforms and achieve nearly unity power factor.

There are various control schemes such as peak current control, average current control, hysteresis current control, borderline control, discontinuous current PWM (pulse width modulation) control [33], [35], nonlinear carrier control (NCC), input current shaping, charge and quasi-charge control [34]. In this chapter, the most popular control techniques that are peak current control, average current control, hysteresis current control, borderline control, discontinuous current PWM control



Figure 5.1 Principle scheme of a boost PFC.

are reviewed and compared on the basis of the boost topology in order to express advantages and drawbacks of each solution.

5.2 Popular PFC Control Techniques

Unity power factor rectifiers can be used in "Continuous Conduction Mode", "Discontinuous Conduction Mode", and boundary between these conduction modes. "Continuous Conduction Mode" is used in high and middle power applications while "Discontinuous Conduction Mode" is used in low power applications generally.

In "Discontinuous Conduction Mode", current amplitude goes zero and stays zero then it rises. It means that the current is not continuous whereas in "Continuous Conduction Mode", the current magnitude doesn't reach zero. On the other hand, in "Boundary Conduction Mode", the current amplitude goes zero but don't stay at zero current level and it rises immediately. The current waveforms in these conduction modes are shown in figure 5.2.



Figure 5.2 Current Waveforms in "Discontinuous Conduction Mode", "Continuous Conduction Mode", "Boundary Conduction Mode".

Peak current control, average current control, hysteresis current control work in "Continuous Conduction Mode" but borderline control in "Boundary Conduction Mode". Discontinuous current PWM control works in "Discontinuous Conduction Mode". Their current waveforms are listed in table 5.1.



 Table 5.1 Current waveforms in popular current control techniques.

These control techniques are considered here on the basis of the boost PFC (boost power factor corrector). However, these control techniques can be applied to other topologies [33], [35].

5.2.1 Peak Current Control

The basic peak current control scheme is shown in figure 5.2. The switch is turned on by a clock signal at the beginning of each cycle. It is turned off when the inductor current (if there is external ramp (compensating ramp), sum of the inductor current and external ramp) reaches sinusoidal current reference [33], [34], [35]. This sinusoidal current reference is generally obtained by multiplying the output of the voltage error amplifier and the scaled replica of the rectified line voltage. The voltage error amplifier sets the current reference amplitude. The current reference signal is always proportional to and synchronized with the line voltage, so unity power factor is achieved. The converter operates in continuous inductor current mode (CICM) as shown in figure 5.3. Continuous inductor current mode provides reduced devices current stress and input filter requirements [33]. The bridge diodes can be slow devices with continuous input current while the boost diode has to be fast diode.



Figure 5.3 Peak Current Control Scheme.

Advantages and disadvantages of peak current control method are listed here [33]:

Advantages:

- Constant switching frequency,
- Only the switch current must be sensed and this can be accomplished by a current transformer, thus avoiding the losses due to the sensing resistor,
- No need of current error amplifier and its compensation network,
- Possibility of a true switch current limiting.

Disadvantages:

- Presence of subharmonic oscillations at duty cycles greater than 50%, so a compensation ramp is needed,
- Input current distortion which increases at high line voltages and light load is worsened by the presence of the compensation ramp,
- Control more sensitive to commutation noises.

Changing the current reference waveshape, the input current distortion can be reduced and better input current waveform can be obtained. Also, the compensation ramp can be avoided by keeping the duty cycle below 50% if the power factor corrector is not designed for universal input operation [33].

5.2.2 Average Current Control

Average current control scheme shown in figure 5.4 has better input current waveform. In this control method, the inductor current is sensed and filtered by the current error amplifier. The current error amplifier's output drives a PWM (pulse width modulation) modulator. The inner current loop tends to minimize the error between the input current and the current reference [33], [35]. The current reference signal is obtained in a similar way as in the peak current control [33], [34]. The rectifier with average current control method works also in continuous inductor current mode. Hence, the same considerations done with regard to the peak current control has the following advantages and disadvantages [33], [35]:

Advantages:

- Constant switching frequency,
- No need of compensation ramp,
- Control is less sensitive to commutation noises, due to current filtering,
- Better input current waveforms than for the peak current control since the duty cycle is close to one at near the zero crossing of the line voltage, so reducing the dead angle in the input current.

Disadvantages:

- Inductor current must be sensed,
- A current error amplifier is needed and its compensation network design must be taken into account with the different converter operating points during the line cycle.



Figure 5.4 Average Current Control Scheme.

5.2.3 Hysteresis Control

Hysteresis current control scheme is shown in figure 5.5. Two sinusoidal current references namely $I_{P,ref}$, which is the peak current reference for the inductor current, $I_{V,ref}$, which is the current reference for the valley of the inductor current are generated in this type of control. The switch is turned on when the inductor current goes below the lower current reference, $I_{V,ref}$ and it is turned off when the inductor current which the converter also works in CCIM (Continuous Inductor Current Mode)[33].

Advantages:

- No need of compensation ramp,
- Low distorted input current waveforms.

Disadvantages:

- Variable switching frequency,
- Inductor current must be sensed,
- Control sensitive to commutation noises.

Its advantages and disadvantages [33] are given above. This current control technique has very high control stability and robustness [35].



Figure 5.5 Hysteresis Control Scheme.

5.2.4 Borderline Control

Borderline control scheme is shown in figure 5.6. The switch on-time is held constant during the line cycle and switch is turned on when the inductor current falls to zero. In this approach, the converter operates at the boundary between CCM ("Continuous Conduction Mode") and DCM ("Discontinuous Conduction Mode"). In this way, the boost diode is turned off softly, so there is no recovery loss of the boost diode. The switch is turned on at zero current, and this reduces the commutation losses of the switch. However, the higher current peaks increases the conduction losses and the device stresses. This may result in heavier input filters for some topologies. The instantaneous input current has triangular waveshape, whose peak is proportional to line voltage [33], [35]. In this way, the average input current of the rectifier becomes proportional to the line voltage without duty-cycle modulation. The switch current sensing can be eliminated by modulating the switch-on time with respect to the voltage error amplifier's output without using the multiplier [33].



Figure 5.6 Borderline Control Scheme.

Advantages:

- No need of a compensation ramp,
- No need of a current error amplifier,
- For controllers using switch current sensing, switch current limitation can be introduced.

Disadvantages:

- Variable switching frequency,
- Inductor's voltage must be sensed in order to detect the zeroing of its current,
- For controllers in which the switch current is sensed, control is sensitive to commutation noises [33].

5.2.5 Discontinuous Current PWM Control

Discontinuous current PWM control scheme is shown in figure 5.7 and it eliminates the internal current control loop to operate the switch at constant on time and switching frequency. This control technique causes some harmonic distortion in the line current with the boost PFC, but allows unity power when it is used with converter topologies like Flyback, Cuk and Sepic with the converter operating in "Discontinuous conduction mode" (DCM) [33], [35].



Figure 5.7 Discontinuous current PWM control scheme.

Advantages:

- Constant switching frequency,
- No need of current sensing,
- Simple PWM control.

Disadvantages:

- Higher device current stress than for borderline control,
- Input current distortion with boost topology [33].

CHAPTER 6

SIMULATIONS OF PROPOSED CONVERTER AND CLASSICAL UNITY POWER FACTOR RECTIFIER

6.1 Introduction

Control techniques for UPF (unity power factor) rectifiers are discussed in chapter 5. Discontinuous current PWM control has high input current distortion with boost topology. Proposed converter and classical converter are boost converter topologies, so discontinuous current PWM control is not much convenient for the converter topologies. Peak current control, hysteresis control, borderline control techniques are sensitive to commutation noise. Hysteresis control and borderline control techniques have variable switching frequency, but peak current control has fixed switching frequency. However, average current control technique is less sensitive to commutation noise, has fixed switching frequency and better input current waveforms than peak current control [33]. Thus, average current control technique is chosen for the implementation of the proposed converter and classical converter's simulations. Several integrated circuits exist in the market that use average current control technique such as UC3854A/B, UC3855, TK3854A, ML4821, TDA4815, TDA4819 etc. UC3854B (UC1854B, UC2854B, and UC3854B are only different from each other in thermal characteristics) is chosen as PFC IC (Power Factor Corrector Integrated Circuit) used average current control technique.

Average current control technique, and circuit scheme and necessary calculations are given in the Unitrode application note "UC3854 Controlled Power Factor Correction Circuit Design" in Appendix A and related information about rectifier design with UC3854B is found reference [42].

In the design procedure, the rectifiers' specifications used are as follows:

- 1. Pout = 3 kW
- 2. Output voltage = 440 V
- 3. Output voltage ripple = 10 %
- 4. Input voltage, $V_{in} = 220 15 \% + 30 \% V rms$
- 5. Line frequency = 50 Hz
- 6. Switching frequency, $f_{sw} = 50 \text{ kHz}$
- 7. Maximum input current ripple, $\Delta I = 10$ % peak to peak
- 8. Efficiency, $\eta \ge 95$ %
- 9. Input power factor, $pf \cong 1$

In this chapter, it is decided which simulation program and the IC model in simulation program are used for the simulations made. In these simulations, inductors used are ideal inductors and snubbers for power semiconductors in the simulation circuits are not used. Afterwards, power semiconductors to be used in the rectifiers' implementation stage are selected since their parameters are used in the simulations for comparing the classical and proposed rectifier's performances here.

Then, the followings are decided through this chapter:

- 1. Whether the proposed rectifier can be controlled with the UC3854B IC or not is decided.
- 2. Do the designed proposed rectifier and classical rectifier's specifications coincide with the specifications given above via simulations made with the selected simulation software? If not, the rectifiers' circuit will be redesigned.
- 3. The proposed rectifier and the classical rectifier's performances are investigated at steady state and in dynamic conditions. Also, their power semiconductor stresses are investigated and compared via the simulations. It is decided whether the proposed rectifier will be implemented or not according to the results. These are the most important decision and purposes in this chapter.

The two rectifiers' performances are compared in the following ways:

 Their performances are investigated at steady state by connecting different 3 kW loads to their outputs in each simulation. The followings are used as the 3 kW load:

- 3 kW purely resistive load
- Two different inductive resistive loads. One of them is 10 uH inductor serially connected to a resistor and the other is 100 uH inductor serially connected to the resistor.
- 2. The two rectifiers are simulated in dynamic conditions. One purely resistive 3 kW load is connected to the proposed rectifier's output, after a while it is disconnected and then connected again in the simulation environment. The classical unity power factor rectifier's performance in dynamic conditions is investigated similarly.

In chapter 7, whether the simulations made in this chapter and its results are correct or not is investigated by checking simulations and implementation results with 1 and 2 kW purely resistive load connected each time to proposed rectifier's output in simulation and laboratory environment.

6.2 Choice Of The Simulation Software And The Simulation Models

At this stage it is very important to determine which simulation medium is most suitable for the purpose here. This is very important as there will be a large number of simulations and the time taken for a simulation should be short while the accuracy is good. In this section, various alternatives for simulation shall be evaluated.

For simulation of electronic circuits, PSPICE is a common choice. This software is best if details on device level are important, but the solution time may be excessive for studying long duration behaviour of electronic circuits. The other alternatives available are MATLAB-Simulink, PSIM, a combination of PSIM and Simulink and SIMPLORER. Here, only the first 3 options are studied.

In this section, PSIM with UC3854B unity power factor IC model using discrete components, PSIM with UC3854B model already found in PSIM model library, PSIM and MATLAB-Simulink R14 with UC3854B model found in PSIM model library, MATLAB-Simulink R14 with UC3854B IC model using discrete blocks in Simulink are used for simulating the unity power factor rectifier and

compared especially for their simulation speed.

UC3854B IC model drawn using discrete components in PSIM, the same IC model already in PSIM, and the IC model drawn using discrete components in Simulink R14 are shown respectively in figure 6.1, 6.2, 6.3. Working principle of the IC is explained in section 6.4 by using classical and proposed rectifier topologies and their control circuits' block diagram given in figure 6.7 and figure 6.8. Hence, detailed description about working principle of the IC simulation models is not given in this section.



Figure 6.1 UC3854B model created using discrete components for PSIM 6.1 simulation program.

UC3854B IC model in figure 6.2 is found already in PSIM 6.1 as stated above. In order to increase the simulation speed, this model is simplified but the model's main characteristics are maintained. In this way, the model in figure 6.1 is obtained. The modifications are:

- PKLMT, VCC, CT, SS, RSET, ENA, VREF pins and their related circuit are removed from the model in figure 6.2 since they are unnecessary to make simulations at steady state and the simulations made in this chapter.
- Setting of the PWM signal frequency, which is the switching frequency, f_{sw} is achieved by using CT, RSET, their related circuits and an extra circuit in UC3854B model in figure 6.2 while it is achieved via "triangle voltage source" whose frequency is at the switching frequency.

				UC38:	54B
1:	GND	16:	GTDRV		
2:	PKLMT	15:	VCC	1	36 0
3:	CAOUT	14:	CT	l	15 Î
4:	ISENSE	13:	SS	j,	12 0
5:	MOUT	12:	RSET	, s	12 0
6:	IAC	11:	VSENSE	44	11 0
7:	VAOUT	10:	ENA	4 7	10 k
8:	VRMS	9:	VREF	4.	9 0

Figure 6.2 UC3854B model in PSIM.

The IC model in figure 6.3 is similar to the one in figure 6.1, Simulink R14 has no opamp and comparator model, so subtract and PID blocks are used instead of the opamps and the comparator in figure 6.1. The "triangle voltage source" in figure 6.1 is replaced by "repeating sequence" in figure 6.3 since the "repeating sequence" in Simulink R14 is the counterpart of "triangle source" in PSIM 6.1. On the other hand, the model in figure 6.1 and the one in figure 6.3 have same functions and they are sufficient to study the performance of the proposed and classical unity power factor rectifier.



Figure 6.3 UC3854B model drawn using discrete blocks in Simulink R14.

In figure 6.1, 6.2 and 6.3; VSENSE, IAC, VAOUT, VRMS, MULTOUT, ISENSE, CAOUT, GTDRIVE are the UC3854B's pins and detailed descriptions are given in its datasheet in Appendix B.

In order to compare the PSIM and Simulink simulation programs, classical unity power factor rectifier shown in figure 6.7 is simulated by using International Rectifier IRGP50B60PD IGBT and ST BYT30PI800 power diode's parameters at steady state. The IGBT and diode are chosen arbitrarily to use their parameters in simulations to compare their speed. The diode block forward voltage drop parameter is taken as 1.9 V; it is the voltage drop over ST BYT30PI800 power diode in conduction. Snubber resistance parameter in the diode block used in the simulations is taken as 500 k Ω in Simulink. The diode block's other parameters are not altered. In PSIM simulation, only the IRGP50B60PD and BYT30PI800 power semiconductor's forward voltage drop parameter is used since current fall time, rise time, tail time for the IGBT and transit time for the diodes can not be entered in PSIM's IGBT and diode model. The rest of the block parameters are taken as the default values.

The rectifier is simulated by using MATLAB-Simulink, PSIM 6.1 simulation programs, PSIM 6.1 and MATLAB-Simulink R14 co-simulation with 3 kW purely

resistive load at steady state.

In PSIM and MATLAB-Simulink co-simulation, the UC3854B model in figure 6.1 is used in PSIM, but diode and IGBT blocks are used in Simulink. Simulink's accelerator mode is used for increasing the simulation speed. In the co-simulation, Simulink doesn't permit the use of fixed step solver in Simulink side due to IGBT block used in it. According to PSIM user manual, in co-simulation, fixed step solver should be used to obtain correct results, otherwise zero-order-hold blocks must be placed at the Simcoupler model block's input, and it is what is done in the simulation. Fixed step size is used and the time step is taken as 1E-07 s in PSIM side while variable step size is used in Simulink side.

In MATLAB-Simulink simulation, simulation is done in the accelerator mode and the UC3854B in figure 6.3 is used. Variable step size is used in the simulation.

Two simulations are made with the PSIM simulation program. The UC3854B IC model in figure 6.1 is used in one simulation while the model in figure 6.2 is used in the other one. Also, the time step is taken as 1E-07 s and fixed step size is used for the two simulations since only fixed step size can be used in PSIM simulation program.

The simulations' results are given in table 6.1 and table 6.2. The results of the simulations made by using PSIM 6.1 with the UC3854B IC model in figure 6.1 and figure 6.2 are given in table 6.1 while table 6.2 shows the results of simulations made with MATLAB-Simulink R14, PSIM and MATLAB co-simulation. The simulations of PSIM 6.1 with the UC3854B IC model in figure 6.1 and figure 6.2, MATLAB-Simulink R14 with the IC model in figure 6.3 gives similar results. On the other hand, the PSIM and MATLAB co-simulation predicts the IGBT and boost diode's power loss extremely higher than the others, so this simulation result is relatively unrealistic compared to the others.

The simulations are made by using a computer, which has 256 MB RAM and Pentium 4 1.7 GHz CPU. Investigating table 6.1, fastest simulation speed is achieved using the PSIM 6.1 with UC3854B IC model in figure 6.1. Hence, PSIM 6.1 with this IC model is used for the rectifiers' simulation due to its highest speed in the rest of this chapter.



Table 6.1 Results of classical unity power factor rectifier's simulation made by using PSIM 6.1 with the UC3854B IC model in figure 6.1, figure 6.2 at steady state with 3 kW purely resistive load and comparison of the simulations' speed.

	PSIM MATLAB co-simulation (with	MATLAB (with UC3854B	
	UC3854B model in figure 6.1)	model in figure 6.3)	
Input power	3300 VA	3122.6 VA	
Input power			
factor	0.98927	0.9898	
Output power	3022.46 watt	3000 watt	
Input current	25 A peak, 20.10 A peak, 13.5 A (average)	23.9 A peak, 19.95 A peak	
Input current waveform	80.00 20.00 0.00 0.00 0.00 20.00 30.00 375.0080.0085.0090.0095.000 0.0	20 10 0 10 -20 -20 -20 -20 -20 -20 -20 -2	
Output voltage	432.97 V-450.23 V	431.0 V-448.7 V	
	00.00 00.00 00.00 00.00 00.00 00.00 0.00 0.00 0.00 0.25 0.30 0.35 0.40 0.45 0.40	450 400 350 250 250 150 100 50 0 0 0 0 0 0 0 0 0 0 0 0 0	
Output voltage			
waveform	11.0000	11.165	
D1 power loss	11.9809 Watt	11.165 watt	
D2 power loss	11.9623 Wall	11.105 Watt	
D3 power loss	11.7/00 Wall 11.9808 watt	11.105 Wall	
Boost diode nower	11.7000 watt	11.105 watt	
	85 714 watt	13 115 watt	
IGBT power loss	100.09 watt	11.95 watt	
	Step size in PSIM = 1e-7	Step size in Matlab = auto	
	Step size in Matlab = auto	Blocks' step size in Matlab=5e-	
Step size	Blocks' step size in Matlab = 5e-7	7	
Solution time	0.5 s	0.5 s	
Total simulation			
time	45 minute	2.5-3 hour	

Table 6.2 Results of classical unity power factor rectifier's simulation made by usingMATLAB-Simulink R14, PSIM 6.1 and Simulink co-simulation at steady state with3 kW purely resistive load and comparison of the simulations' speed.

6.3 Component Selection

The classical unity power factor rectifier and proposed rectifier are simulated using PSIM 6.1 simulation program with the UC3854B IC model in figure 6.1 and simulation results are given in the following sections. In order to use IGBT and diode parameters in these simulations, International Rectifier, Fairchild Semiconductor, ON Semiconductor, IXYS' IGBTs and diodes are compared.

ON Semiconductor company doesn't have IGBT whose absolute maximum voltage stress is 600 V or higher. Consequently there is no ON Semiconductor IGBT in table C.1 in Appendix C to compare. The IGBT in the rectifier in this project is hard switched at 50 kHz, so the selected IGBT have to be capable of operating at least at 50 kHz hard switching operation. Also, the rectifier should give 3000 W output power and its input current can be as much as 28 A while mains supply permissible minimum voltage is 187 Vrms. The rectifier's average output voltage is 440 V. There is an output capacitor rated 450 V nominal, 500 V maximum at its output side. These prevents voltage between the IGBT's collector and emitter terminal to exceed 500 V; so an IGBT whose absolute maximum voltage stress is 600 V and it is enough in these rectifier circuits. While maximum permissible current in the rectifier circuit is 28 A, the transistor current rating should be higher than 28 A due to possible failure or unknown spikes in the circuit. Figure 6.4 shows that maximum voltage seen between the IGBT's collector and emitter terminal is 450 volts, and its collector current is 28 A. Its current rating should be nearly 1.5 times its maximum collector current due to mentioned reasons, then it is 42 A. IXSH30N60B2D1 and IRG4PC50UD has higher fall time than the other IGBTs in the table C.1 in Appendix C. Comparing IRGP50B60PD, IRGP50B60PD1 and IRG4PC50W-ND; IRGP50B60PD has highest price, IRGP4PC50W-ND has highest saturation voltage drop, fall time and rise time but the lowest price. HGTG20N60A4D has lowest rise time and fall time than the other IGBTs except IRGP50B60PD1 and it has lowest price than the others in table C.1 in Appendix C, so HGTG20N60A4D IGBT is selected to use in the design and the unity power factor rectifier implementation.

Ultrafast diode selection is next issue. In order to determine the ultrafast



Figure 6.4 IGBT's voltage and current waveform while input voltage is 187 V rms.

diode's parameters, one should know that UC3854B PFC IC can operate up to 95 % duty cycle. The PFC boost stage diode's current and voltage waveform are shown in figure 6.5. Also, the voltage and current waveform are given between 395 and 395.5 ms in figure 6.6 to show the waveforms in detail.



Figure 6.5 Ultrafast diode current and voltage waveform when V_{in} is equal to 187 V rms.

The current waveform in figure 6.5 is PWM waveform drawing a nearly rectified sinusoidal waveform boundary. The current waveform has 28A maximum value when mains supply's voltage is 187 Vrms. Due to the maximum 95 % duty cycle, the ultrafast diode's current rating should be determined using the sinusoidal waveform boundary.



Figure 6.6 Figure 6.5's zoomed version.

The ultrafast diodes' average rectified forward current ratings are given in table C.2 in Appendix C. Average value of a sinusoidal waveform is equal to peak value of a sine wave multiplied by $2/\pi$. The rectified sinusoidal current waveform with 28 A maximum value in figure 6.5 has nearly 17.83 A average value. At least 1.5 times the value should be selected to prevent possible spikes or unknown failures and it is 26.7 A. Ultrafast diode with 30 A current rating will be proper for the PFC application.

International Rectifier, Fairchild Semiconductor, ON Semiconductor, IXYS' ultrafast diodes are compared and ultrafast diodes whose voltage and current rating match the criteria in table 6.3 is listed in table 6.4. ISL9R3060P2, 30EPH06, DSEP30-06A/B diodes are faster than the other diodes in table C.2 in Appendix C while RURP3060 is the cheapest ultrafast diode in this table. Properties of some

ultrafast diode models are listed in table 6.4, and there is not so much difference based on their selection criteria, so cheapest one, namely RURP3060 is selected.

Heat Sink Temp	Current Waveform Boundary (Continuous)	Turn On Voltage	Reverse Blocking Voltage	Operating Frequency	Transit Time
	, , , , , , , , , , , , , , , , , , ,	Between 1 V			less than 1
80 Celsius	28 A	and 2.2 V	600 V	50 kHz	us

 Table 6.3 Ultrafast Diode Selection Criteria.

	Turn on		USD Price
Model	Voltage	Transit Time	(Each/1000)
		$I_F = 1A dI_F/dt = 100A/us$	
	$I_{\rm F}$ =30 A T=25 ^o C 1.5 V max,	55 ns max, I_F =30A d I_F /dt=100A/us	
RURP3060	T=150 °C 1.3V max	60 ns max	1.11384
	Version A 1 25 V (I_{E} =30 A T=25		
	^o C 1.6 V max.	$I_{\rm F}=1$ A. di/dt=100A/us. $V_{\rm R}=30$ V.	
	T=150 ^o C 1.25 V max), Version B	$T_{VI}=25 {}^{0}C$	
DSEP30- 1.56 V (I_F =30 A T=25 °C 2.51 V		Version A 35 ns max, Version B	
06A/B/BR	max, T=150 °C 1.56 V max)	30 ns max	1.68
	2.4 V (I _F =30 A T=25 °C 2.1 V		
	typical 2.4 V max,	35 ns (I_F =30 A T=25 0 C 36ns	
ISL9R3060P2	$T=125 \ {}^{0}C \ 1.7 \ V \ typical \ 2.1 \ V \ max)$	typical 110 ns max)	1.581
	2.1 V (I _F =30 A, T=25 ^o C 2.0 V		
	typical, 2.6 V max);	40 ns (T=25 0 C I _F =30 A, trr=31 ns	
	I_F =30 A, T=150 0 C 1.34 V typical	typical; T=125 ${}^{0}C$ I _F =30 A, trr=77	
30EPH06	1.75V max)	ns typical)	1.3427

Table 6.4 Ultrafast diode selection table.

Similar to the IGBT and ultrafast diode selection, International Rectifier, Fairchild Semiconductor, ON Semiconductor, IXYS bridge diodes are investigated to identify bridge diode's parameters to use in the simulations. International Rectifier, Fairchild Semiconductor, ON Semiconductor don't produce bridge diodes as discrete diodes. Also, IXYS company generally does not produce bridge diodes as discrete diodes except some bridge diodes. Bridge diodes have to be discrete diodes to measure the diode's power loss one by one if necessary. Proper discrete diodes for our rectifier is shown in table 6.5. DSI30-08A has higher non-repetitive peak surge current, I_{FSM}, than DSI30-08AC and the rest of their parameters are nearly same, and their prices are same. Hence DSI30-08A diode is selected for the simulations.

		Io	IFSM			
Diode	VRRM	(rectified)	Max	VFM Max		
Code	Max (V)	Max (A)	(A)	(V)	Price	Comment
DSI30-	800	30	210		2.9095 \$	
08AC				1.45 V	(Future	Proper for
				(I _F =45 A,	electronics e-	rectifier
				$T=25^{0}C$	store)	bridge
DSI30-	800	30	300	less than	2.9095 \$	
08A				1.45 V	(Future	Proper for
				(I _F =45 A,	electronics e-	rectifier
				$T=25^{0}C)$	store)	bridge

 Table 6.5 Bridge diode selection table.

6.4 Proposed and Classical Unity Power Factor Rectifiers' Comparison Via Simulations

In this section, performances of the proposed rectifier and classical unity power factor rectifier, which are designed according to the specifications given in section 6.1 are investigated and compared at steady state and in dynamic conditions via simulations as stated before in the section. These simulations are made by using PSIM 6.1 simulation program and UC3854B IC model in figure 6.1 since the simulation program with the IC model in figure 6.1 has the highest simulation speed and enough accuracy as mentioned before in section 6.2. In these simulations, the rectifiers' mains supply is 50 Hz, 220 V rms sinusoidal single phase mains supply.

The proposed rectifier and classical unity power factor rectifier are simulated, their performance is compared on the basis of efficiency, input power factor, input current ripple, input current's THD, output voltage ripple, output voltage stability, etc. The investigation is done at steady state and their steady state input current, voltage waveform and output voltage waveform are given in the following sections by using:

- 1. A 3 kW purely resistive load.
- 2. Two 3 kW inductive resistive loads. These loads are used for understanding whether the rectifiers can maintain their unity power factor operation with different inductive loads. One of the loads has 100 uH inductor serially connected to a resistor and the other 10 uH. The range is chosen large enough to observe the effect of the inductor size.

One purely resistive 3 kW load is connected to the proposed and classical rectifier's output, after a while it is disconnected and then connected again in the simulation environment. In this way, their behaviour, especially their output voltage's settling time, in dynamic conditions are investigated.

PSIM 6.1 simulation program's time step is taken as 1E-7 s between 0 and 1 s in simulation time at their steady state analysis. After that, the time step is decreased to 1E-8 s between 1s and 1050 ms to increase the simulations' accuracy although the 1E-7 s time step is enough for the analysis. On the other hand, the simulation program's time step is taken as only 1E-7 s in the rectifiers' dynamic analysis with the simulations.

In these simulations, no snubbers for the IGBT and the diodes are used while the implemented circuit has R-C and overvoltage snubbers for the IGBT as mentioned in chapter 7. Also, the inductors used in these simulations are ideal inductors.

The selected power semiconductors are given in section 6.3. Only their conduction voltage drop parameters are used in these simulations since the PSIM 6.1 simulation program permits to use only the parameters in the simulations.

The classical unity power rectifier topology and its operation are explained in section 4.3. This simulated rectifier and its control circuit's block diagram is given in figure 6.7. The boost inductor, L_{s} , the output capacitor, C_{out} and their calculated values according to the specifications given in section 6.1 are also shown in this figure.

In this block diagram, low pass filter is necessary to filter the harmonics in

the line voltage. I_{ac} signal is a current signal, which represents the mains supply voltage in its circuit. It is proportional to the mains supply and taken over the diode bridge. The proportionality is shown with K2 block and a 650 k Ω resistor is used instead of the K2 block in the simulation environment. Its shape is a rectified sinusoidal one. V_{rms} is the voltage sample over the diode bridge and it is taken by using the low pass filter. Also, its shape is rectified sinusoidal one. This extra circuitry (V_{rms} and DIVIDER SQUARER part of AXB/C² MULTIPLIER DIVIDER SQUARER circuitry in figure 6.7) is needed to keep the rectifier's power constant irrespective of the input voltage changes. I_{ac} signal is multiplied by the voltage controller's output signal and divided by the square of V_{rms} signal. When the voltage and V_{rms} signal is halved. Half of the I_{ac} signal times one fourth of V_{rms} signal gives two times the original $I_{multout}$ current. Although the mains supply's voltage is halved and mains supply's current is duplicated, the rectifier's power is kept constant in this way.

The diode bridge's current is sensed and K1 block represents the ratio between the sensed current and the current, Isense in the control circuit block diagram. The ratio is taken as 0.04 in the simulations made through the following sections. The current, I_{sense} is subtracted from the I_{multout} current, it is processed by current controller and the current controller's output forces the PWM block to produce correct duty cycle and fixed frequency PWM so that the mains supply's voltage and current is in phase. Sinusoidal voltage and current in phase denote unity power factor system. Hence, irrespective of whether the load is resistive, capacitive or inductive, mains supply experiences resistive load. In the block diagram and in the simulation environment, V_{sense} voltage is taken by using a voltage divider which consists of the resistors, RVI and RVD. The voltage is subtracted from the reference voltage (reference voltage is 3V in UC3854B), this error voltage is given to the voltage controller, it is multiplied by I_{ac} signal and divided by the square of the V_{rms} signal. This means that when the voltage error signal is decreased, I_{multout} signal becomes lower, so the PWM signal's duty cycle given to the IGBT's gate is decreased. When the voltage error signal is high, duty cycle of the PWM signal becomes higher in the same way. This forces the output voltage, Vout to be higher. In this way, the output voltage is kept constant and proportional to the reference voltage in the block


Figure 6.7 Classical unity power factor rectifier and its control circuit's block diagram.

diagram. The current controller, voltage controller are actually current error amplifier and voltage error amplifier respectively. Their detailed descriptions are given in Appendix A and reference [42].

The proposed single phase, single switch unity power factor rectifier topology's working principle is illustrated and explained in detail in section 4.2. The rectifier topology and its control block diagram are shown altogether in figure 6.8. The boost inductors, L1 and L2, the output capacitor, C_{out} and also, their calculated values according to the specifications given in section 6.1 are shown in figure 6.8. The output capacitor's capacitance is equal to the one in figure 6.7 and the inductors, L1 and L2's total inductance are same with the boost inductance, L_s . These should be emphasized.



Figure 6.8 Proposed rectifier topology and its control block diagram.

With respect to the control system, the only difference between in figure 6.7 and figure 6.8 is where I_{ac} and V_{rms} signals are taken to the controller. D1, D2, D6 and D7 form a diode bridge. The signals can not be taken from this diode bridge 's output side since boost inductors, L1 and L2 are connected between mains supply and this diode bridge in figure 6.8, and the I_{ac} signal waveshape have to be rectified sinusoidal one in the suggested rectifier circuit as in the classical unity power factor rectifier circuit. The I_{ac} and V_{rms} signals are taken from a 10:1 transformer connected to the mains supply as in figure 6.8 for the suggested rectifier. In this topology, there should be a low pass filter to reduce the harmonics, which are due to the transistor switching and possible harmonics on the mains supply too. The low pass filter circuit design for classical unity power factor rectifier with the PFC IC is given in detail in Appendix A. However, resistance and capacitance values in the low pass filter circuit and K2's value in the classical unity power factor rectifier and the proposed rectifier are different. Their values are given in section 7.2.4.

The rest of the UPF controller design in Appendix A and reference [42] is

applicable to this configuration and its control circuitry. "AXB/C² MULTIPLIER DIVIDER SQUARER", the voltage controller, current controller, K1 and PWM blocks in figure 6.8 are equal to the ones in figure 6.7. In order to inhibit the diodes, D1, D2, D6, D7's switching losses, a high value resistance such as 500 kohm is connected parallel to them as in figure 6.8. Its value should be high enough to minimize the I²R losses over the resistance and low enough to keep the diodes in conduction.

The classical unity power factor rectifier's I_{ac} and V_{rms} signals can be taken by using a transformer and a diode bridge like B1 as in figure 6.8, and the same low pass filter in figure 6.8 can be used in the classical unity power factor rectifier as in the proposed rectifier when UC3854B IC is used. The classical unity power factor rectifier is simulated with PSIM 6.1 simulation program and UC3854B IC model in figure 6.1 with 3 kW purely resistive load at steady state to observe the classical unity power factor rectifier's performance by taking I_{ac} and V_{rms} signals via a 10:1 transformer and a diode bridge like B1 as in figure 6.8. The simulation program's time step is taken same as the two rectifier's simulations at steady state with 3 kW loads mentioned before in this section. This simulation results are given briefly in section 6.4.3.

6.4.1 Classical And Proposed UPF Rectifier Simulation With 3 kW Purely Resistive Load And Their Performance Comparison

Classical unity power factor rectifier and proposed rectifier, whose block diagrams are given in figure 6.7 and figure 6.8 respectively are simulated by using PSIM 6.1 simulation program and the UC3854B IC model in figure 6.1 with 3 kW purely resistive load at steady state in this section. In this way, whether the two rectifiers, especially proposed rectifier can be controlled with this IC will be understood. Then, the two rectifiers' performances will be compared considering upon their efficiency, input power factor, THD (Total Harmonic Distortion), input current ripple, output voltage ripple, IGBT and diodes' stresses and losses, etc.

6.4.1.1 Classical UPF Rectifier Simulation With 3 kW Purely Resistive Load

The classical unity power factor rectifier with 3 kW purely resistive load is simulated. 64.5 ohm resistor is connected to the output of the unity power factor rectifier whose block diagram is displayed in figure 6.7 as the 3 kW load.



Figure 6.9 Classical unity power rectifier's input voltage and current waveform with 3 kW purely resistive load at steady state.

The rectifier's input current and output voltage waveforms are shown respectively in figure 6.9 and 6.10 respectively. The input current is in phase with the input voltage, so this rectifier with the 3 kW purely resistive load achieves unity power factor operation. The output voltage in figure 6.10 swings between 432.26 and 448.1 V. Hence, 3.6 % output voltage ripple is obtained while the rectifier is designed for maximum 10 % output voltage ripple. On the other hand, the rectifier's

mean output voltage stays constant at steady state and this means that the output voltage is stable. Its output power obtained from the figure 6.10 for the given load is 3006 W. The peak input current's magnitude changes between 22.46 A and 18.77 A. This means that the rectifier's input current ripple is 8.95 % peak to peak. It is lower than the given input current ripple specification, which is 10 % peak to peak in section 6.1.



Figure 6.10 Classical unity power rectifier's output voltage waveform with 3 kW purely resistive load at steady state.

Power viewed from the mains line connected to the rectifier's input is shown in figure D.1(a) and its power factor is displayed in figure D.1(b) in Appendix D.1.1. Figures D.2, D.3, D.4, D.5, D.6 in Appendix D.1.1 show its input current FFT waveform and THD, IGBT's current, voltage waveform, one of the bridge diode, namely D1's current, voltage waveform and the boost diode, D5's current and voltage waveform respectively.

The power loss of the IGBT in figure 6.7 is calculated by using equation 6.1. In this equation, T, V_{CE} , i_C are the mains supply's voltage period, the IGBT's voltage between its collector and emitter terminals, and its collector current respectively.

$$Powerloss = \frac{1}{T} \int_{t=t_1}^{t=t_1+T} (V_{CE} \cdot i_C) dt$$
(6.1)

Power losses of the power diodes shown in figure 6.7 are also calculated by using equation 6.1, but the voltage between their anode and cathode is used instead of the voltage, V_{CE} .

The rectifier's efficiency, input power factor, THD of its input current, input current ripple, output voltage ripple and the IGBT, boost diode, D5, bridge diodes, D1, D2, D3, D4's voltage and current stresses, and their power losses are given in table 6.6. Its input current's harmonics up to 50th harmonic is listed in the part related with the classical UPF rectifier in table 6.7.

6.4.1.2 Proposed Converter's Simulation With 3 kW Purely Resistive Load

Proposed rectifier is simulated to compare its performance with the classical unity power factor rectifier when the 3 kW purely resistive load, specifically 64.5 resistor, is connected to their output at steady state.

In order to calculate the proposed rectifier's input current ripple, its input current waveform with input voltage waveform is shown in figure 6.11. According to this figure, input peak current changes between 19.07 and 22.74 A, this means that input current ripple is 8.78 %. Figure 6.12 shows the rectifier's output voltage at steady state and it swings between 448.24 and 432.16 V. The rectifier's mean output voltage is 440.20 V. Hence, the output voltage ripple is nearly 3.65 %. On the other hand, its mean output voltage stays constant. This means that its output voltage is stable at steady state. The rectifier's output power is calculated using wattmeter block in PSIM 6.1 and it is 3006.32 watts. According to the figure 6.11 and 6.12, unity power factor operation at its input side and DC voltage at its output side are achieved and the output voltage ripple and input current ripple are lower than the related specifications given in section 6.1. This means that the proposed rectifier can be controlled by using UC3854B IC.

The proposed rectifier's input power and power factor graph is given in figure



Figure 6.11 Proposed rectifier's input voltage and current waveform with 3 kW purely resistive load at steady state.



Figure 6.12 Proposed rectifier's output voltage waveform with 3 kW purely resistive load at steady state.

D.7(a) and D.7(b) in Appendix D.2. Its input current's FFT waveform and THD, IGBT's current and voltage waveform, one of the boost diodes, D3's current and voltage waveform are given in figure D.8, D.9, D.10, D.11 in Appendix D.2 respectively. The bridge diodes, D1 and D7's current and voltage waveforms are given in figure D.12 and D.13 in the same part in Appendix D.

The rectifier's efficiency, input power factor, THD of its input current, input current ripple, output voltage ripple and the IGBT, boost diodes, D3 and D5, bridge diodes, D1, D2, D6, D7's voltage and current stresses, and their power losses are given in the part related with the proposed rectifier in table 6.6. Its input current's harmonics up to 50th harmonic is listed in the part related with the classical UPF rectifier in table 6.7.

6.4.1.3 Evaluation Of Simulation Results Of Classical And Proposed UPF Rectifier With 3 kW Purely Resistive Load

The classical UPF rectifier and proposed rectifier's efficiency, input power factor, input current's THD, input current ripple, output voltage ripple, IGBT and power diodes' stresses and their power losses are obtained via the simulations with the 3 kW purely resistive load at steady state and are listed in table 6.6.

Proposed rectifier's efficiency and input current ripple are better than the classical unity power factor rectifier's ones according to the data in table 6.6 while classical unity power factor rectifier's input power factor, input current's THD, output voltage ripple are better than the proposed rectifier's ones. The diode bridge consisting the diodes, D1, D2, D6, D7 in the proposed rectifier has lower power losses than the classical unity power factor rectifier's efficiency higher than the classical UPF rectifier's one.

Voltage and current stresses of the IGBT and boost diodes in proposed rectifiers are nearly the same as in the classical UPF rectifier while the diode bridge in proposed rectifier has higher voltage stresses than the one in classical UPF rectifier.

	Classical UPF Rectifier	Proposed UPF Rectifier
	With 3 kW Purely Resistive	With 3 kW Purely Resistive
	Load	Load
Efficiency (%)	97.29	97.45
Input Power Factor	0.9873	0.9864
THD (%)	16.06	16.59
Input Current Ripple	8.95	8.78
(%)		
Output Voltage Ripple	3.6	3.65
(%)		
IGBT Stress	449.42 V, 22.46 A	448.71 V, 22.74 A
Boost Diode Stress	446.1 V, 22.46 A	445.27 V, 22.74 A
Bridge Diodes' Stresses	309.87 V, 22.26 A	449.73 V, 22.74 A for D1, D2
_		449.57 V, 22.74 A for D6, D7
IGBT's Power Loss	10.04 W	9.88 W
Boost Diodes' Power	8.89 W	8.90 W
Loss		
Diode Bridge's Power	25.28 W	2.6 W for each D1, D2
Loss		6.28 W for each D6, D7
		(Total 17.76 W)

Table 6.6 Performance comparison table between proposed rectifier and classical unity power factor rectifier with 3 kW purely resistive load (simulated).

The proposed rectifier and classical unity power factor rectifier's input current harmonics up to 50th harmonic are obtained by using the FFT waveforms in figure D.2 and figure D.8 in Appendix D.1 and they are listed in table 6.7. The two rectifiers' input current harmonics have similar magnitudes and don't have even harmonics.

THD (Total Harmonic Distortion) in their input current can be calculated by using equation 6.1 for the first 50 harmonics in table 6.7.

$$THD = \frac{\sqrt{\sum_{h=2}^{\infty} I_h^2}}{I_1}$$
(6.1)

, where I_h is rms value of hth harmonic and h is an integer, and I_1 is fundamental current component in rms [39]. In this way, THDs of the classical unity power factor rectifier and proposed rectifier's input currents are calculated as 14.12 % and 14.76 % respectively. On the other hand, the THD values in table 6.6 are higher than the THD calculated by considering the first 50 harmonics as expected. This indicates that

	Classical UPF Proposed UPF			Classical UPF	Proposed UPF
	Rectifier	Rectifier		Rectifier	Rectifier
Harmonic	Magnitude	Magnitude	Harmonic	Magnitude	Magnitude
#	(A)	(A)	#	(A)	(A)
1	19.60	19.56	26	0.00	0.00
2	0.00	0.00	27	0.11	0.10
3	1.75	2.08	28	0.00	0.00
4	0.00	0.00	29	0.20	0.19
5	1.46	1.40	30	0.00	0.00
6	0.00	0.00	31	0.17	0.15
7	1.16	1.06	32	0.00	0.00
8	0.00	0.00	33	0.05	0.05
9	0.55	0.48	34	0.00	0.00
10	0.00	0.00	35	0.10	0.10
11	0.12	0.12	36	0.00	0.00
12	0.00	0.00	37	0.15	0.14
13	0.44	0.41	38	0.00	0.00
14	0.00	0.00	39	0.10	0.10
15	0.47	0.42	40	0.00	0.00
16	0.00	0.00	41	0.03	0.03
17	0.23	0.20	42	0.00	0.00
18	0.00	0.00	43	0.09	0.08
19	0.12	0.12	44	0.00	0.00
20	0.00	0.00	45	0.10	0.10
21	0.29	0.27	46	0.00	0.00
22	0.00	0.00	47	0.06	0.06
23	0.28	0.25	48	0.00	0.00
24	0.00	0.00	49	0.03	0.03
25	0.11	0.10	50	0.00	0.00
THD (considering first 50 harmonics)		THD (considering first 50 harmonics)			
= 14.12 % for classical UPF rectifier		= 14.76 % for proposed UPF rectifier			
THD (all harmonics) = 16.06%		THD (all harmonics) = 16.59%			
for classical UPF rectifier		for proposed UPF rectifier			

Table 6.7 Proposed and classical UPF rectifier's input current harmonics' number and magnitudes up to 50^{th} harmonic with 3 kW purely resitive load at steady state.

most of the distortion is due to the first 50 harmonics. As discussed in section 7.4, the harmonics around the carrier frequency are the main contributors to THD from the remaining harmonics.

The proposed rectifier is implemented and it is tested with 1 kW and 2 kW purely resistive loads and it is simulated with these loads. Its simulation and measured results are given in section 7.4.1 and 7.4.2. Also, it is simulated with 300 W and 600 W purely resistive loads by using the simulation circuit with the conditions mentioned in section 7.4, but this simulation results are not given in detail

in this thesis. The proposed rectifier's efficiency, its input current's THD and input power factor graphs are drawn by combining these results with the results given in the related part of table 6.6.

The followings are observed from the figure 6.13, 6.14 and 6.15:

- 1. The simulated and measured input power factor, THD and the simulated efficiency become better as the load is increased to its nominal value, which is 3 kW.
- 2. I²R losses in the implemented circuit and the non-ideal boost inductors used in the circuit increase and the measured efficiency is decreased as the load increases according to the measurements with 1 kW and 2 kW purely resistive load. However, it is expected that the measured efficiency will decrease as its load is decreased to such values of 600 W, 300 W or more as the efficiency graph obtained via simulations and shown in figure 6.13.



Figure 6.13 Proposed rectifier's efficiency graphs obtained with measurements in laboratory and simulation with different purely resistive loads at steady state.



Figure 6.14 Proposed rectifier's input power factor graphs obtained with measurements in laboratory and simulation with different purely resistive loads at steady state.



Figure 6.15 THD graphs of proposed rectifier's input current obtained with measurements in laboratory and simulation with different purely resistive loads at steady state.

Also, the proposed rectifier is re-simulated with 1 kW and 2 kW purely resistive loads at steady state by using the simulation circuit with the conditions mentioned in section 7.4 and connecting the 36 m Ω measured winding resistance serially to boost inductors, L1 and L2 in this simulation circuit. Its input power and output power are 1048.31 VA, 996.88 W with 1 kW purely resistive load and 2091.03 VA, 2023.78 W with 2 kW purely resistive load respectively, and its output voltage is nearly 435 V with 219.2 V rms input voltage according to these simulations. The efficiency is calculated as 95.09 % with 1 kW purely resistive load and 96.78 % with 2 kW purely resistive load with the boost inductor resistance included in the simulations. These efficiency values are less than the corresponding efficiency values in figure 6.13. Note that the switching losses of power semiconductors in proposed rectifier are not included in these simulations.

HGTG20N60A4D IGBT's total switching energy loss graph vs. its gate resistance is given in its datasheet in Appendix B. 6 Ω gate resistance is used in implemented proposed rectifier. According to this graph, the IGBT's switching loss for one switching is nearly 0.35 mJ when its collector to emitter current is 10 A and the voltage difference between its collector and emitter is 390 V. The simulation circuit's average input current and output voltage are 4.3 A, 435 V with 1 kW purely resistive load and 8.56 A, 435 V with 2 kW purely resistive load respectively. It is assumed that the switching loss is directly proportional to the current magnitude and voltage magnitude. The IGBT's switching losses are calculated as 8.4 W with 1 kW purely resistive load and 16.9 W with 2 kW purely resistive load by using the current and voltage values and the total switching energy loss graph at steady state. The simulation circuit's efficiency will be 94.34 % with 1 kW purely resistive load and 96.01 % with 2 kW purely resistive load at steady state by including the IGBT's switching losses.

It can be observed that the losses of the converter are predicted somewhat higher but still the difference between the measurements and predictions is very small, and is about 1.5 %.

6.4.2 Classical UPF Rectifier And Proposed Rectifier's Simulation With 3 kW Inductive-Resistive Load And Their Performance Comparison

Classical unity power factor rectifier and proposed rectifier, whose block diagrams are given in figure 6.7 and figure 6.8 respectively are tested via simulations to understand whether they lose their unity power factor operation with inductive-resistive loads or not. Two different 3 kW inductive-resistive loads are used in the simulations made in this section to verify this. One of them is 100 uH inductor, whose inductance is big enough to check the unity power factor operation, serially connected to a 64.5 Ω resistor, the other is 10 uH inductor serially connected to a 64.5 Ω resistor. Also, their performances considering upon their efficiency, input power factor, THD (Total Harmonic Distortion), input current ripple, output voltage ripple, IGBT and diodes' stresses and losses are investigated with these inductive loads at steady state by making the simulations. PSIM 6.1 simulation program and the UC3854B IC model in figure 6.1 are used in this section as the previous section.

6.4.2.1 Classical Unity Power Factor Rectifier's Simulation With Inductive-Resistive Load

In this section, only the classical unity power factor rectifier with the 3 kW inductive-resistive loads are simulated and its performances are investigated with the simulations made.

6.4.2.1.1 Classical Unity Power Factor Rectifier's Simulation With Inductive-Resistive Load (100uH)

Aim of the simulations is here to observe the classical unity power factor rectifier's output voltage ripple, input current ripple, efficiency and input power factor and input current's THD (Total Harmonic Distortion) with a inductiveresistive 3 kW load which has 100 uH ideal inductor serially connected to a 64.5 Ω resistor at steady state.



Figure 6.16 Classical unity power rectifier's input voltage and current waveform with 3 kW inductive-resistive load (100 uH) at steady state.



Figure 6.17 Classical unity power rectifier's output voltage waveform with 3 kW inductive-resistive load (100 uH) at steady state.

The rectifier's input current and voltage together are given in figure 6.16. The figure shows that input voltage and current are almost in phase. Its input current ripple is obtained by finding the peak input current's magnitude from this figure. It is 8.95 % peak to peak since the input current changes between 22.46 and 18.82 A. The rectifier's output voltage swings between 432.26 and 448.1 volts, so the output voltage ripple is 3.6 %. On the other hand, its mean output voltage stays constant. This means that the output voltage is stable. The rectifier's input current ripple and output voltage ripple is less than the ones in the specifications given in section 6.1 with this load. Also, unity power factor operation is maintained with this inductive-resistive load.

Power viewed from the mains line connected to the rectifier's input is shown in figure D.14(a) and its power factor is displayed in figure D.14(b) in Appendix D.1.1. Figures D.15, D.16, D.17, D.18, D.19 in Appendix D.1.1 show its input current's FFT waveform and THD, IGBT's current, voltage waveform, one of the bridge diode, namely D1's current, voltage waveform and the boost diode, D5's current, voltage waveform respectively.

The rectifier's efficiency, input power factor, THD of its input current, input current ripple, output voltage ripple and the IGBT, boost diode, D5, bridge diodes, D1, D2, D3, D4's voltage and current stresses, and their power losses are given in the related part of table 6.8. Its input current's harmonics up to 50th harmonic is listed in the part related with the classical UPF rectifier in table 6.9.

6.4.2.1.2 Classical Unity Power Factor Rectifier's Simulation With Inductive-Resistive Load (10uH)

UPF rectifiers should not change their behaviors as much as possible whether the load is inductive, capacitive or purely resistive. In section 6.4.2.1.1, the classical unity power factor rectifier is simulated with 100 uH inductor serially connected to 64.5 Ω resistor. In this section, our goal is to study whether THD, input current and output voltage ripple, etc. change or not at steady state when the rectifier's load is changed with 10 uH inductor serially connected to 64.5 Ω resistor.



Figure 6.18 Classical unity power rectifier's input voltage and current waveform with 3 kW inductive-resistive load (10 uH) at steady state.



Figure 6.19 Classical unity power rectifier's output voltage waveform with 3 kW inductive-resistive load (10 uH) at steady state.

Figure 6.18 and 6.19 show the rectifier's input voltage, input current and output voltage respectively. The input current changes between 22.45 and 18.84 A at its peak. The rectifier's maximum output voltage is 448.102 V and its minimum output voltage is 432.168 V. However, the rectifier's mean output voltage stays constant and this means that it is stable. Output voltage and input current ripple are calculated as 3.62 %, 8.74 % respectively. The rectifier gives 3006 W with this load.

The IGBT and diodes' voltage and current waveform are not given in this section since their waveform shapes are similar to ones in previous section. The rectifier's input power and input power factor are shown in figure D.20(a) and D.20(b) in Appendix D.2.1.2. Its input current's FFT waveform and THD graph are given in figure D.21 and D.22 in also Appendix D.2.1.2.

The rectifier's efficiency, input power factor, THD of its input current, input current ripple, output voltage ripple and the IGBT, boost diode, D5, bridge diodes, D1, D2, D3, D4's voltage and current stresses, and their power losses are given in the related part of table 6.8. Its input current's harmonics up to 50th harmonic is listed in the part related with the classical UPF rectifier in table 6.10.

6.4.2.2 Proposed Converter's Simulation With Inductive-Resistive Load

In this section, only the proposed rectifier with the 3 kW inductive-resistive loads is simulated and its performances are investigated with the simulations made.

6.4.2.2.1 Proposed Converter's Simulation With Inductive-Resistive Load (100 uH)

The classical unity power factor rectifier is simulated when its loads are inductive and purely resistive, and also, the proposed rectifier is simulated by connecting purely resistive load to its output before. The related results are given before in previous sections. In this section, proposed rectifier's simulation results with 3 kW inductive-resistive load which consists of a 100 uH inductor and 64.5 Ω resistor serially connected to the inductor at steady state are given. Its output voltage ripple, input current ripple, efficiency and the THD (total harmonic distortion), etc are investigated.

Figure 6.20 and figure 6.21 show the proposed rectifier's input current with its input voltage and its output voltage waveform respectively. Its output voltage swings between 448.244 and 432.151 V with this load at steady state, and its mean output voltage is closely 440.2 V. This mean output voltage stays constant through the simulation at steady state. The output voltage ripple is nearly 3.66 %. Also, its output power is 3006.32 W. Input peak current changes between 22.74 and 19.07 A, so input current ripple is 8.78 %. Proposed converter's output voltage ripple and input current ripple with this inductive-resistive load are moderately higher than the ones with 3 kW purely resistive load given in section 6.4.1.2.



Figure 6.20 Proposed rectifier's input voltage and current waveform with 3 kW inductive-resistive load (100 uH) at steady state.



Figure 6.21 Proposed rectifier's output voltage waveform with 3 kW inductiveresistive load (100 uH) at steady state.

The proposed rectifier's input power and power factor graph is given in figure D.23(a) and D.23(b) in Appendix D.2.2.1. Its input current's FFT waveform and THD, IGBT's current and voltage waveform, one of the boost diodes, D3's current and voltage waveform are given in figure D.24, D.25, D.26, D.27 in Appendix D.2.2.1 respectively. The bridge diodes, D1 and D7's current and voltage waveforms are given in figure D.28 and D.29 in the same part in Appendix D.

The rectifier's efficiency, input power factor, THD of its input current, input current ripple, output voltage ripple and the IGBT, boost diodes, D3 and D5, bridge diodes, D1, D2, D6, D7's voltage and current stresses, and their power losses are given in the part related with the proposed rectifier in table 6.8. Its input current's harmonics up to 50th harmonic is listed in the part related with the classical UPF rectifier in table 6.9.

6.4.2.2.2 Proposed Converter's Simulation With Inductive-Resistive Load (10 uH)

Proposed converter is simulated by placing a 100 uH inductor serially connected to 64.5 ohm to its output as a load and the simulation results are given in

previous section. In this section, the load is changed with 10 uH inductor serially connected to 64.5 Ω resistor and the converter is resimulated to investigate whether its power factor, total harmonic distortion, input current ripple, etc changes or not.

Proposed rectifier's input current and input voltage waveform are given in figure 6.22, and figure 6.23 shows its output voltage waveform. Its output voltage ripple and input current ripple do not change much to observe their differences when the load's inductance is 10 uH or 100 uH. This means that its output voltage varies between 448.244, 432.151 V and input peak current changes between 22.74 and 19.07 A. Similarly, its mean output voltage stays constant at steady state. The input current ripple and output voltage ripple are 8.78 % and 3.66 % as in the previous section. Also, its output power is 3006.32 W.

The IGBT and diodes' voltage and current waveform are not given in this section since their waveform shapes are similar to ones in previous section. The rectifier's input power and input power factor are shown in figure D.30(a) and D.30(b) in Appendix D.2.2.2. Also, its input current's FFT waveform and THD graph are given in figure D.31 and D.32 in Appendix D.2.2.2 respectively.



Figure 6.22 Proposed rectifier's input voltage and current waveform with 3 kW inductive-resistive load (10 uH) at steady state.



Figure 6.23 Proposed rectifier's output voltage waveform with 3 kW inductiveresistive load (10 uH) at steady state.

All important data such as the rectifier's efficiency, input power factor are given in table 6.8. Its input current's harmonics up to 50^{th} harmonic is also listed in the part related with the proposed rectifier in table 6.10. The results are evaluated in section 6.4.2.3.

6.4.2.3 Evaluation Of Simulation Results Of Classical And Proposed UPF Rectifier With 3 kW Inductive-Resistive Loads (100 uH and 10 uH)

Proposed and classical unity power factor rectifier are investigated upon their efficiency, input power factor, input current's THD, input current ripple, output voltage ripple, IGBT and diodes' stresses and losses with the two different 3 kW inductive-resistive loads. They are listed in table 6.8.

The proposed rectifier gives same results with the 3 kW loads, which are 100 uH inductor serially connected to a 64.5 Ω resistor load and 10 uH inductor serially connected to a 64.5 Ω resistor load according to data in table 6.8. Similarly, the classical unity power factor rectifier gives same results with this two loads.

	Classical	Proposed	Classical	Proposed
	UPF	UPF	UPF	UPF
	With	With	With	With
	3 kW	3 kW	3 kW	3 kW
	Inductive-	Inductive-	Inductive-	Inductive-
	Resistive	Resistive	Resistive	Resistive
	Load	Load	Load	Load
	(100 uH)	(100 uH)	(10 uH)	(10 uH)
Efficiency	97.29	97.45	97.29	97.45
(%)				
Input	0.9873	0.9864	0.9873	0.9864
Power				
Factor	16.06	1.6.70		1.6.70
THD	16.06	16.59	16.06	16.59
(%)		0.70	0.71	0.70
Input	8.95	8.78	8.74	8.78
Current				
Ripple (%)	2.00	2.66	2.62	2.00
Output	3.60	3.66	3.62	3.66
Voltage				
Kipple (76)	440.0 14	140 70 M	440 40 17	440 71 14
IGBT	449.0 V,	448.72 V,	449.42 V,	448.71 V,
Stress	22.40 A	22.75 A	22.45 A	22./5 A
Boost	446.12 V,	445.27 V,	446.12 V,	445.27 V,
Stross	23.47 A	22.74 A for D3 and D5	22.47 A	22.74 A for D3 and D5
Det des	200.07 M	101 D5 and D5	200.07 M	
Bridge Diodos'	309.87 V,	449.72 V,	309.87 V,	449.73 V,
Stresses	for each	for D1 D2	for each	for D1 D2
511 03503	diode	449.56 V	diode	449.56 V
	410 40	22.73 A		22.74 A
		for D6, D7		for D6, D7
IGBT's	10.04 W	9 88 W	10 04 W	9 88 W
Power	10.01 W	9.00 11	10.01 10	9.00 11
Loss				
Boost	8.89 W	8.9 W	8.89 W	8.9 W
Diodes'	0.07 11	(4.45 W	0.09 11	(4.45 W
Power		for each D3		for each D3
Loss		and D5)		and D5)
Diode	25.28 W	2.6 W for each D1	25.28 W	2.6 W for each D1
Bridge's	(6.32 W	D2	(6.32 W	D2
Power	for each	6.28 W for each D6,	for each	6.28 W for each D6,
Loss	diode)	D7	diode)	D7
		(Total 17.76 W)		(Total 17.76 W)

 Table 6.8 Performance comparison table between proposed rectifier and classical unity power factor rectifier with two 3 kW inductive-resistive (100 uH and 10 uH) load.

	Classical UPF	Proposed UPF		Classical UPF	Proposed UPF
	Rectifier	Rectifier		Rectifier	Rectifier
	(100 uH)	(100 uH)		(100 uH)	(100 uH)
Harmonic #	Magnitude (A)	Magnitude (A)	Harmonic #	Magnitude (A)	Magnitude (A)
1	19.6	19.56	26	0	0
2	0	0	27	0.11	0.1
3	1.75	2.08	28	0	0
4	0	0	29	0.2	0.19
5	1.46	1.4	30	0	0
6	0	0	31	0.17	0.15
7	1.16	1.06	32	0	0
8	0	0	33	0.05	0.05
9	0.55	0.48	34	0	0
10	0	0	35	0.1	0.09
11	0.12	0.12	36	0	0
12	0	0	37	0.15	0.14
13	0.44	0.41	38	0	0
14	0	0	39	0.1	0.09
15	0.47	0.42	40	0	0
16	0	0	41	0.03	0.03
17	0.23	0.2	42	0	0
18	0	0	43	0.09	0.08
19	0.12	0.12	44	0	0
20	0	0	45	0.1	0.1
21	0.3	0.27	46	0	0
22	0	0	47	0.06	0.06
23	0.28	0.25	48	0	0
24	0	0	49	0.03	0.03
25	0.11	0.1	50	0	0

Table 6.9 Proposed and classical UPF rectifier's input current harmonics' number and magnitudes up to 50th harmonic with 3 kW inductive-resistive load (100 uH) at steady state.

The rectifiers' harmonic number and magnitudes up to 50th harmonic with the 3 kW inductive-resistive loads (100 uH and 10 uH) at steady state are given in table 6.9 and table 6.10. There are small differences between the two rectifiers' harmonic magnitudes up to 50th harmonic in table 6.9 and table 6.10. Also, the even harmonic magnitudes in table 6.9 and table 6.10 are 0 for the two rectifiers. The harmonic magnitudes are the same except the 21st harmonic magnitudes in the parts related with the classical unity power factor rectifier in the two table. However, the difference between the 21st harmonic magnitude in table 6.9 and the one in table 6.10 is negligibly small.

	Classical UPF	Proposed UPF		Classical UPF	Proposed UPF
	Rectifier	Rectifier		Rectifier	Rectifier
	(10 uH)	(10 uH)		(10 uH)	(10 uH)
Harmonic #	Magnitude (A)	Magnitude (A)	Harmonic #	Magnitude (A)	Magnitude(A)
1	19.60	19.56	26	0.00	0
2	0.00	0	27	0.11	0.1
3	1.75	2.08	28	0.00	0
4	0.00	0	29	0.20	0.19
5	1.46	1.4	30	0.00	0
6	0.00	0	31	0.17	0.15
7	1.16	1.06	32	0.00	0
8	0.00	0	33	0.05	0.05
9	0.55	0.48	34	0.00	0
10	0.00	0	35	0.10	0.09
11	0.12	0.12	36	0.00	0
12	0.00	0	37	0.15	0.14
13	0.44	0.41	38	0.00	0
14	0.00	0	39	0.10	0.09
15	0.47	0.42	40	0.00	0
16	0.00	0	41	0.03	0.03
17	0.23	0.2	42	0.00	0
18	0.00	0	43	0.09	0.08
19	0.12	0.12	44	0.00	0
20	0.00	0	45	0.10	0.1
21	0.29	0.27	46	0.00	0
22	0.00	0	47	0.06	0.06
23	0.28	0.25	48	0.00	0
24	0.00	0	49	0.03	0.03
25	0.11	0.1	50	0.00	0

Table 6.10 Proposed and classical UPF rectifier's input current harmonics' number and magnitudes up to 50th harmonic with 3 kW inductive-resistive load (10 uH) at steady state.

The rectifiers maintain their unity power factor operation with these inductive-resistive loads. Their efficiency, input current ripples and output voltage ripples given in table 6.8 with the inductive resistive loads are within the specified values given in section 6.1.

6.4.3 Classical Unity Power Factor Rectifier's Simulation With 3 kW Resistive Load By Taking Reference Voltage From A Transformer Connected To Mains Supply

Classical unity power factor rectifier, whose block diagram is shown in figure 6.7, is simulated by taking I_{ac} and V_{rms} signal feedbacks from a transformer connected to mains supply as in figure 6.8 with PSIM 6.1 simulation program and UC3854B IC model in figure 6.1 to investigate whether taking these signals in this way affects its performance with 3 kW purely resistive load considering THD, output voltage ripple, input current ripple, etc.

The rectifier's input power and input power factor graphs are shown in figure D.33(a) and D.33(b) in Appendix D.3. Its input current with input voltage and its output voltage waveforms are shown in figure D.34(a) and D.34(b) in Appendix D.3. On the other hand, its input current's FFT waveform and input current's THD graph are shown in figure D.35(a) and D.35(b) in the same part in Appendix D. These figures are obtained via the simulation.

	Classical UPF Rectifier With 3 kW Purely Resistive Load Taking Reference Voltage By Using A Transformer
Efficiency (%)	97.21
Input Power Factor	0.9864
THD (%)	16.63
Input Current Ripple (%)	8.70
Output Voltage Ripple (%)	3.66
IGBT Stress	449.46 V, 22.79 A
Boost Diode Stress	446.16 V, 22.80 A
Bridge Diodes' Stresses	309.90 V, 22.80 A
IGBT's Power Loss	9.93 W
Boost Diodes' Power Loss	8.89 W
Diode Bridge's Power Loss	25.20 W (Each diode has 6.3 W loss)

 Table 6.11 Classical UPF rectifier performance table when its reference voltage is taken by using a transformer.

The rectifier's results such as efficiency, input power factor obtained via the simulation are listed in table 6.11. When the I_{ac} and V_{rms} signal feedbacks are taken from a transformer connected to mains supply as in figure 6.8, classical unity power factor rectifier has worse results than the ones when the feedbacks are taken from the bridge rectifier as in figure 6.7 by considering its input power factor, efficiency, output voltage ripple, THD, but its input current ripple given in table 6.11 is lower than its input current ripple in table 6.6. Also, the proposed rectifier with 3 kW purely resistive load has better efficiency, output voltage ripple and input current's THD given in table 6.6 than the classical unity power factor rectifier's ones in table 6.11. Proposed rectifier's power factor given in table 6.6 and the power factor in table 6.11 are same, but its input current ripple given in table 6.11 is lower than the one given in table 6.6. These show that the classical unity power factor rectifier's performance results are related with where the signal feedbacks are taken.

6.4.4 Classical Unity Power Factor And Proposed Rectifier's Simulation With 3 kW Purely Resistive Load In Dynamic Conditions And Their Comparison

The classical unity power factor rectifier and proposed rectifier, whose block diagrams are given in figure 6.7 and 6.8 respectively, are simulated in dynamic conditions. In the simulations, PSIM 6.1 simulation program and the UC3854B IC model in figure 6.1 are used. 3 kW purely resistive loads are connected to the rectifiers' output, after a while they are disconnected and then connected again in PSIM 6.1 simulation environment. In this way, the rectifiers' output voltage and input current waveforms are obtained in the dynamic conditions, and their settling times are calculated.

6.4.4.1 Classical Unity Power Factor Rectifier's Simulation With 3 kW Purely Resistive Load In Dynamic Conditions

Classical unity power factor rectifier whose block diagram is shown in figure 6.7 is simulated by connecting 3 kW purely resistive load and disconnecting it between 1s -1.05 s. and connecting again at the end of this period.







Figure 6.24 (a) Classical unity power factor rectifier's output voltage and input current waveform when 3 kW purely resistive load is connected, it is disconnected between 1 s. and 1.05 s. and connected again, (b) its zoomed version.

Figure 6.24 shows the rectifier's input current and output voltage in mentioned conditions above. Its maximum output voltage is nearly 447.84 V at steady state before the 3 kW purely resistive load is disconnected. Its \pm 5 % values are 424.45 V and 473.23 V. Time difference between the time, which 3 kW purely resistive load is reconnected at and the first time, which the rectifier's maximum output voltage reaches between these values at, gives settling time. The rectifier's maximum output voltage reaches 433.496 V at 1.09765 s. 433.496 V is between the voltage values. Then, the rectifier's settling time is 47.65 ms.

6.4.4.2 Proposed Rectifier's Simulation With 3 kW Purely Resistive Load In Dynamic Conditions



Figure 6.25 (a) Proposed rectifier's output voltage and input current waveform when 3 kW purely resistive load is connected, it is disconnected between 1 s. and 1.05 s. and connected again, (b) its zoomed version.

3 kW purely resistive load is connected to the proposed rectifier's output and it is disconnected between 1 s. and 1.5 s in PSIM 6.1 simulation environment. Then it is reconnected. In this way, the rectifier's settling time is calculated similarly in section 6.4.4.1. The rectifier's input current and output voltage waveform obtained by making a simulation with the simulation program and the waveforms are given in figure 6.25.

The proposed rectifier's maximum output voltage is 449.821 V. First time, its maximum output voltage reaches a value, which is between ± 5 % of 449.821 V at 1.09761 s. It is 434.193 V and the rectifier's settling time is 47.61 ms. The settling time is nearly same with the classical unity power factor rectifier's one given in section 6.4.4.1.

6.5 Conclusion

Classical unity power factor rectifier and proposed rectifier with 3 kW purely resistive load are compared on basis of their efficiency, power factor, input current's THD, input current ripple, output voltage ripple, transistor and diode number, and their stresses, etc. Some of them are given in table 6.12.

Proposed rectifier is better than the other one on the basis of efficiency and input current ripple. On the other hand, the proposed rectifier's output voltage ripple and input current's THD are a little bit higher than the other one and its power factor is lower than the other. However, these are nearly same for the two rectifiers. An extra ultrafast boost power diode is needed in the proposed rectifier. On the other hand, the classical unity power factor rectifier needs an extra free-wheeling diode to achieve the boost inductor's core resetting. The two rectifier's transistor and boost diode current and voltage stresses are nearly same. Proposed rectifier's bridge diodes voltage stresses are higher than the classical unity power factor rectifier's ones.

Proposed rectifier and classical unity power factor rectifier's simulation results are also obtained before by connecting serially connected 64.5 Ω resistor and 100 uH inductor to their outputs. They are given in table 6.13.

The efficiencies, power factors and THDs are same in table 6.12 and table

6.13. The rest of the parameters are nearly same. Proposed rectifier and classical unity power factor rectifier are simulated before by connecting serially connected 64.5 Ω resistor and 10 uH inductor to their outputs and similar performance results are obtained to the ones in table 6.13, so they are not given here. Thus, the rectifiers' performances don't differ much whether their loads are inductive or purely resistive.

	Classical unity power factor rectifier with 3 kW purely resistive load	Proposed rectifier with 3 kW purely resistive load	
Efficiency (%)	97.29	97.45	
Power Factor	0.9873	0.9864	
THD (%)	16.06	16.59	
Input Current Ripple (%)	8.95	8.78	
Output Voltage Ripple (%)	3.6	3.65	
Transistor Stress	449.42 V, 22.46 A	448.71 V, 22.74 A	
Boost Diode Stress	446.1 V, 22.46 A	445.27 V, 22.74 A	
Bridge Diode Stress	309.87 V, 22.26 A for each	449.73 V, 22.74 A for D1, D2 449.57 V, 22.74 A for D6, D7	
Number of Power	1 IGBT with	1 IGBT with	
Transistors	reverse-recovery diode	reverse-recovery diode	
Number of Power Diodes	1 ultrafast boost diode, 4 bridge diode. 1 ultrafast free-wheeling diode	2 ultrafast boost diode, 4 bridge diode	

Table 6.12 Performance comparison table between proposed rectifier and classical unity power factor rectifier with 3 kW purely resistive load.

Proposed rectifier topology has better efficiency, output voltage ripple and THD than classical unity power factor rectifier's ones when its I_{ac} and V_{rms} signal feedbacks are taken from a transformer connected to mains supply as in figure 6.8 and with 3 kW purely resistive load. The simulation results in this way are given before in section 6.4.3. These show that the rectifiers' performance depends on where the I_{ac} and V_{rms} signal feedbacks are taken.

The rectifiers are simulated by using dynamic load connected to their outputs and results are given in section 6.4.4.1 and 6.4.4.2. The proposed rectifier's settling

	Classical unity power factor rectifier with 3 kW inductive-resistive load (100 uH)	Proposed rectifier with 3 kW inductive- resistive load (100 uH)	
Efficiency (%)	97.29	97.45	
Power Factor	0.9873	0.9864	
THD (%)	16.06	16.59	
Input Current Ripple	8.95	8.78	
(%)			
Output Voltage	3.6	3.66	
Ripple (%)			
Transistor Stress	449.0 V, 22.46 A	448.72 V, 22.75 A	
Boost Diode Stress	446.12 V, 23.47 A	445.27 V, 22.74 A	
Bridge Diode Stress	309.87 V, 22.45 A for each	449.72 V, 22.74 A for D1, D2	
		449.56 V, 22.73 A for D6, D7	

Table 6.13 Performance comparison table between proposed rectifier and classicalunity power factor rectifier with serially connected 64.5 Ω resistor and 100 uHinductor load.

time is lower than the other one's settling time, but they are nearly same.

Consequently, proposed rectifier is decided to implement due to its higher efficiency and the reasons given above.

In the next chapter, implementation of the proposed rectifier is described. The following chapter also presents the measurement results on this rectifier. These results are used for assessing how reliable the simulations given in this chapter are.

CHAPTER 7

HARDWARE DESIGN IMPLEMENTATION AND RESULTS

7.1 Introduction

In this chapter, the proposed unity power factor rectifier, whose specification given in chapter 6 is implemented. Its block diagram is given in figure 7.1. The inductor used in proposed unity power factor rectifier topology is designed and its design procedure is given in this chapter. Information about snubbers is given briefly. In order to protect IGBT used in the unity power factor rectifier circuit, overvoltage and R-C snubber is designed and design procedure is given. Information about power semiconductors used in the proposed rectifier are also given. The proposed rectifier circuit's control circuit is explained.

Unity power factor boost rectifier's hardware implemented uses average current control and it contains a control circuit in the same electronic card to run the proposed rectifier and classical unity power factor rectifier, whose specifications are mentioned in chapter 6. In this way, the two rectifiers can work using the same electronic card and electronic components used in their control circuit are efficiently used. Otherwise; the number of resistors, capacitors and current sensor, diode bridge, ultrafast diodes, IGBT, etc would be doubled on the electronic card implemented or two electronic cards would had to be designed and implemented.

The proposed rectifier is tested in laboratory environment by connecting 1 kW and 2 kW nearly purely resistive loads. The results are presented in this chapter. Simulations are made in same conditions to identify whether the simulations are



Figure 7.1 Implemented proposed rectifier's block diagram.

reliable, so that some other tests can be performed in the simulation environment. In this manner, considerable time saving can be achieved in improving the circuit.

7.2 Hardware Design And Implementation

Implemented proposed rectifier's block diagram given in figure 7.1 contains some blocks and they are given in the same figure. They are inductor, overvoltage snubber, R-C snubber, current sensing circuit, low pass filter, the mains supply feedback block, control circuit, power supply, the proposed topology blocks. Each block diagram and its corresponding circuit is explained in this section.

7.2.1 Inductor Design

The inductors, L1 and L2 shown in figure 7.1 are designed on a same core. Each inductor has 250 uH inductance and their total inductance is 500 uH. Its power rating is nearly 3100 W and its maksimum permissible current flowing through the inductors are 27.5 A. The inductors are designed as if one 500 uH inductor is designed. There are two coils on a C core, each represents a 250 uH inductance.

Selection of best magnetic core in inductor design stage is very critical in a given application. There are various magnetic cores such as ferrite, powdered iron, electrical steel, orthonol etc[2], [36]. Amorphous metal cores has high saturation flux density, low frequency dependent losses. Implementation of an efficient and small size inductor operating at high frequencies e.g. 50 kHz is possible by using amorphous metal cores [37].

In order to begin inductor design, the unity power factor rectifier's necessary specifications should be remembered since they are used in inductor design stage. Inductor efficiency should be decided since overall rectifier efficiency is dependent upon the inductor's efficiency. It should be as high as possible. 99 % or higher inductor efficiency is enough for this application. It will decrease the rectifier's efficiency 1 % and it will cause nearly 30 W power loss in the rectifier. In inductor design procedure given below, the rectifier's efficiency is taken as 97 % according to the simulation results in chapter 6. The inductor is designed and design procedures are given in the following steps [37], [38]:

<u>Step #1:</u>

Thus, necessary specifications are taken as the following ones:

- The rectifiers' output power, $P_{out} = 3000 \,\mathrm{W}$
- Their switching frequency, $f_{sw} = 50 \text{ kHz}$
- Their output dc voltage, $V_{out} = 440 \text{ V}$

- Their minimum input voltage, $V_{in(ac)\min} = 187$ V rms
- Their maximum input voltage = 286 V rms
- The inductor's maximum temperature rise above ambient temperature, $\Delta T = 50$ °C
- Its efficiency, $\eta_{preg} \ge 99 \%$
- Assumed rectifiers' efficiencies = 97 %
- The rectifiers' input current ripple, $\Delta I = 10$ % peak to peak.

The total inductance, L1+L2 in figure 7.1 used in the proposed unity power factor rectifier or classical unity power factor rectifier must be calculated before the inductor design stage. The rectifiers' peak input current without considering its input current ripple calculated as 23.37 A by taking the rectifiers' input power, P_{in} as 3090 W and their minimum input voltage, $V_{in(rms)min}$ 187 V rms, and using equation 7.1. Hence; the rectifiers' input current ripple, ΔI is 4.674 A since it is taken as 0.1 times the input current's peak to peak value.

$$I_{pk} = \frac{\sqrt{2} \times P_{in}}{V_{in(rms)\min}}$$
(7.1)

Duty cycle, D expression is given in chapter 2 in equation 2.2. Equation 2.2 gives D nearly 0.4 by taking the output DC voltage, V_o , input minimum peak voltage, $V_{in(peak)\min}$ as 440 V and 265 V respectively [2]. The inductance is calculated as 0.454 mH by using equation 2.2 and taking ΔI as calculated before. Taking inductance as 0.5 mH is appropriate to reduce the current ripple and guarantee it to be within the specified tolerances.

For a teoritically optimum design [37], inductor cupper and core losses should be nearly equal and this is stated in equation 7.2.

$$Powerloss = P_{tot} = 2 \times P_{core} = 2 \times P_{cu}$$

$$(7.2)$$

Metglas amorphous metal C-cores have a core loss equation and it is given by the following formula.

$$P_{core} = 6.5 \times f_{sw}^{1.51} \times B_{ac}^{1.74} \times wt \text{ in watts}$$

$$(7.3)$$

In equation 7.3, the switching frequency, f_{sw} ; ac flux density, B_{ac} ; core weight, *wt* are in kHz, tesla, kg respectively.

Equation 2.2 can be rewritten as in (7.4) and taking inductance, L as 0.5 mH
,and placing the necessary parameter values in this equation; ΔI is recalculated as 4.22 A peak to peak for this inductance value.

$$L = \frac{\sqrt{2}V_{in(ac)\min} \times \left[1 - \left(\left(\sqrt{2}V_{in(ac)\min}\right)/V_{out}\right]\right]}{\Delta I \times f_{sw}}$$
(7.4)

The current ripple, ΔI , flux density, B and AC flux density, B_{ac} are expressed in equation 7.5.

$$\Delta I = \left(\frac{2B_{ac}}{B}\right) \left[\frac{\sqrt{2} \times (P_{out})}{\eta_{preg} \times V_{in(ac)\min}}\right]$$
(7.5)

By putting maximum peak flux density, B_{max} in equation 7.5 instead of B, and using the values in specifications given before, following statement is derived for the inductor being designed.

$$\frac{B_{ac}}{B_{\text{max}}} = 0.09021$$

Energy storage requirement of an inductor, E can be calculated using equation 7.6

$$E = 0.5(L \times I_{peak}^2) \tag{7.6}$$

 I_{peak} in this equation is here the rectifiers' input peak current with considering the its input current ripple and equation 2.3 expressed I_{pk} and $\Delta I \cdot I_{peak}$ is calculated as nearly 25.5 A using equation 2.3 but control circuit designed using UC3854B IC permits 27.5 A peak current. Hence, I_{peak} current is taken 27.5 A in equation 7.6 and the rest of the calculations. The inductance L is 0.5 mH, and then E is 0.189 joules.

Core area product, $W_a A_c$ is given in equation 7.7. In this equation, energy storage requirement of inductor, *E* is in joules, *J* is current density in A/cm², k is the window utilization factor. Current density, J is taken as 500 A/cm² to reduce the core volume since amorphous metal cores have generally higher prices than other types of core [2].

If the current flowing through the inductor has high frequency component, cupper resistance in the inductor will increase due to skin effect. Litz wire or foil winding can be used to reduce the resistance due to this effect [37]. The inductor design is made by the assumption that foil winding is used in the inductor. Hence, k can be taken as 0.4.

$$W_{a}A_{c} = \frac{2E}{B_{\max} \times J \times k}$$

$$= \frac{2 \times (0.189 \text{ joules}) \times 10^{4}}{B_{\max} \times 500 \times 0.4} = \frac{18.4}{B_{\max}}$$
(7.7)

Above statement is derived by placing the necessary parameter values in equation 7.7. Efficiency should be equal or higher than 99 % and inductor should not be exceed 50 °C from ambient temperature according to the specifications given before. Placing $B_{ac} / B_{max} = 0.09021$ expression in equation 7.3 to simplify and rewriting the equation in terms of B_{max} for the inductor being designed, the following expression is derived.

$$P_{core} = 36.35 \times (B_{\text{max}})^{1.74} \times wt$$

Using two above statements, core area product, $W_a A_c$ is calculated, core number which has equal or higher and nearest core area product in table 7.2 is selected, and then the core loss is calculated by different B_{max} values in Tesla. Total inductor losses are also calculated. The results are given in table 7.1.

B-H curve for the Metglas amorphous metal is given in figure 7.2 according to its datasheet and saturation flux density can be acquired from this curve. The saturation flux density for the Metglas amorphous metal cores is 1.56 T and its datasheet is given in Appendix B in the thesis. Value of B_{max} can be taken as high as 1.4 T for the amorphous metal cores [38].

However, the core losses in the inductor can be reduced by using smaller B_{max} values as shown in table 7.1. The purpose here to find the smallest core size and the designed inductor, whose efficiency is equal or higher than 99 % while carrying the peak current of 27.5 A, and the converter transfers 3 kW power. Also, selected core's temperature rise must not exceed 50 ^oC above ambient temperature. When B_{max} is equal to 0.8, 0.9, 1, 1.1, the inductor's efficiency is higher or equal to 99 % according to table 7.1.

Table 7.2 taken from reference [38] gives core numbers, their core area product, convective surface area and approximate power dissipation in watts for a given temperature rise. When B_{max} is equal to 1 or 1.1, total inductor loss causes



Figure 7.2 B-H curve for Metglas amorphous metal cores in kilogauss and oersted.

B _{max} (Tesla)	W _a A _c (cm ⁴) (Required)	Core Number	Core loss (watts)	Total inductor loss (watts)
0.8	23	AMCC-32	11.339	22.7
0.9	20.44	AMCC-25	11.5	23
1	18.4	AMCC-25	13.813	27.6
1.1	16.72	AMCC-20	14.586	29.2
1.2	15.33	AMCC-20	16.9728	34
1.3	14.15	AMCC-16B	16.0664	32.1
1.4	13.14	AMCC-16B	18.2756	36.6

Table 7.1 Required core area product, core number, core and total inductor losses for
the 500 uH inductor by different B_{max} values.

temperature rise higher than 50 0 C above ambient temperature according to table 7.2. While B_{max} is equal to 0.8 T, the inductor temperature rise is lower than 50 0 C, but higher volume core, namely AMCC-32 is required. On the other hand, total inductor loss gives nearly 50 0 C temperature rise above the ambient temperature and the inductor's efficiency is higher than 99 % when B_{max} is equal to 0.9 T. Hence, AMCC-25 amorphous metal core is selected and its B_{max} value will be taken as nearly 0.9 T.

			Approx Given T	imate Pov emperatu	wer Dissi 1re Rise	pation in	Watts for	r a
Core Number	W _a A _c (cm ⁴)	SA (cm ²)	20 °C	25 °C	30 ⁰ C	35 °C	40 °C	50 °C
AMCC-6.3	5.8	103.4	3.8	4.9	6.1	7.4	8.7	11.3
AMCC-8	7	115.9	4.2	5.5	6.9	8.3	9.7	12.7
AMCC-10	9.4	132.1	4.8	6.3	7.8	9.4	11	14.5
AMCC- 16A	12	143.1	5.2	6.8	8.5	10.2	12	15.7
AMCC- 16B	15	160.3	5.8	7.6	9.5	11.4	13.4	17.6
AMCC-20	17.6	172.3	6.3	8.2	10.2	12.3	14.4	18.9
AMCC-25	22.7	202.2	7.4	9.6	12	14.4	16.9	22.2
AMCC-32	26.9	216	7.9	10.3	12.8	15.4	18.1	23.7
AMCC-40	31.2	230	8.4	11	13.6	16.4	19.3	25.2
AMCC-50	46.2	303.5	11.1	14.5	18	21.7	25.4	33.2

 Table 7.2 Approximate power dissipation values in watts for a given temperature rise and core number.

<u>Step #2:</u>

When AMCC-25 amorphous metal core is selected and its B_{max} value is 0.9 T, the inductor efficiency is 99.23 % according to the total inductor loss in table 7.1 by taking the rectifiers' efficiency as if 100 %. Inductor total loss formula is given in equation 7.8. In this equation, η_L is inductor efficiency, P_{out} is rectifier's output power and η_{preg} is its efficiency. Using this formula, total inductor losses are recalculated.

$$P_{tot} = (1 - \eta_L) \times (P_{out} \div \eta_{preg}) \text{ in watts}$$
(7.8)

Then, the inductor loss, P_{tot} is nearly 23.8 watts. P_{core} and P_{cu} losses in the inductor are 11.9 W. 22 W dissipated in an AMCC-25 will give temperature rise to nearly 50 $^{\circ}$ C. Thus, these show that right core, namely AMCC-25 is selected in step 1. Power loss for the core weight formula is given in equation 7.9.

$$P = P_{tot} / 2wt \tag{7.9}$$

The core weight, *wt* is 0.38 kg for AMCC-25 core and then *P* is 31.32 W/kg. Equation 7.3 and 7.9 gives equation 7.10 expressing ac flux density, B_{ac} in terms of *P* and f_{sw} .

$$B_{ac} = \left[P / (6.5 f_{sw}^{1.51})^{1/1.74} \right]$$
(7.10)

In equation 7.10, B_{ac} , P, f_{sw} are in terms of T, W/kg, kHz. Then, B_{ac} is 0.0828 T.

In step 1, L and ΔI are calculated as 0.5 mH, 4.22 A. B_{max} is recalculated more precisely by putting these values in equation 7.5 and it is found to be 0.9179 T, which is acceptable.

<u>Step #3:</u>

Core area product, $W_a A_c$ is recalculated for given J, k values and by using equation 7.7 and taking B_{max} as 0.9179 T in order to check that the selected core in step 1 has enough core area product. The core area product is found to be 20.59 cm⁴. AMCC-25's core area product is 22.7 cm⁴ as seen in table 7.2. Then, it is right core to implement the inductor. The amorphous metal core has the following dimensions related with figure 7.3:

- a = 1.3 cm
- b = 1.5 cm
- c = 5.6 cm
- d = 2.5 cm
- e = 4.1 cm
- f = 8.2 cm

Also, the AMCC-25 core has the characteristics given below:

- Magnetic path length of core, $l_m = 19.6$ cm
- Core cross-section, $A_c = 2.7 \text{ cm}^2$



Figure 7.3 C core dimensions.

- Core weight, wt = 0.38 kg
- Core area product, $W_a A_c = 22.7 \text{ cm}^4$
- Convective surface area, $SA = 202.2 \text{ cm}^2$.

Step #4:

In this step, the number of turns required, N is calculated. It is expressed in terms of L, I_{peak} , B_{max} , A_c in equation 7.11 and the values of these variables are 0.5 mH, 27.5 A, 0.9179 T and 2.7 cm² for the inductor being designed and core selected. N is calculated as 56 turns.

$$N = \frac{L \times I_{peak} \times 10^4}{B_{max} \times A_c}$$
 turns (7.11)

Step #5:

In this step, the inductor's airgap value is calculated. Total air gap value required is expressed in equation 7.12 where l_m is magnetic core length, $\mu\Delta$ is incremental permeability at operating on B-H loop in figure 7.2. $\mu\Delta$ is nearly equal to 1000 for the operating point.

$$l_g = \frac{0.4 \times \pi \times N \times I_{peak} \times 10^{-4}}{B_{max}} - \frac{l_m}{\mu\Delta}$$
(7.12)

$$l_g = \frac{0.4 \times \pi \times 56 \times 27.5 \times 10^{-4}}{0.9179} - \frac{19.6}{1000}$$

Necessary parameters is put in equation 7.12; total air gap, l_g is calculated as 0.19 cm and then it is 0.095 cm/leg.

<u>Step #6:</u>

Fringing flux decreases the air gap reluctance and so increases inductance by a factor F. It can be calculated using the following formula approximately. In the equation, l_g is inductor's total air gap calculated in step 5 and the rest of the parameters related with the core's dimension and they are given in step 3.

$$F = \frac{(a + (l_g/2)) \times (d + (l_g/2))}{a \times d}$$
(7.13)

Then, the amount of fringing flux, F is calculated as 1.114.

<u>Step #7:</u>

In this step, the turn number is corrected by taking into account the fringing flux. In step 1; number of turns, N is calculated. However, the amount of fringing flux, F affects the effective core cross-section. Equation 7.14 expresses the number of turns, N by including the fringing flux affect. Necessary parameter values in the equation are given before in this sub-chapter. N is calculated as 52.65 by putting them in equation 7.14. N should be higher or equal to what is calculated. The core has two coils on both sides of the C cores as in figure 7.3. N must be even number, so it is taken as 54.

$$N = \sqrt{\frac{L \times \left[l_g + (l_m/\mu\Delta)\right] \times (10^8)}{0.4 \times \pi \times A_c \times F}} \text{ turns}$$
(7.14)

<u>Step #8:</u>

The cross-sectional area of the conductor used in the inductor is calculated as 0.062 cm² by taking b, c, k, N as 1.5 cm, 5.6 cm, 0.4 and 54 and putting them in equation 7.15.

$$A_x = \frac{b \times c \times k}{N} \tag{7.15}$$

The resistivity of copper as a function of temperature is expressed in equation 7.16. Assuming that the maximum ambient temperature is 30 0 C and the inductor gives 50 0 C additional temperature rise, *T* is taken 80 0 C. Then, the resistivity of copper is 2.16 $\mu\Omega$ -cm.

$$\rho = 1.724 \times [1 + 0.0042 \times (T - 20)] \times (10^{-6}) \,\Omega\text{-cm}$$
(7.16)

The resistance per unit length for the conductor is 34.7 $\mu\Omega$ -cm and is calculated by using (7.17)

$$R_{unit} = \frac{\rho}{A_x} \tag{7.17}$$

<u>Step #9:</u>

Total cupper losses are calculated in this step. Total resistance for the winding is expressed by equation 7.18.

$$\Omega_{Tot} = R_{unit} \times MTL \times N \tag{7.18}$$

Mean turn length is approximated by equation 7.19 for the C-core.

$$MTL = 2 \times (a + 2b + d) \tag{7.19}$$

Then it is 13.6 cm for AMCC-25 amorphous metal core. Hence, the coil resistance is for the inductor is 25.48 m Ω . Equation 7.20 gives total cupper losses in watts without regarding the skin effect.

$$P_{cu} = I_{rms}^2 \times \Omega_{tot} \tag{7.20}$$

Total cupper losses without including the skin effect is calculated nearly 5W by taking the rectifiers' input current, I_{rms} in rms as 14 A.

Step #10:

The ac flux density in the air gap and the core is recalculated with regarding the corrected number of turns, N and air gap, l_g by using the equation 7.21. B_{ac} is calculated as 0.075 T by taking N, ΔI , l_g as 54, 4.22 A, 0.19 cm respectively.

$$B_{ac} = \frac{0.4 \times \pi \times N \times (\Delta I/2) \times (10^{-4})}{l_g}$$
(7.21)

Step #11:

The core losses are recalculated by using the recalculated AC flux density in the air gap and the core. It is nearly 10 W by using equation 7.3 since the recalculated B_{ac} , switching frequency, f_{sw} , the core weight, *wt* are 0.075 T, 50 kHz, 0.38 kg.

Step #12:

The estimated convective surface area for the C-core is calculated in this step. It is expressed in equation 7.22. Necessary parameters in this equation are given before for the AMCC-25 core in step 3. Hence, it is 202.2 cm² for the core.

$$SA = 2f \times (b+d) + 2 \times (b+d) \times (b+e) + 2f \times (b+e)$$
(7.22)

Step #13:

Total power loss in the inductor is sum of the core and cupper losses and it is expressed in equation 7.23. Total power loss in the inductor is nearly 15 watts.

$$P_{tot} = P_{core} + P_{cu} \tag{7.23}$$

Approximate temperature rise in the inductor can be calculated by using the equation 7.24.

$$\Delta T = \left[\frac{P_{tot}}{SA}\right]^{0.833} \tag{7.24}$$

In equation 7.24, total power loss in the inductor, P_{tot} and SA are in mW and cm² respectively. ΔT is temperature rise above the ambient temperature in ⁰C and nearly 36 ⁰C.

The inductor is designed by assuming that copper foil winding is used for the winding. However, the copper foil winding couldn't be used in the inductor because a copper foil with electrical insulation and specifications calculated above created practical problems. Litz wire, which has 10 wires with 0.7 mm diameter, is used instead of the copper foil. The Litz wire's copper cross-section is nearly 0.0385 cm². It is less than what is calculated above. On the other hand, the rectifiers' input current and current passing through the inductor is 14 A rms as stated in step 9. Thus, current density in the Litz wire is nearly 364 A/cm² for the inductor implemented. The Litz wire's total resistance, Ω_{Tot} is measured by a micro-ohm meter and it is 36 m Ω . Total copper losses are recalculated by using equation 7.20 and it is nearly 7 W. Sum of the core and copper losses is 17 W since the core losses is what is calculated in step 11. The approximate temperature rise is recalculated by using equation 7.24 and it is nearly 40 °C. This temperature rise is lower than the maximum permissible temperature rise given in step 1. The implemented inductor has two coils with each one having 250 uH inductance and its total inductance is 500 uH. The inductance values are what are measured by RLC meter in a laboratory. Each part had 250 uH represents the L1 and L2 in figure 7.1. When the two parts is serially connected, it represents the L_s in figure 6.7. It has 27 turns on one leg and 27 turns on the other leg of the AMCC-25 amorphous metal core.

7.2.2 Snubber Design

In power converters, power semiconductor devices can be under stresses beyond its ratings due to noise, voltage or current transients, etc. This problem can be solved by either using a power semiconductor with higher rating or using snubbers. In the proposed unity power factor rectifier circuit, snubbers are used only for the IGBT in implemented circuit.

Snubber circuits are used with transistors to enhance their switching trajectory. Basic types of the snubbers[39] are:

- **Turn-off snubbers:** Turn-off snubbers are used for reducing the voltage stress across the transistor while the transistor turns off. There are various turn-off snubbers such as RCD, RC, C snubber.
- **Turn-on snubbers:** Turn-on snubbers are used for reducing the current stress over the transistor while the transistor turns on. It also reduces the voltage across the transistor during turn on. A simple turn-on snubber can be implemented by connecting a small inductance serially to the transistor. A large inductance causes lower turn-on voltages and lower turn-on losses. However, the large inductance causes overvoltages across the transistor and inceases the transistor's turn-off voltages.
- **Overvoltage snubber:** Stray inductances cause overvoltages across the transistor during the turn-off. It can be reduced by using a overvoltage snubber. It should be used with turn-off snubber.

An overvoltage and turn-off, namely R-C snubber is used in the implemented rectifier.

7.2.2.1 Overvoltage Snubber Design

Overvoltages across the transistor occur during the transistor turn-off due to stray inductances. In order to prevent the overvoltage, an overvoltage snubber as shown in figure 7.4(a) is used in the rectifiers' circuit [39,40].



Figure 7.4 (a) Overvoltage snubber, (b) its equivalent circuit during transistor turnoff.

The switch, actually a switching transistor is conducting initially. Voltage across the capacitor, C_{ov} is equal to V_d . During turn-off, the circuit in figure 7.4(a) is equal to the one in figure 7.4(b). The energy stored in stray inductance, L_{σ} during the switch in conduction is transferred to C_{ov} through the diode, D_{ov} . During this stage, the overvoltage across the switch, ΔV_{CE} is equal to voltage change across the capacitor, ΔV_{Cov} . Hence [39, 40],

$$\frac{C_{ov} \times \Delta V_{CE,\max}^2}{2} = \frac{L_{\sigma} \times I_o^2}{2}$$
(7.24)

When the current through L_{σ} is decreased and the current direction is reversed; the

overvoltage, ΔV_{CE} is diminishing and voltage across the capacitor, C_{ov} becomes equal to V_d by decreasing through the resistance, R_{ov} . Hence, recovery time of C_{ov} must satisfy the following condition, where $t_{off(min)}$ is minimum turn-off time of the transistor like an IGBT [39, 40].

$$2.3 \times R_{ov} \times C_{ov} \approx t_{off(\min)} \tag{7.25}$$

In C_{ov} calculation, stray inductance, L_{σ} value is not known and not measured; only it is taken 20 nH as in reference [40]. C_{ov} is calculated as 343.75 nF by using equation 7.24, setting the ΔV_{CE} to 0.1 times the rectifiers' output voltage, taking L_{σ} nearly 20 nH and I_{o} equal to I_{peak} (27.5 A as in inductor design stage). Its value is taken as 300 nF.

The UC3854B IC can gives PWM signals had up to 95 % duty cycle. In order to guarantee the recovery time of C_{ov} through the resistance, R_{ov} , $t_{off(min)}$ is calculated 0.5 µs by using equation 7.26 as if the maximum duty cycle, D_{max} is 97.5 %.

$$t_{off(\min)} = \frac{1 - D_{\max}}{f_{sw}}$$
(7.26)



Figure 7.5 Overvoltage snubber in implemented circuit.

The value of R_{ov} is found as 0.724 Ω by using the equation 7.25. It is taken as 0.56 ohm. Power loss over the resistor is checked by making a simulation and it is nearly 0.5 W. However, 0.56 ohm, 5W power resistor is used in the overvoltage snubber circuit to prevent damage on the resistor due to possible excessive heat. International Rectifier's 8ETX06 ultrafast diode is used as snubber diode in the circuit and its datasheet is given in Appendix B. The implemented overvoltage snubber circuit whose block diagram is shown in figure 7.1 is given in figure 7.5.

7.2.2.2 R-C Snubber Design

C snubber is simplest turn-off snubber and it is connected to paralel to the switching device. However, it can cause oscillations with stray inductances or inductance such as boost inductor in proposed rectifier or classical unity power factor rectifier. A series resistance to the capacitor is added to prevent the possible resonance and oscillations with the inductances. This snubber type is R-C snubber. There are other turn-off snubber types such as RCD snubber, but RC snubber is the simplest snubber type except C snubber to protect the IGBT from overvoltages during the turn-off [39, 44].

R-C snubber connected to a transistor with a stray inductance is shown in figure 7.6. The stray inductance, L_{σ} causes an overvoltage during transistor turn-off since it is unclamped with the free-wheeling diode, D_f . This overvoltage is decreased to an acceptable level by the R-C snubber absorbing the energy associated with the stray inductance, L_{σ} [39, 44].

R-C snubber optimal design curves are given in figure 7.7 [41]. R and C values in R-C snubber are calculated by using these curves based on the overshoot,

 \hat{V} and initial current factor, χ [41].

$$C = L_{\sigma} \times \left(\frac{I_l}{\chi \times V_s}\right)^2 \tag{7.27}$$

$$R = 2 \times \xi \times V_s \times \chi / I_l \tag{7.28}$$



Figure 7.6 R-C snubber connected to a transistor.



Figure 7.7 R-C snubber optimal design curves.

The value of stray inductance, L_{σ} is not known and it is assumed to be 20 nH. The

overshoot, \hat{V} is taken as 0.05 times the V_s equal to 440 V for the rectifiers, so χ is nearly 0.25 according to figure 7.7. I_l is taken as 27.5 A as the rectifiers' peak input current in inductor design stage in this chapter. Then, C is calculated as 1.25 nF. In implemented circuit, two 400 V, 2.2 nF capacitors are serially connected to each other and it has 1.1 nF equivalent capacitance. ξ is taken as 2 according to the curves. Then, equation 7.28 gives R as 16 ohm. Each 33 ohm two resistors are parallel connected to each other, their equivalent resistance is 16.5 ohm in the implemented circuit. Equation 7.29 gives the power loss on the resistor when the IGBT's fall time, t_{fv} and rise time, t_{rv} are higher than $R \times C$ time constant. Otherwise, the power loss on the resistor will be equal to $2P_{co} + P_{Lo}$ [41].

$$P_{R} = P_{Ron} + P_{Roff} = \frac{R \times C}{R \times C + t_{fv}} \times P_{CO} + \frac{R \times C}{R \times C + t_{rv}} \times (P_{CO} + P_{LO})$$
(7.29)

$$P_{co} = \frac{C \times V_s^2 \times f_{sw}}{2} \tag{7.30}$$

$$P_{LO} = \frac{L_{\sigma} \times I_l^2 \times f_{sw}}{2} \tag{7.31}$$



Figure 7.8 R-C snubber circuit.

 P_{co} and P_{Lo} are calculated as nearly 5.3 W and 0.38 W by using equation 7.30 and 7.31 and taking switching frequency, f_{sw} 50 kHz. The IGBT used in implemented circuit has typical 13 ns rise and 55 ns fall times. Then, equation 7.29 gives the power loss on the resistor as nearly 4.6 W.

Each of the two resistors used in implemented circuit has 4W power rating and this is enough according to the calculations. The implemented R-C snubber circuit is given in figure 7.8.

7.2.3 Current Sensing Circuit

UC3854B power factor corrector integrated circuit needs a current feedback which represents the rectifiers' input current in rectified form. The current feedback can be taken by using current sense resistor if the rectifiers' input power are low such as 100 W, 250 W. If the rectifiers' input power are on the order of kilowatts, power consumed over the current sense resistor will be high, so the rectifiers' efficiency will be severely downgraded.



Figure 7.9 (a) LEM current sensor module, (b) current sensing circuit with the current sensor.

In order to prevent the efficiency downgrading severely, a current sensing circuit is used and its block diagram is shown in figure 7.1. LA 55-P/SP1 current sensor module is used in the current sensing circuit shown in figure 7.9(b) and its datasheet is given in Appendix A. A current carrying conductor is inserted in the current sensor module's hole shown in figure 7.9(a). In order to achieve best magnetic coupling, the conductor have to be wound over the top edge of the device according to its datasheet and this is what is done in the current sensing circuit implemented.

This current sensor output acts as a current source proportional to the current measured. A 40 ohm resistor is connected between ground and the sensor output. Voltage drop over the resistor gives voltage signal proportional the current sensed. This voltage is buffered by using LM310 voltage buffer in implemented circuit. The current sensing circuit whose block diagram is shown in figure 7.1 is given in figure 7.9(b). The implemented current sensing circuit's overall gain is 0.04.

7.2.4 Low Pass Filter and Mains Supply Feedback Circuit

Low pass filter and mains supply's sample blocks are shown in figure 7.1. I_{ac} signal in figure 7.1 is a current signal proportional to mains supply's voltage and in rectified form. The control circuit with the power factor corrector IC forces the rectifier's input current to follow the I_{ac} signal waveshape. V_{rms} is also mains supply's voltage sample in rectified form. Low pass filter in the block diagram is necessary to filter the harmonics in the line voltage. In order to reduce the effects of harmonics to produced PWM signal, V_{rms} is taken from the low pass filter. On the other hand, a unity power factor rectifier design with some different PFC ICs such as UCC3818 needs different low pass filter. The low pass filter and its design are specific to UC3854B IC. Thus, its design is not given here. The application note in Appendix A should be used for a low pass filter design in circuits contain the IC. This extra circuitry with MULTIPLIER DIVIDER SQUARER block is needed to keep the rectifier's power constant irrespective of the input voltage changes and this is explained in chapter 6 in detail. Proposed rectifiers' low pass filter design with UC3854B and the application note, and where mains supply's voltage sample is taken somewhat differ from the classical unity power factor rectifier's one which is given straight forward in the application note. The differences are:

• Mains supply's sample block in figure 7.1 contains the proportional block, K2, whose output is I_{ac} current signal. Its input and output signals are voltage and current signal respectively, so it is accomplished by using a resistor in implemented circuit. The resistor has 65 kΩ resistance in proposed rectifier circuit and 650 kΩ resistance is used in classical unity power factor rectifier circuit because the sample is taken by using a 10:1 transformer as shown in figure 7.10 in the proposed rectifier circuit. In order to prevent the I_{ac} current signal to be affected by the capacitances in the low pass filter in figure 7.11, a diode is placed between the resistor and the low pass filter. The diode, namely 1N4004's datasheet is given in Appendix B.

The transformer, named reference transformer inputs are connected to mains supply. An R-C low pass filter, whose cut-off frequency is 10 kHz is connected to its output to decrease the possible noise that is comming from the rectifiers' other parts as shown in figure 7.10.



Figure 7.10 Circuit of mains supply's sample block in implemented circuit.

• The voltage signal, VVF1 in implemented circuit over RFF1NEW in figure 7.11 is taken by using a 10:1 transformer and a diode bridge, namely SB402,



Figure 7.11 Low pass filter in proposed rectifier circuit.



Figure 7.12 Low pass filter in classical unity power factor rectifier.

whose datasheet is given in Appendix B. Hence, this have to be considered when their values are calculated by using the application note. This signal's voltage value have to be used in their calculations. Two capacitors, CFF2 and CFF2NECES's equivalent capacitance is important in this figure. Their equivalent capacitance and CFF1's capacitance values are depends upon the values of the resistors, RFF2 and RFF3. This should be cared about.

For the conventional rectifier configuration, it is possible to take the voltage feedback from a transformer and use the low pass filter in figure 7.11. However, these may affect the conventional rectifier performance negatively. This is given in detail in section 6.4.3. Hence, the low pass filter in figure 7.12 is required for better performance for the conventional rectifier configuration and the voltage feedback is taken and the low pass filter in the figure is connected as in figure 6.7. RFF1P10LD, RFF1P20LD and RFF1P30LD's equivalent resistance is important in the figure. The equivalent resistance and RFF1NEW in figure 7.11 are different. Calculated values of RFF2, RFF3 is same for the two rectifiers' control circuit, so the ones of CFF1, CFF2 and CFF2NECES. The rest of the capacitances and resistances is same for the two rectifiers.

7.2.5 Control Circuit

The control block contains the current error amplifier, voltage error amplifier and PWM block. In order to obtain the sinusoidal input current nearly in phase with the rectifier's input voltage and DC output voltage at its output side, the control circuit gives necessary PWM signals to the IGBT.

Necessary circuit to implement the PWM block is inside the UC3854B IC. Hence, the circuit is not given here. The IC datasheet is given in Appendix B. The circuit converts the signal coming from the current error amplifier to PWM signals and the PWM signals are given to the IGBT's gate pin.

Current error amplifier and voltage error amplifier are originally simple integrator implemented by using an opamp. The simple integrator is shown in figure 7.13. The integrator's transfer function is given in equation 7.32.

$$G(s) = \frac{1}{s \times R_{ci} \times C_{cp}}$$
(7.32)



Figure 7.13 An integrator.

In order to increase a unity power factor rectifier's performance, which uses average current control, a pole and a zero are added the transfer function in equation 7.32. Then, the transfer function will be as in equation 7.33 [43].

$$G_{ca}(s) = \frac{1}{R_{ci} \times C_{cp}} \times \frac{s + \omega_z}{s \times (s + \omega_p)}$$
(7.33)

, where $\omega_z = \frac{1}{R_{cz} \times C_{cz}}$ and $\omega_p = \frac{C_{cz} + C_{cp}}{R_{cz} \times C_{cz} \times C_{cp}}$.



Figure 7.14 Error amplifier with compensator.

The transfer function in equation 7.33 consists one integrator (1/s), one zero at ω_z and one pole at ω_p . The integrator eliminates steady state error. The zero is used for guaranteeing the stability of the rectifier's input current, and the pole eliminates high frequency noise due to the power transistor's switching in the rectifier. The zero should be set to $f_{sw}/2\pi$ [43], the pole must be above $f_{sw}/2$ according to the application note in Appendix A. The current error amplifier has a zero at nearly 7900 Hz, and a pole at 50 kHz and its gain is nearly 7.4 in implemented circuit. It is shown in figure 7.15.

Though current error amplifier and voltage error amplifier can be implemented with same type of error amplifier, the pole and zero must be set at different frequencies [43].



Figure 7.15 Current error amplifier in implemented circuit.

The voltage error amplifier has a pole at nearly 12 Hz, a zero at a 10^{th} of the pole frequency in implemented circuit and its gain is nearly 0.008. It is shown in figure 7.16. The resistors, R9 and R10's total resistance in figure 7.16 represents the R_{ci} in figure 7.14. The total resistance and R6 in figure 7.16 form a voltage divider. In this way, the rectifier's output voltage is sensed. The opamps in figure 7.15 and 7.16 are in UC3854B IC and the other electrical components are connected to the IC.

The current error amplifier forces the rectifier's input current to become

nearly in phase with the input voltage while the voltage error amplifier forces the rectifier's output voltage to be DC voltage. The voltage error amplifier gain must be much more less than the current error amplifier gain to decrease the effect of the output voltage ripple to the input current's THD. A detailed description is given in Appendix A.



Figure 7.16 Voltage error amplifier in implemented circuit.

7.2.6 Power Supply Block

LEM current sensor, LM310 buffer and UC3854B IC needs DC supply voltages. The DC supply produced +18 V, +12 V and -12 V DC voltages is given in figure 7.17. The connector, named SUPPLYTRANSOUTPUT's inputs are connected a 5 W transformer which has two +25 V and 0 V outputs. One 0 V and +25 V is shorted to each other. These are used as grounding purpose and also these are shorted to earthground. The other +25 V and 0 V output are connected to diode bridge. The transformer input is also connected to mains supply.

LM7812, LM7912 and LM7818 are used for producing +12 V, -12 V, +18 V respectively. Their datasheets is given in Appendix A. +12 V and -12 V supplies are used for giving necessary supply voltages to the LEM current sensor and LM310



Figure 7.17 Power supply which produces +12 V, -12 V, +18 V DC voltage.

buffer. UC3854B is supplied by the +18 V.

R1 and C9 form a low pass filter whose cut-off frequency is nearly 3.1 kHz and it is used to prevent 10 kHz noise coming from the LM7812 IC to pass through the +12 V supply output. The detailed descriptions about the noise is found from its datasheet in Appendix B. Similarly, R2 and C10, R3 and C13 are used for -12 V and +18 V supply and same reason.

7.2.7 Proposed Topology Block

Proposed topology block in figure 7.1 contains the inductors, L1 and L2, the ultrafast diodes D3, D5, bridge diodes, D1, D2, D6, D7, a switching transistor, output capacitor, C_{out} . Information about the inductors and their design stage are given in section 7.2.1. The power semiconductors are selected in chapter 6 and why they are selected are given in the chapter. Fairchild Semiconductor's RURG5060 ultrafast diodes are used as the ultrafast diodes. Its average rectified forward current rating is 50 A at case temperature of 102 0 C and its voltage rating is 600 V. The diodes, D1,

D2, D6 and D7 form a diode bridge and IXYS VBO 30-12N07 diode bridge module is used instead of these diodes. The bridge diode module has 35 A average current rating and 1200 V maximum voltage stress. Fairchild HGTG20N60A4D IGBT is used as the switching transistor. Its voltage rating is 600 V. It has maximum 40 A continuous collector current rating at case temperature of 110 $^{\circ}$ C. However, its current rating downgrades to 37 A at 50 kHz switching speed and junction temperature of 125 $^{\circ}$ C. The power semiconductor's datasheets are given in Appendix B.

7.3 Proposed Rectifier And Classical UPF Rectifier's Schematic Diagram

The proposed rectifier and classical unity power factor rectifier are simulated by using average current control technique in chapter 6. The two rectifiers' control circuit can be implemented in the same printed circuit board since the same current control technique and same power factor corrector integrated circuit is used in the two boost rectifiers. This is what is done in this thesis work.

Implemented proposed rectifier and classical unity power factor rectifier's schematic diagram is given in same schematic diagram in this section. The schematic diagram is splitted in to two parts and they are given in figure 7.18 and figure 7.19. The boost inductor and output capacitor are placed outside the printed circuit board and so the schematic. They are connected in the rectifiers' circuit according to figure 6.7 and 6.8.

The output capacitor is charged initialy by using 200 ohm, 200 W power resistor and it is disconnected from the circuit by using a power switch manually. A load is connected after the capacitor is fully charged and the voltage upon the capacitor is reached the specified output voltage given in chapter 6. A circuit for automatically disconnecting the charging resistance is also designed. However, it is not tested in this thesis work and it is not shown in the schematic. On the other hand, it is shown in the printed circuit board given in figure E.1 in Appendix E.

Passing from classical unity power factor rectifier's control circuit to proposed rectifier's one is achieved by using and connecting jumpers to connectors,



Figure 7.18 The proposed and classical unity power factor rectifier PCB schematic not including +12 V, -12 V, +18 V supply voltages and reference voltages, VVAC and VVF1.



Figure 7.19 Printed circuit board schematic shown +12 V, -12 V, +18 V voltage supply and reference voltages, VVAC and VVF1 proportional to the mains supply in rectified form.

X1 and X2 shown in figure 7.18 manually. The proposed rectifier's control circuit works, when pins, X1-1, X1-2 are shorted to each other; and X2-1, X2-2 are connected to each other by using jumpers. On the other hand, when X1-2, X1-3 are connected to each other, and X2-2, X2-3 are shorted to each other; classical unity power factor rectifier's control circuit is established.

The rectifiers' printed circuit board is given in figure E.1. Its top and bottom layers are shown in figure E.2 and figure E.3.

7.4 Experimental Results And Comparison With Simulations In Same Conditions

In this section, proposed rectifier circuit implemented is tested in laboratory and the results are given. The proposed rectifier is tested by connecting 1 kW and 2 kW resistive loads. Nearly unity power factor in mains supply side and DC voltage at the output side are achieved. The rectifier's output voltage ripple, FFT (Fast Fourier Transform) up to 50th harmonic and THD (Total Harmonic Distortion) of input current are given.

The rectifier's input current harmonics are measured by HIOKI 3194 Motor/Harmonic Hi Tester. It shows the harmonics up to 50th harmonic. Also, the FFT waveform is obtained by Agilent 100 MHz DSO6014A scope. Important harmonics near the rectifier's switching frequency is obtained by using the FFT waveform and the important harmonics and the harmonics up to 50th harmonic measured by the HIOKI 3194 together with the harmonics obtained via simulation are given in a table. On the other hand, HIOKI 3194 is capable of measuring THD by considering first 3000 harmonics. The tester measures the harmonics more precisely than the scope's FFT measurement option. Hence, the THD measured by the tester is more reliable. This is also given in this section.

Simulations are made in same conditions with experiments to identify whether the results taken in laboratory is similar or not. In simulation circuit, R-C snubber and overvoltage snubber designed are connected to the IGBT. However, fall time and rise time parameters can not be entered in power semiconductors parameter part in PSIM 6.1 simulation software, but their conduction voltage drop parameters are used in the simulations. Also, UC3854 B IC model in figure 6.1 is used in these simulations. Simulation time step is taken as 1E-007. Also, ideal inductors are used in the simulations.

7.4.1 Proposed Rectifier's Experimental Results With 1 kW Resistive Load And Comparison With Simulation In Same Conditions

Proposed rectifier is tested by connecting 1 kW nearly purely resistive load. Its input current, input voltage and output voltage waveform is acquired by using Agilent DS06014A oscilloscope. Its input power, power factor, input rms current, input rms voltage, output power, output mean voltage, output rms current and input current harmonics up to 50th harmonic are measured by using HIOKI 3194 Motor/Harmonic Hi Tester. The results are given below.

Figure 7.20 shows the rectifier's input voltage in green color and input current in yellow color. The waveforms are acquired by using the scope in real time mode, a current probe and differential probe. There are voltage ripples on the input voltage due to the IGBT switching at 50 kHz. The input current waveshape and the ripples on it are due to nature of the average current control technique used in implemented circuit. The recording in figure 7.20 shows that the input voltage and current are nearly in phase. Hence, nearly unity power factor operation is achieved. The rectifier's input current, voltage, input power and power factor are measured as 4.7590 A rms, 219.49 V rms, 1044.6 VA and 98.7 % respectively by HIOKI 3194 Motor/Harmonic Hi Tester. Its output power is also measured by same equipment and is 1004.8 W. Therefore; the rectifier's efficiency is nearly 96.2 % at steady state.

The rectifier's output voltage is measured by the scope in the averaging mode. It is shown in green color in figure 7.21. Also, the input current measured in averaging mode is shown in yellow color in the figure. The output voltage swings between maximum 437.8 V and minimum 433.5 V at steady state. Its output voltage ripple is calculated by dividing the difference between the maximum and minimum voltages to the rectifier's average output voltage. The output voltage ripple is less



Figure 7.20 Proposed rectifier's input current and input voltage waveform with 1 kW nearly purely resistive load.



Figure 7.21 Proposed rectifier's input current and input voltage waveform at 1 kW nearly purely resistive load.

than 1 % (0.99 %) according to the figure 7.21.

The rectifier's input current harmonics up to 50th are measured by HIOKI 3194 Motor/ Harmonic Hi Tester and they are given in table 7.3. The input current's FFT is obtained by using the scope at its 50 kHz center frequency and given in figure 7.22. Note that these harmonic numbers correspond to about 50 kHz switching frequency. The harmonics had relatively higher magnitudes near the switching frequency are obtained by using the FFT waveform and given also in this table. They are 997 and 999th harmonics and their magnitudes are relatively higher than the nearest harmonics and their magnitudes are shown in figure 7.22.

The input current's THD is calculated as 6.91 % from the equation 6.1 and the data correspond to the measurement in table 7.3. The THD is measured as 15.37 % by using the HIOKI 3194 Motor/Harmonic Hi Tester and considering up to first 3000th harmonics. This result indicates that for accurate prediction of THD, it is essential to consider all the harmonics up to the switching frequency.

The proposed rectifier is simulated by using PSIM simulation program and connecting 1 kW purely resistive load to the rectifier's output. Its input current and input voltage waveform are given in figure 7.23.

The rectifier's input current and input voltage are 4.85 A rms and 219.2 V rms according to the figure 7.23. Its input power and power factor are obtained by using "power factor meter" block in the PSIM simulation program. They are 1062.5 VA and 97.3 % respectively.

Figure 7.24 shows its output voltage with the mentioned conditions. Its steady state maximum, minimum and average output voltages are 436.98 V, 431.54 V and 434.28 V respectively. Hence, its output voltage ripple is 1.25 %. Its output power is obtained by using wattmeter block in the PSIM simulation program and it is 1011.9 W. Its efficiency is 95.24 % since its input power is 1062.5 VA.

The rectifier's input current harmonics up to 50th harmonic and the harmonics had high magnitudes near the switching frequency are obtained by using PSIM 6.1 simulation program and they are also listed in table 7.3.

The relevant FFT waveforms are given in figure 7.25. THD of the input current is calculated by using equation 6.1, the data in table 7.3 in simulation and it is nearly 22.28 %. THD of the input current is also calculated by using THD block in PSIM 6.1 and it is nearly 23.9 %. The THD block in PSIM 6.1 considers the input



Figure 7.22 FFT of proposed rectifier's input current when 1 kW purely resistive load is connected to the rectifier's output and 100 MHz DSO6014A scope's scale, offset, span, center frequency and sample rate are 20 dB/division, –48 dB, 100 kHz, 50 kHz and 200 kSa/s respectively.



Figure 7.23 Proposed rectifier's input current and input voltage waveform obtained using PSIM 6.1 simulation program at steady state when 1 kW purely resistive load is connected.



Figure 7.24 Proposed rectifier's output voltage waveform obtained using PSIM 6.1 simulation program at steady state when 1 kW purely resistive load is connected.





	Measured	Simulation		Measured	Simulation
Harmonic #	Magnitude (A)	Magnitude (A)	Harmonic #	Magnitude (A)	Magnitude (A)
1	6.5288	6.667	28	0.0031	0.000
2	0.0368	0.000	29	0.0257	0.022
3	0.2560	0.549	30	0.0054	0.000
4	0.0118	0.000	31	0.0308	0.012
5	0.1539	0.233	32	0.0068	0.000
6	0.0262	0.000	33	0.0695	0.006
7	0.1498	0.166	34	0.0084	0.000
8	0.0083	0.000	35	0.0683	0.018
9	0.2632	0.081	36	0.0060	0.000
10	0.0079	0.000	37	0.0606	0.018
11	0.2556	0.010	38	0.0061	0.000
12	0.0083	0.000	39	0.0267	0.007
13	0.2033	0.040	40	0.0063	0.000
14	0.0054	0.000	41	0.0278	0.008
15	0.1290	0.044	42	0.0058	0.000
16	0.0044	0.000	43	0.0380	0.015
17	0.0350	0.020	44	0.0068	0.000
18	0.0060	0.000	45	0.0421	0.012
19	0.0562	0.013	46	0.0060	0.000
20	0.0054	0.000	47	0.0364	0.002
21	0.0943	0.028	48	0.0058	0.000
22	0.0076	0.000	49	0.0128	0.011
23	0.1157	0.022	50	0.0054	0.000
24	0.0050	0.000	997	0.0966	0.267
25	0.0955	0.003	999	0.2738	0.934
26	0.0033	0.000	1001		0.893
27	0.0546	0.018	1003		0.260

Table 7.3 Proposed rectifier's input current harmonics up to 50th harmonic measured by using HIOKI 3194 Motor/Harmonic Hi Tester, 997th and 999th harmonic obtained by DSO6014A scope and the input current harmonics up to 50th harmonic and the harmonics had higher magnitudes near the 50 kHz switching frequency obtained by PSIM 6.1 simulation program with 1 kW load at steady state.

current's all harmonics in the calculation.

Important harmonics correspond to measurement part in table 7.3 are between 1st and 25th harmonic and near the 50 kHz switching frequency while the important harmonics correspond to simulation part are between 1st and 9th harmonic and near the switching frequency except even harmonics. The all even harmonics' magnitudes are very small for the parts correspond to the simulation and measurement part in table 7.3. The 1st harmonics in simulation part and the one in measurement part in

table 7.3 have similar magnitudes, but the other harmonics somewhat differ in the two part in this table due to non ideal behaviors of the mains supply and electrical components in the implemented circuit.

7.4.2 Proposed Rectifier's Experimental Results With 2 kW Resistive Load And Comparison With Simulations In Same Conditions

Similarly, the proposed rectifier is tested by connecting 2 kW nearly purely resistive load and using same test equipments in the laboratory environment. Its input current, input voltage and output voltage waveform, input power, power factor, input rms current, input rms voltage, output power, output mean voltage, output rms current and input current harmonics up to fifth harmonic are obtained in the laboratory.

The rectifier's input voltage in green color and input current in yellow color obtained by using scope are shown in figure 7.26. Its input current and voltage are 9.689 A rms and 218.99 V rms. Also, its input power and power factor are 2120.8 VA and 99.13 % respectively. These measurements are made by HIOKI Hi Tester. The rectifier's efficiency is 95.63 % because its output power is 2028.3 W at steady state.

The rectifier's output voltage in green color and input current obtained by using the scope in averaging mode are shown in figure 7.27. The output voltage has maximum 438.5 V, minimum 429.0 V and average 433.6 V values. Its output voltage ripple is about 2 % (2.19 %) and it is calculated similarly as in section 7.5.1.

The input current's FFT waveform is given in logarithmic scale in figure 7.28 and the harmonic that has relatively higher magnitude near the switching frequency is marked in the figure. The harmonic and first 50 harmonics measured with HIOKI 3194 are given in table 7.4 in measurement part. The harmonic corresponding to about 50 kHz switching frequency is 999th harmonic and its magnitude is 0.328 A.

THD of the rectifier's input current is calculated as nearly 8.72 % by using equation 6.1 and the values listed in table 7.4 in measurement part. The THD is more precisely measured as 11.95 % by using the HIOKI 3194 Motor/Harmonic Hi Tester



Figure 7.26 Proposed rectifier's input current and input voltage waveform with 2 kW nearly purely resistive load.



Figure 7.27 Proposed rectifier's input current and input voltage waveform with 2 kW nearly purely resistive load.


Figure 7.28 FFT of proposed rectifier's input current when 2 kW purely resistive load is connected to the rectifier's output and 100 MHz DSO6014A scope's scale, offset, span, center frequency and sample rate are 10 dB/division, -20 dB, 100 kHz, 50 kHz and 200 kSa/s respectively.

and considering up to first 3000th harmonics.

The rectifier's input current and input voltage waveform obtained via PSIM 6.1 simulation program are shown in figure 7.29, and their values are 9.523 A rms and 219.2 V rms respectively. Its input power and output power are 2087.64 VA and 2023.6 W respectively. Hence, its efficiency is nearly 96.93 %. Also, its input power factor is 98.57 %.

Its average output voltage obtained by its output voltage waveform shown in figure 7.30 is 434.28 V. Maximum and minimum voltages in figure 7.30 are 439.74 V and 428.72 V respectively. Thus, its output voltage ripple is nearly 2.54 %.

FFT waveform of the proposed rectifier's input current is obtained by using the simulation program and it is given in figure 7.31. The harmonics that are up to 50th harmonic and have higher magnitudes near at the switching speed are obtained by using the figure 7.31, and given in the part corresponds to the simulation in table



Figure 7.29 Proposed rectifier's input current and input voltage waveform obtained using PSIM 6.1 simulation program at steady state when 2 kW purely resistive load is connected.



Figure 7.30 Proposed rectifier's output voltage waveform obtained by using PSIM 6.1 simulation program at steady state when 2 kW purely resistive load is connected.

7.4. The rectifier's input current's THD is calculated as 16.79 % by using them. THD block in PSIM 6.1 measured its input current's THD as 17.0 % considering on the current's all harmonics. This value is more correct.



Figure 7.31 (a) FFT waveform of proposed rectifier's input current obtained by using PSIM 6.1 simulation program between 0 and 50500 Hz at steady state when 2 kW purely resistive load is connected, (b) the FFT waveform between 47 and 55 kHz.

The harmonics between the 1st and 35th harmonic and the one near the 50 kHz switching frequency are important in measurement part in table 7.4. However, all even harmonics except the 2nd harmonic don't have important magnitudes in the part in table 7.4. On the other hand, the harmonics between 1st and 31st harmonic and near the switching frequency have important magnitudes except the even harmonics in simulation part in this table. There are some differences between the harmonic magnitudes in simulation part and measurement part in table 7.4. Cause of the differences is same with the differences between the harmonic magnitudes in the parts correspond to the simulation and measurement in table 7.3.

	Measured	Simulation		Measured	Simulation
Harmonic #	Magnitude (A)	Magnitude (A)	Harmonic #	Magnitude (A)	Magnitude (A)
1	12.928	13.278	28	0.011	0.001
2	0.117	0.004	29	0.023	0.095
3	0.364	1.332	30	0.012	0.000
4	0.012	0.001	31	0.047	0.094
5	0.448	0.766	32	0.014	0.001
6	0.033	0.001	33	0.099	0.038
7	0.342	0.580	34	0.009	0.001
8	0.011	0.000	35	0.099	0.031
9	0.479	0.284	36	0.018	0.001
10	0.020	0.001	37	0.066	0.071
11	0.458	0.015	38	0.012	0.000
12	0.008	0.001	39	0.034	0.063
13	0.303	0.196	40	0.011	0.001
14	0.011	0.000	41	0.050	0.017
15	0.190	0.222	42	0.011	0.001
16	0.019	0.001	43	0.054	0.033
17	0.033	0.118	44	0.012	0.000
18	0.014	0.001	45	0.054	0.058
19	0.102	0.030	46	0.014	0.000
20	0.008	0.001	47	0.046	0.045
21	0.157	0.132	48	0.012	0.001
22	0.014	0.000	49	0.016	0.011
23	0.173	0.142	50	0.009	0.001
24	0.019	0.001	997		0.279
25	0.137	0.070	999	0.328	0.980
26	0.008	0.001	1001		0.945
27	0.071	0.030	1003		0.282

Table 7.4 Proposed rectifier's input current harmonics up to 50th harmonic measured by using HIOKI 3194 Motor/Harmonic Hi Tester, 999th harmonic obtained by DSO6014A scope and its harmonics up to 50th harmonic, harmonics that have higher magnitudes near at switching speed obtained by using PSIM 6.1 with 2 kW load at steady state.

7.5 Conclusion

The proposed rectifier's circuit block is given and it is explained block by block in this chapter. The rectifier's schematic and printed circuit board is shown. Experimental results and simulation results in same conditions are also given. They are listed in table 7.5.

	Simulation Circuit (1 kW load)	Implemented Circuit (1 kW load)	Error Magnitude (%)	Simulation Circuit (2 kW load)	Implemented Circuit (2 kW load)	Error Magnitude (%)
Efficiency (%)	95.24	96.2	0.96	96.93	95.63	1.3
Power Factor	0.973	0.987	1.4	0.9857	0.9913	0.56
Output Voltage Ripple (%)	1.25	0.99	0.26	2.54	2.19	0.35
THD (%)	23.9	15.37	8.53	17.0	11.95	5.05

 Table 7.5
 Simulation and implemented rectifier circuit results with different loads.

The THD values in this table are the ones measured by the THD block in PSIM 6.1 for the simulation circuit with 1 kW and 2 kW purely resistive load while the THD values measured by the THD measurement menu in HIOKI 3194 Motor Harmonic/ Hi Tester for the implemented circuit with the two loads are entered in this table since their measurements are very accurate.

Implemented rectifier with 1 kW load has higher efficiency, power factor, lower output voltage ripple and THD than the simulated circuit. Simulated circuit contains same snubbers and power semiconductors parameters as the implemented circuit. However, fall time and rise time parameters can not be entered in power semiconductors parameter part in PSIM 6.1 simulation software. Also, the boost inductors, L1 and L2 in figure 6.8 are ideal inductors in simulation circuit while the inductor designed by using amorphous metal core in this chapter is used in implemented circuit. UC3854B IC drives the IGBT better in implemented circuit than the approximate model in figure 6.1 used in the simulations. Hence, actual circuit has better performance than the simulation circuit with 1 kW load. However, as the load is increased, I^2R losses in the boost inductors and power carrying conductors in real circuit increase. Thus, the simulation circuit with 2 kW load has

higher efficiency than the real rectifier circuit with same load. On the other hand, the real rectifier circuit has better performance than the simulation circuit considering power factor, THD and output voltage ripple with 2 kW load.

However, the error magnitudes (%) between the simulation circuit's results and implemented circuit's ones given in table 7.5 are less than 1.5 % for the circuits with 1 kW and 2 kW purely resistive loads, but the THD errors between their results are less than 9 %. The causes of the errors are not only that UC3854B IC drives the IGBT better in the implemented circuit than the approximate model in figure 6.1 used in the simulations but the measurement equipments' measurements errors. On the other hand, the errors between the simulation and implemented circuit's results are very small. These results clearly indicate that performance of the proposed circuit can be investigated and its comparison with the conventional circuit can be reliably made in the simulation environment. In this way, considerable time saving can be achieved in improving the circuit.

CHAPTER 8

CONCLUSION

8.1 General

Electrical power consumers are generally a pollutant of AC electrical power distribution network. As number of electrical power consumer increases, the distortion over electrical power distribution network becomes very important. Total harmonic distortion of unity power factor rectifiers is very small, they are used for reducing overall distortion on the AC electrical power distribution network.

Unity power factor rectifiers use energy as efficient as possible since mains supply seems unity power factor rectifiers nearly a resistance whether their load is inductive, capacitive or purely resistive. This means that their reactive powers at the input sides of the rectifiers are nearly zero.

The unity power factor rectifiers' efficiency depends on their topologies. There are various unity power factor rectifier topologies. Research upon higher efficiency unity power factor rectifier topologies is very popular.

In this thesis, the main objective is to study unity power factor rectifiers and if possible develope a new single transistor unity power factor rectifier topology.

Fist of all, some unity power factor rectifier topologies are investigated. Then, one of them is selected for further study.

Afterwards, a new unity power factor topology is proposed and compared with the selected unity power factor rectifier topology.

Control techniques for power factor converters are investigated to find a

proper control technique. Average current control technique is selected to use in the selected unity power factor rectifier and proposed unity power factor rectifier's simulations and their implementations.

The two rectifier topologies are compared with each other on the basis of efficiency, output voltage ripple, total harmonic distortion, input curent ripple, etc. by simulating them.

Also, the two rectifiers are implemented on a same PCB (Printed Circuit Board). The implemented proposed rectifier is tested in a laboratory and it is simulated with 1 kW and 2 kW purely resistive load by using PSIM 6.1 simulation program. The implemented proposed rectifier's measured results and the simulation results are compared with each other in this thesis.

8.2 Experimental And Simulation Results

Simulation speed is very important, so first a good model is sought for the controller IC, namely UC3854B. PSIM 6.1 has already UC3854B IC model in its library. This model is simplified to increase the simulation speed, so two UC3854B IC models are developed. The IC model given in figure 6.1 is for PSIM 6.1 simulation program and the other one given in figure 6.2 is for MATLAB-Simulink R14 simulation program.

The classical unity power factor rectifier is simulated within 0.5 s with 3 kW purely resistive load by using PSIM 6.1 with the IC model in figure 6.1, PSIM 6.1 with the IC model in its library, PSIM MATLAB co-simulation with the IC model in figure 6.1, MATLAB-Simulink R14 with the IC model in figure 6.3. Their simulation speeds are 9.5 minutes, 19 minutes, 45 minutes and 2.5-3 hours respectively. PSIM 6.1 with the IC model in figure 6.1 is selected to use in all simulations made through this thesis work due to its higher speed.

Classical and proposed unity power factor rectifiers are designed according to the specifications given in section 6.1. Their output capacitance and boost inductance are calculated as 1.5 mF and 500 uH. The boost inductor is designed by using an amorphous metal core and its design is given in section 7.2.1 in detail.

The proposed rectifier is implemented and tested in a laboratory with 1 kW and 2 kW purely resistive loads and their results are measured. Also, the proposed rectifier are simulated in nearly same conditions with the tested circuit in a laboratory. The proposed rectifier's simulation circuit has same snubbers and power semiconductors parameters as the implemented circuit. However, only conduction voltage drop parameter can be entered in power semiconductors parameter part in PSIM 6.1 simulation software and this is made in the simulation circuit. Also, UC3854B IC model in figure 6.1 is used in the simulation circuit, and ideal boost inductors are used in this circuit while the inductor designed by using amorphous metal core in section 7.2.1 is used in implemented circuit.



Figure 8.1 Proposed UPF rectifier's implemented and simulation circuits' performance diagrams with 1 kW purely resistive load at steady state.



Figure 8.2 Proposed UPF rectifier's implemented and simulation circuits' performance diagrams with 2 kW purely resistive load at steady state.

The proposed rectifier's implemented circuit and simulation circuit's results with 1 kW and 2 kW purely resistive loads at steady state are given in figure 8.1 and figure 8.2 respectively. The implemented circuit has better input power factor, input current's THD and output voltage ripple with 1 kW and 2 kW purely resistive loads than the simulation circuit. This shows that UC3854B IC drives the IGBT better in implemented circuit than the approximate model in figure 6.1 used in the simulations. On the other hand, implemented circuit has higher efficiency than the simulation circuit with 1 kW purely resistive load while simulation circuit has higher efficiency than the implemented circuit with 2 kW purely resistive load because I^2R losses in the boost inductors and power carrying conductors in real circuit increase as the load is increased.

However, the errors between the simulation and implemented circuit's results are less than 1.5 %, but the THD errors between their results are less than 9 %. This shows that the simulations are found to be in an agreement with measurements and

the approximate UC3854B IC model in figure 6.1 is correct. Therefore, the rest of the study is done via simulations.

The implemented circuit's measured performance results are within the design specifications given in section 6.1. This indicates that proposed rectifier design is correctly made.

The performances of the two topologies are studied with 3 kW purely resistive load and inductive-resistive loads in the simulation environment. Classical and proposed unity power factor rectifier are compared with each other on the basis of their efficiency, input power factor, output voltage ripple and input current's THD (Total Harmonic Distortion) by simulating them using PSIM 6.1 simulation software with the UC3854B IC model in figure 6.1.



Figure 8.3 Classical and proposed UPF rectifier's performance diagrams obtained via simulation with 3 kW purely resistive load at steady state.

In these simulations, the rectifiers' output power, output voltage, mains supply's voltage, mains supply's frequency, and switching frequency are taken as 3 kW, 440 V, 220 V rms, 50 Hz, 50 kHz respectively. The simulation circuit used in these simulations doesn't contain any snubber. Ideal boost inductors are used in these simulations.

The classical UPF rectifier and proposed converter's performances are summarized in figure 8.3 for a 3 kW resistive load. It can be observed that the proposed unity power factor rectifier efficiency is higher than the classical UPF rectifier but not significiantly. However, the classical UPF rectifier's input power factor, input current's THD and output voltage ripple are better than the proposed rectifier.



Figure 8.4 Classical and proposed UPF rectifier's performance diagram with 3 kW inductive-resistive load (100 uH).

The two rectifiers give similar results in simulations made in this thesis work with 3 kW purely resistive load and inductive-resistive loads. The rectifiers' simulation results with 3 kW purely resistive load (100 uH) are given in figure 8.4 and the figure proves this. Hence, the study is pursued with resistive load alone.

The results given in figure 8.3 and figure 8.4 also show that the classical and proposed unity power factor rectifiers' design methodology and their design are correct since their simulation results are within their design specifications given in section 6.1.

The proposed rectifier's experimental and simulation results with purely resistive loads are combined and its efficiency, input power factor, input current's THD variations with different resistive loads are obtained. They are given before in figure 6.13, figure 6.14 and figure 6.15 respectively. According to these figures, these performances of the proposed rectifier's implemented circuit and simulation circuit become better as the load is increased to its 3 kW nominal value, but the implemented rectifier's efficiency with 2 kW purely resistive load is 0.57 % less than the one with 1 kW purely resistive load since I²R losses in the implemented circuit increases.

The two rectifiers are simulated in dynamic conditions. The proposed and classical unity power factor rectifier's settling times with respect to their output voltages are 47.61 ms and 47.65 ms respectively according to the simulations. Their settling times are nearly same.

8.3 Conclusion And Future Work

A new single switching transistor unity power factor rectifier topology is developed in this thesis. The classical UPF rectifier needs one free-wheeling diode parallely connected to its boost inductor and boost diode to reset the boost inductor's core while the proposed rectifier doesn't need this free-wheeling diode since the proposed rectifier's boost inductors are placed at its AC side. However, the proposed rectifier uses one extra boost diode. Thus, these two rectifier topologies have the same number of power diodes and switching power transistor. On the other hand, this proposed rectifier and the classical unity power factor rectifier give similar performance results, so more investigation up on this proposed rectifier topology is not necessary.

In this thesis work, modeling of a power factor corrector IC, namely UC3854B IC and implementation of the unity power factor rectifier are learned. These experiences will be used for development of a SEPIC converter in TUBITAK-BAP project whose project number is 105E053.

REFERENCES

[1] Ramesh Srinivasan, Ramesh Oruganti, "A Unity Power Factor Converter Using Half-Bridge Boost Topology", IEEE TRANSACTIONS ON POWER ELECTRONICS, VOL. 13, NO. 3, MAY 1998.

[2] I. Çadırcı, S.Varma, M. Ermiş, T. Gülsoy, "A 20 kW, 20 kHz Unity Power Factor Boost Converter for Three-Phase Motor Drive Applications from an Unregulated Single-Phase Supply", IEEE Transactions on Energy Conversion, Vol. 14, No. 3, September 1999.

[3] Jee-Woo Lim, Bong-Hwan Kwon, "A Power-Factor Controller for Single-Phase PWM Rectifiers", IEEE TRANSACTIONS ON INDUSTRIAL ELECTRONICS, VOL. 46, NO. 5, OCTOBER 1999.

[4] Yungtaek Jang, Robert W. Erickson, "New Single-Switch Three-Phase High-Power-Factor Rectifiers Using Multiresonant Zero-Current Switching", IEEE TRANSACTIONS ON POWER ELECTRONICS, VOL. 13, NO. 1, JANUARY 1998.

[5] Ali I. Maswood, Shen Wei, Zhou Jiarrgyo, "Near Unity Input Displacement Factor for Voltage Source PWM Drives", IEEE, 2001.

[6] M. Tou, K. Al-Haddad, G. Oliver and V. Rajagopalan, "Analysis and Design of Single Controlled Switch Three Phase Rectifier with Unity Power Factor and Sinusoidal Input Current", IEEE, 1995.

[7] C. A. Gallo, J.A. Corrêa Pinto, L. C. de Freitas, V.J. Farias, E. A. A. Coelho and J.B. Vieira Jr, "An Unity High Power Factor Power Supply Rectifier Using A PWM AC/DC Full Bridge Soft – Switching", IEEE, 2002.

[8] Karel Jezernik, "VSS Control of Unity Power Factor", IEEE TRANSACTIONS ON INDUSTRIAL ELECTRONICS, VOL. 46, NO. 2, APRIL 1999.

[9] J. Fernando Silva, J.F. Martins, "Space Vector Alfa Beta Current Regulator for Sliding Mode Controlled Unity Power Factor PWM Rectifiers", IEEE, 1998.

[10] Johann Miniböck, Johann W. Kolar, "Wide Input Voltage Range High Power Density High Efficiency 10 kW Three-Phase Three-Level Unity Power Factor PWM Rectifier", IEEE, 2002.

[11] J. Fernando Silva, "Sliding-Mode Control of Boost-Type Unity-Power-Factor PWM Rectifiers", IEEE TRANSACTIONS ON INDUSTRIAL ELECTRONICS, VOL. 46, NO. 3, JUNE 1999.

[12] Doron Shmilovitz, Dariusz Czarkowski, Zivan Zabar, Soo-Yeub Yoo, "A Novel Reversible Boost Rectifier With Unity Power Factor", IEEE, 1999.

[13] Mats Karlsson, Christer Thoren and Tadeus Wolpert, "A novel approach to the design of three-phase AC/DC power converters with Unity Power Factor", IEEE, 1999.

[14] Guan-Chyun Hsieh, Chien-Ming Wang, "ZCS-PWM Full-Wave Boost Rectifier with Unity Power Factor and Low Conduction Losses", IEEE TRANSACTIONS ON INDUSTRIAL ELECTRONICS, VOL. 46, NO. 4, AUGUST 1999.

[15] Li Zhang and H R Tayebi, "Optimal Control of a CSI Induction Machine Drive for High Performance Current Regulation and Unity Power Factor", IEEE, 1998.

[16] M.A. de Rooij, J.A. Ferreira, J.D. van Wyk, "A Three Phase, Soft Switching, Transformer Isolated, Unity Power Factor Front End Converter", IEEE, 1998.

[17] Jose R. Espinoza, Geza Joos, Luis A. Cea, Lautaro D. Salazar, "Space Vector Modulation Techniques for Current Source Rectifiers Operating at Switching Frequencies up to 1 kHz", IEEE, 1998.

[18] Chongming Qiao, Keyue M. Smedley, Zheren Lai, Mehmet Nabant, "An Improved Integration-Reset Controlled Single Phase Unity-Power-Factor Boost Rectifier with Lower Distortion", IEEE, 1999.

[19] Cicero Marcos Tavares Cruz, Ivo Barbi, "Unit Power Factor Active Clamping Single Phase Three Level Rectifier", IEEE, 2001.

[20] Joan Salaet, Salvador Alepuz, Alex Gilabert, Josep Bordonau, Juan Peracaula, "D-Q Modeling and Control of a Single-Phase Three-Level Boost Rectifier with Power Factor Correction and Neutral-Point Voltage Balancing", IEEE, 2002.

[21] H. Kanaan, H.F. Blanchette, K. Al-Haddad, R. Chaffai, L. Duguay, F. Fnaiech, "A New Multiple Loops Linear Control Scheme Applied to A Current-Injection Three Phase Unity-Power-Factor Rectifier", IEEE, 2002.

[22] Alexandre Ferrari de Souza, Ivo Barbi, "A New ZVS Semiresonant High Power Factor Rectifier with Reduced Conduction Losses", IEEE TRANSACTIONS ON INDUSTRIAL ELECTRONICS, VOL. 46, NO. 1, FEBRUARY 1999.

[23] H. Maçbahi, J. Xu, A. Cheriti, V. Rajagopalan, "A Soft-Switched SEPIC Based AC-DC Converter with Unity Power Factor and Sinusoidal Input Current", IEEE, 1998.

[24] Shinji Sato, Yutaka Suehiro, Masayoshi Yamamoto, Mutsuo Nakaoka, "High Power-Factor 3-Phase PWM Rectifier", IEEE, 2002.

[25] Fabio T. Wakabayashi, Marcelo J. Bonata, and Carlos A. Canesin, "Novel High-Power-Factor ZCS-PWM Preregulators", IEEE TRANSACTIONS ON INDUSTRIAL ELECTRONICS, VOL. 48, NO. 2, APRIL 2001.

[26] Zaohong Yang, Paresh C. Sen, "A Novel Technique to Achieve Unity Power Factor and Fact Transient Response in AC-to-DC Converters", IEEE TRANSACTIONS ON POWER ELECTRONICS, VOL. 16, NO. 6, NOVEMBER 2001.

[27] Jinrong Qian, Fred C. Lee, "Voltage-Source Charge-Pump Power-Factor-Correction AC/DC Converters", IEEE TRANSACTIONS ON POWER ELECTRONICS, VOL. 14, NO. 2, MARCH 1999.

[28] Michael Lan Zhang, Bin Wu, Yuan Xiao, Frank A. Dewinter, Reza Sotudeh, "A Multilevel Buck Converter Based Rectifier With Sinusoidal Inputs and Unity Power Factor for Medium Voltage (4160–7200 V) Applications", IEEE TRANSACTIONS ON POWER ELECTRONICS, VOL. 17, NO. 6, NOVEMBER 2002.

[29] In-Dong Kim and Bimal K. Bose, "A New ZCS Turn-on and ZVS Turn-off Unity Power Factor PWM Rectifier with Reduced Conduction Loss and No Auxiliary Switches", IEEE, 1998.

[30] Henrique A. C. Braga, Ivo Barbi, "A 3-kW Unity-Power-Factor Rectifier Based on a Two-Cell Boost Converter Using a New Parallel-Connection Technique", IEEE TRANSACTIONS ON POWER ELECTRONICS, VOL. 14, NO. 1, JANUARY 1999.

[31] Oscar L'opez, Luis Garc'ıa de Vicu^{*}na, Miguel Castilla, Jos'e Matas, Mariano L'opez, "Sliding-Mode-Control Design of a High-Power-Factor Buck-Boost Rectifier", IEEE TRANSACTIONS ON INDUSTRIAL ELECTRONICS, VOL. 46, NO. 3, JUNE 1999.

[32] Ivan Eidt Colling, Ivo Barbi, "Reversible Unity Power Factor Step-Up/Step-Down AC–DC Converter Controlled by Sliding Mode", IEEE TRANSACTIONS ON POWER ELECTRONICS, VOL. 16, NO. 2, MARCH 2001.

[33] L. Rossetto, G. Spiazzi, P. Tenti, "Control Techniques For Power Factor Correction Converters", Department of Electrical Engineering, University of Padova-ITALY, 1994.

[34] Jian Sun, Wei-Chung Wu, Bass, R.M., "Large-signal characterization of singlephase PFC circuits with different types of current control", Applied Power Electronics Conference and Exposition, APEC, 1998.

[35] L. Rossetto, G. Spiazzi, P. Tenti, "High-Quality Rectifiers With High-Frequency Insulation-An Overview", Department of Electrical Engineering, University of Padova-ITALY, 1995.

[36] Colonel Wm. T. McLyman, "Transformer and Inductor Design Handbook", MARCEL DEKKER INC, 1978.

[37] J. W. McClean, "Inductor Design Using Amorphous Metal C-Cores", IEEE 1996.

[38] "Power Factor Correction Inductor Design For Switched Mode Power Supplies Using Metglas Powerlite C-Cores", Metglas Application Guide.

[39] Ned Mohan, Tore M. Undeland, William P. Robbins, "Power Electronics Converters, Applications and Design", John Wiley & SONS, Second Edition.

[40] William P. Robbins, "Snubber Circuits", Lecture Notes, Dept. Of Electrical Engineering, University of Minnesota, 1997.

[41] Bary W Williams, "Power Electronics Devices, Drivers, Applications, and Passive Components".

[42] Bill Andreycak, "UC3854A and UC3854B Advanced Power Factor Correction Control ICs" Texas Instruments Application Information DN-44, 2005.

[43] M. R. Sahid, N. A. Azli, " Study On The Performance Of The Boost Power Factor Correction (PFC) Circuit With Variable Inductor Current Sense Resistor Values", IEEE, 2003.

APPENDIX A: UC3854 Controlled Power Factor Correction Circuit Design



U-134

UC3854 Controlled Power Factor Correction Circuit Design

PHILIP C. TODD

ABSTRACT

This Application Note describes the concepts and design of a boost preregulator for power factor correction. This note covers the important specifications for power factor correction, the boost power circuit design and the UC3854 integrated circuit which controls the converter. A complete design procedure is given which includes the tradeoffs necessary in the process. This design procedure is directly applicable to the UC3854A/B as well as the UC3854. The recommendations in Unitrode Design Note DN-39 cover other areas of the circuit and, while not discussed here, must be considered in any design. This application note supersedes Application Note U- 125 "Power Factor Correction With the UC3854."

INTRODUCTION

The objective of active power factor correction is to make the input to a power supply look like a simple resistor. An active power factor corrector does this by programming the input current in response to the input voltage. As long as the ratio between the voltage and current is a constant the input will be resistive and the power factor will be 1.0. When the ratio deviates from a constant the input will contain phase displacement, harmonic distortion or both and either one will degrade the power factor.

The most general definition of power factor is the ratio of real power to apparent power.

$$PF = \frac{P}{(Vrms \times Irms)}$$
 or $PF = \frac{Watts}{V.A.}$

Where P is the real input power and Vrms and Irms are the root mean square (RMS) voltage and current of the load, or power factor corrector input in this case. If the load is a pure resistance the real power and the product of the RMS voltage and current will be the same and the power factor will be 1.0. If the load is not a pure resistance the power factor will be below 1.0.

Phase displacement is a measure of the reactance of the input impedance of the active power factor corrector. Any amount of reactance, either inductive or capacitive will cause phase displacement of the input current waveform with respect to the input voltage waveform. The phase displacement of the voltage and current is the classic definition of power factor which is the cosine of the phase angle between the voltage and current sinusoids.

$PF = Cos \theta$

The amount of displacement between the voltage and current indicates the degree to which the load is reactive. If the reactance is a small part of the impedance the phase displacement will be small. An active power factor corrector will generate phase displacement of the input current if there is phase shift in the feedforward signals or in the control loops. Any filtering of the AC line current will also produce phase displacement.

Harmonic distortion is a measure of the non-linearity of the input impedance of the active power factor corrector. Any variation of the input impedance as a function of the input voltage will cause distortion of the input current and this distortion is the other contributor to poor power factor. Distortion increases the RMS value of the current without increasing the total power being drawn. A non-linear load will therefore have a poor power factor because the RMS value of the current is high but the total power delivered is small. If the non-linearity is small the harmonic distortion will be low. Distortion in an active power factor corrector comes from



Harmonic Order	Permissible current	Maximum permissible current			
n	mA/W	А			
Odd harmonics					
3	3.4	2.30			
5	1.9	1.14			
7	1.0	0.78			
9	0.5	0.40			
11	0.35	0.33			
13	0.3	0.21			
15 up	3.85/n	$0.15 \times \frac{15}{n}$			
Even harmonics					
2	1.8	1.08			
4	0.7	0.42			
6	0.5	0.30			
>8	$\frac{3}{n}$	<u>1.80</u> n			

Table 1

several sources: the feedforward signals, the feedback loops, the output capacitor, the inductor and the input rectifiers.

An active power factor corrector can easily achieve



a high input power factor, usually much greater than 0.9. But power factor is not a sensitive measure of the distortion or the displacement of the current waveform. It is often more convenient to deal with these quantities directly rather than with the power factor. For example, 3% harmonic distortion alone has a power factor of 0.999. A current with 30% total harmonic distortion still has a power factor or of 0.95. A current with a phase displacement of 25 degrees from the voltage has a power factor of 0.90.

The trend among the world standards organizations responsible for power quality is to specify maximum limits for the amount of current allowed at each of the harmonics of the line frequency. IEC 555-2 specifies each harmonic up through and beyond the 15th and the amount of current permissible at each. Table 1 lists the requirements for IEC 555-2 as of the time of this writing. There are two parts to the specification, a relative distortion and an absolute distortion maximum. Both limits apply to all equipment. This table is included here as an example of a line distortion specification. It is not intended to be used for design purposes. The IEC has not finalized the requirements of IEC 555 at this time and major changes are possible.

Active Power Factor Correction

A boost regulator is an excellent choice for the power stage of an active power factor corrector because the input current is continuous and this produces the lowest level of conducted noise and the best input current waveform. The disadvantage of the boost regulator is the high output voltage required. The output voltage must be greater than the highest expected peak input voltage.

The boost regulator input current must be forced or programmed to be proportional to the input voltage waveform for power factor correction. Feedback is necessary to control the input current and either

U-134



Basic Configuration of High Power Factor Control Circuit

peak current mode control or average current mode control may be used. Both techniques may be implemented with the UC3854. Peak current mode control has a low gain, wide bandwidth current loop which generally makes it unsuitable for a high performance power factor corrector since there is a significant error between the programming signal and the current. This will produce distortion and a poor power factor. concept. An amplifier is used in the feedback loop around the boost power stage so that input current tracks the programming signal with very little error. This is the advantage of average current mode control and it is what makes active power factor correction possible. Average current mode control is relatively easy to implement and is the method described here.

A block diagram of a boost power factor corrector circuit is shown in Figure 1. The power circuit of a



Figure 2. Preregulator Waveforms

boost power factor corrector is the same as that of a dc to dc boost converter. There is a diode bridge ahead of the inductor to rectify the AC input voltage but the large input capacitor which would normally be associated with the AC to DC conversion function has been moved to the output of the boost converter. If a capacitor follows the input diode bridge it is a small one used only for noise control.

The output of the boost regulator is a constant voltage but the input current is programmed by the input voltage to be a half sine wave. The power flow Control Circuits

An active power factor corrector must control both the input current and the output voltage. The current loop is programmed by the rectified line voltage so that the input to the converter will appear to be resistive. The output voltage is controlled by changing the average amplitude of the current programming signal. An analog multiplier creates the current programming signal by multiplying the rectified line voltage with the output of the voltage er-



Figure 3. High Power Factor

into the output capacitor is not constant but is a sine wave at twice the line frequency since power is the instantaneous product of voltage an current. This is shown in Figure 2. The top waveform shows the voltage and the current into the power factor corrector and the second waveform shows the flow of energy into and out of the output capacitor. The output capacitor stores energy when the input voltage is high and releases the energy when the input voltage is low to maintain the output power flow. The third waveform in Figure 2 shows the charging and discharging current. This current has a different shape from the input current and is almost entirely at the second harmonic of the AC line voltage. This flow of energy into and out of the capacitor results in ripple voltage at the second harmonic also and this is shown in the fourth waveform in Figure 2. Note that the voltage ripple is displaced by 90 degrees relative to the current since this is reactive energy storage. The output capacitor must be rated to handle the second harmonic ripple current as well as the high frequency ripple current from the boost converter switch which modulates it.

ror amplifier so that the current programming signal has the shape of the input voltage and an average amplitude which controls the output voltage. Figure 3 is a block diagram which shows the basic control circuit arrangement necessary for an active power factor corrector. The output of the multiplier is the current programming signal and is called Imo for multiplier output current. The multiplier input from the rectified line voltage is shown as a current in Figure 3 rather than as a voltage signal because this is the way it is done in the UC3854.

Figure 3 shows a squarer and a divider as well as a multiplier in the voltage loop. The output of the voltage error amplifier is divided by the square of the average input voltage before it is multiplied by the rectified input voltage signal. This extra circuitry keeps the gain of the voltage loop constant, without it the gain of the voltage loop would change as the square of the average input voltage. The average value of the input voltage is called the feedforward voltage or Vff since it provides an open loop correction which is fed forward into the voltage loop. It is squared and then divided into the voltage error amplifier output voltage (Vvea).

The current programming signal must match the rectified line voltage as closely as possible to maximize the power factor. If the voltage loop bandwidth were large it would modulate the input current to keep the output voltage constant and this would distort the input current horribly. Therefore the voltage loop bandwidth must be less than the input line frequency. But the output voltage transient response must be fast so the voltage loop bandwidth must be made as large as possible. The squarer and divider circuits keep the loop gain constant so the bandwidth can be as close as possible to the line frequency to minimize the transient response of the output voltage. This is especially important for wide input voltage sna

The circuits which keep the loop gain constant make the output of the voltage error amplifier a power control. The output of the voltage error amplifier actually controls the power delivered to the load. This can be seen easily from an example. If the output of the voltage error amplifier is constant and the input voltage is doubled the programming signal will double but it will be divided by the square of the feedforward voltage, or four times the input, which will result in the input current being reduced to half its original value. Twice the input voltage times half the input current results in the same input power as before. The output of the voltage error amplifier, then, controls the input power level of the power factor corrector. This can be used to limit the maximum power which the circuit can draw from the power line. If the output of the voltage error amplifier is clamped at some value that corresponds to some maximum power level, then the active power factor corrector will not draw more than that amount of power from the line as long as the input voltage is within its range.

Input Distortion Sources

The control circuits introduce both distortion and displacement into the input current waveform. These errors come from the input diode bridge, the multiplier circuits and ripple voltage, both on the output and on the feedforward voltage.

There are two modulation processes in an active power factor corrector. The first is the input diode bridge and the second is the multiplier, divider, squarer circuit. Each modulation process generates cross products, harmonics or sidebands between the two inputs. The description of these mathematically can be quite complex. Interestingly enough, however, the two modulators interact and one becomes a demodulator for the other so that the result is quite simple. As shown later, virtually all of the ripple voltages in an active power factor corrector are at the second harmonic of the line frequency. When these voltages go through the multiplier and get programmed into the input current and then go through the input diode bridge the second harmonic voltage amplitude results in two frequency components. One is at the third harmonic of the line frequency and the other is at the fundamental. Both of these components have an amplitude which is half of the amplitude of the original second harmonic voltage. They also have the same phase as the original second harmonic. If the ripple voltage is 10% of the line voltage amplitude and is phase shifted 90 degrees the input current will have a third harmonic which is 5% of the fundamental and is shifted 90 degrees.

The feedforward voltage comes from the rectified AC line which has a second harmonic component that is 66% of the amplitude of the average value. The filter capacitors of the feedforward voltage divider greatly attenuate the second harmonic and effectively remove all of the higher harmonics but some of the second harmonic is still present at the feedforward input. This ripple voltage is squared by the control circuits as shown in Figure 3. This doubles the amplitude of the ripple since it is riding on top of a large DC value. The divider process is transparent to the ripple voltage so it passes on to the multiplier and eventually becomes third harmonic distortion of the input current and a phase displacement. The doubling action of the squarer means that the amplitude of the input current distortion in percent is the same as the amplitude of the ripple voltage, in percent, at the feedforward input.

Needless to say, the feedforward ripple voltage must be kept small to achieve a low distortion input current. The ripple voltage could be made small with a single pole filter with a very low cutoff frequency. However, fast response to changes of the input voltage is also desirable so the response time of the filter must be fast. These two requirements are, of course, in conflict and a compromise must be found. A two pole filter on the feedforward input has a faster transient response than a single pole filter for the same amount of ripple attenuation. Another advantage of the two pole filter has is that the phase shift is twice that of the single pole filter. This results in 180 degrees of phase shift of the second harmonic and brings both the resulting third harmonic and the displacement component of the input current back in phase with the voltage. A second harmonic ripple voltage of 3% at the feedforward input results in a 0.97 power factor just from the displacement component if a single pole filter is used for the feedforward voltage. With a two pole filter there is no displacement component to the power factor because it is in

phase with the input current. The third harmonic component of the input current resulting from the second harmonic at the feedforward input will have the same amplitude as the second harmonic ripple voltage. If 3% second harmonic is present on the feedforward voltage the line current waveform will contain 3% third harmonic distortion.

The output voltage has ripple at the second harmonic due to the ripple current flowing through the output capacitor. This ripple voltage is fed back through the voltage error amplifier to the multiplier and, like the feedforward voltage, programs the input current and results in second harmonic distortion of the input current. Since this ripple voltage does not go through the squarer the amplitude of the distortion and displacement are each half of the amplitude of the ripple voltage. The ripple voltage at the output of the voltage error amplifier must be in phase with the line voltage for the displacement component to be in phase. The voltage error amplifier must shift the second harmonic by 90 degrees so that it will be in phase with the line voltage



Figure 4. Cusp Distortion

The voltage loop of a boost converter with average current mode control has a control to output transfer function which has a single pole roll off characteristic so it could be compensated with a flat gain error amplifier. This produces a very stable loop with 90 degrees of phase margin. However, it provides less than optimum performance. The ripple voltage on the output capacitor is out of phase with the input current by 90 degrees. If the error amplifier has flat gain at the second harmonic frequency the distortion and displacement generated in the input current will be 90 degrees out of phase with the rectified AC line. The power factor can be improved by introducing phase shift into the voltage error amplifier response. This shifts the displacement component of the power factor back into alignment with the input voltage and increases the power factor. The amount of phase shift which can be added is determined by the need to keep the voltage loop stable. If the phase margin is reduced to 45 degrees the phase at the second harmonic will be very close to 90 degrees and this brings the displacement component back in phase with the input voltage.

The bandwidth of the voltage control loop is determined by the amount of input distortion to be contributed by the output ripple voltage. If the output capacitor is small and the distortion must be low then the bandwidth of the loop will be low so that the ripple voltage will be sufficiently attenuated by the error amplifier. Transient response is a function of the loop bandwidth and the lower the bandwidth the slower the transient response and the greater the overshoot. The output capacitor may need to be large to have both fast output transient response and low input current distortion.

The technique used to design the loop compensation is to find the amount of attenuation of the output ripple voltage required in the error amplifier and then work back into the unity gain frequency. The loop will have the maximum bandwidth when the phase margin is the smallest. A 45 degree phase margin is a good compromise which will give good loop stability and fast transient response and which is easy to design. The voltage error amplifier response which results will have flat gain up to the loop unity gain frequency and will have a single pole roll off above that frequency. This gives the maximum amount of attenuation at the second harmonic of the line frequency from a simple circuit, gives the greatest bandwidth and provides a 45 degree phase margin.

Cusp Distortion

Cusp distortion occurs just after the AC line input has crossed zero volts. At this point the amount of current which is required by the programming signal exceeds the available current slew rate. When the input voltage is near zero there is very little voltage across the inductor when the switch is closed so the current cannot ramp up very quickly so the available slew rate is too low and the input current will lag behind the desired value for a short period of time. Once the input current matches the programmed value the control loop is back in operation and the input current will follow the programming signal. The length of time that the current does not track the programmed value is a function of the inductor value. The smaller the inductor value the better the tracking and the lower the distortion but the smaller inductor value will have higher ripple current. The amount of distortion generated by this condition is generally small and is mostly higher order harmonics. This problem is minimized by a sufficiently high switching frequency.

UC3854 Block Diagram

A block diagram of the UC3854 is shown in Figure 5 and is the same as the one in the device data sheet. This integrated circuit contains the circuits necessary to control a power factor corrector. The UC3854 is designed to implement average current mode control but is flexible enough to be used for a wide variety of power topologies and control methods.

The top left corner of Figure 5 contains the under voltage lock out comparator and the enable comparator. The output of both of these comparators must be true to allow the device to operate. The inverting input to the voltage error amplifier is connected to pin 11 and is called Vsens. The diodes shown around the voltage error amplifier are intended to represent the functioning of the internal circuits rather than to show the actual devices. The diodes shown in the block diagram are ideal diodes and indicate that the non-inverting input to the error amplifier is connected to the 7.5Vdc reference voltage under normal operation but is also used for the slow start function. This configuration lets the voltage control loop begin operation before the output voltage has reached its operating point and eliminates the turn-on overshoot which plagues many power supplies. The diode shown between pin 11 and the inverting input of the error amplifier is also an ideal diode and is shown to eliminate confusion about whether there might be an extra diode drop added to the reference or not. In the actual device we do it with differential amplifiers. An internal current source is also provided for charging the slow start timing capacitor.

The output of the voltage error amplifier, Vvea, is available on pin 7 of the UC3854 and it is also an input to the multiplier. The other input to the multiplier is pin 6, lac, and this is the input for the programming wave shape from the input rectifiers. This pin is held at 6.0 volts and is a current input. The feedforward input, Vff, is pin 8 and its value is squared before being fed into the divider input of the multiplier. The lset current from pin 12 is also used in the multiplier to limit the maximum output current. The output current of the multiplier is lmo and it flows out of pin 5 which is also connected to the non-inverting input of the current error amplifier.

The inverting input of the current amplifier is connected to pin 4, the Isens pin. The output of the current error amplifier connects to the pulse width modulation (PWM) comparator where it is compared to the oscillator ramp on pin 14. The oscillator and the comparator drive the set-reset flip-flop which, in turn, drives the high current output on pin 16. The output voltage is clamped internally to the UC3854 at 15 volts so that power MOSFETs will not have their gates over driven. An emergency peak current limit is provided on pin 2 and it will shut the output pulse off when it is pulled slightly below ground. The reference voltage output is connected to pin 9 and the input voltage is connected to pin 15.

DESIGN PROCESS

Power Stage Design

This analysis of the power stage design makes use of a 250W boost converter as an example. The control circuit for a boost power factor corrector does not change much with the power level of the converter. A 5000 watt power factor corrector will have almost the same control circuits as a 50 watt



Figure 5. UC3854 Block Diagram



Figure 6. Complete Schematic of 250W Power Factor Preregulator

corrector. The power stage will be different but the design process will remain the same for all power factor corrector circuits. Since the design process is the same and the power stage is scalable a 250 watt corrector serves well as an example and it can be readily scaled to higher or lower output levels. Figure 6 is the schematic diagram of the circuit. Please refer to this schematic in the discussion of the design process which follows.

Specifications

The design process starts with the specifications for the converter performance. The minimum and maximum line voltage, the maximum output power, and the input line frequency range must be specified. For the example circuit the specifications are:

Maximum power output: 250W

Input voltage range: 80-270Vac

Line frequency range: 47-65Hz

This defines a power supply which will operate almost anywhere in the world. The output voltage of a boost regulator must be greater than the peak of the maximum input voltage and a value 5% to 10% higher than the maximum input voltage is recommended so the output voltage is chosen to be 400Vdc.

Switching Frequency

The choice of switching frequency is generally somewhat arbitrary. The switching frequency must be high enough to make the power circuits small and minimize the distortion and must be low enough to keep the efficiency high. In most applications a switching frequency in the range of 20KHz to 300KHz proves to be an acceptable compromise. The example converter uses a switching frequency of I00KHz as a compromise between size and efficiency. The value of the inductor will be reasonably small and cusp distortion will be minimized, the inductor will be physically small and the loss due to the output diode will not be excessive. Converters operating at higher power levels may find that a lower switching frequency is desirable to minimize the power losses. Turn-on snubbers for the switch will reduce the switching losses and can be very effective in allowing a converter to operate at high switching frequency with very high efficiency.

Inductor Selection

The inductor determines the amount of high frequency ripple current in the input and its value is chosen to give some specific value of ripple current. Inductor value selection begins with the peak





Figure 7

current of the input sinusoid. The maximum peak current occurs at the peak of the minimum line voltage and is given by:

l line (pk) =
$$\frac{\sqrt{2} \times P}{Vin (min)}$$

For the example converter the maximum peak line current is 4.42 amps at a Vin of 80Vac.

The maximum ripple current in a boost converter occurs when the duty factor is 50% which is also when the boost ratio M=Vo/Vin=2. The peak value of inductor current generally does not occur at this point since the peak value is determined by the peak value of the programmed sinusoid. The peak value of inductor ripple current is important for calculating the required attenuation of the input filter. Figure 7 is a graph of the peak to peak ripple current in the inductor versus input voltage for the example converter.

The peak-to-peak ripple current in the inductor is normally chosen to be about 20% of the maximum peak line current. This is a somewhat arbitrary decision since this is usually not the maximum value of the high frequency ripple current. A larger value of ripple current will put the converter into the discontinuous conduction mode for a larger portion of the rectified line current cycle and means that the input filter must be larger to attenuate more high frequency ripple current. The UC3854, with average current mode control, allows the boost stage to move between continuous and discontinuous modes of operation without a performance change.

The value of the inductor is selected from the peak current at the top of the half sine wave at low input voltage, the duty factor D at that input voltage and the switching frequency. The two equations necessary are given below:

$$D = \frac{Vo - Vin}{Vo}$$
$$L = \frac{Vin \times D}{fs \times \Delta I}$$

Where ΔI is the peak-to-peak ripple current. In the example 250W converter D=0.71, ΔI =900ma, and L=0.89mH. For convenience the value of L is rounded up to 1.0mH.

The high frequency ripple current is added to the line current peak so the peak inductor current is the sum of peak line current and half of the peak-to-peak high frequency ripple current. The inductor must be designed to handle this current level. For our example the peak inductor current is 5.0 amps. The peak current limit will be set about 10% higher at 5.5 amps.

Output Capacitor

The factors involved in the selection of the output capacitor are the switching frequency ripple current, the second harmonic ripple current, the DC output voltage, the output ripple voltage and the hold-up time. The total current through the output capacitor is the RMS value of the switching frequency ripple current and the second harmonic of the line current. The large electrolytic capacitors which are normally chosen for the output capacitor have an equivalent series resistance which changes with frequency and is generally high at low frequencies. The amount of current which the capacitor can handle is generally determined by the temperature rise. It is usually not necessary to calculate an exact value for the temperature rise. It is usually adequate to calculate the temperature rise due to the high frequency ripple current and the low frequency ripple current and add them together. The capacitor data sheet will provide the necessary ESR and temperature rise information.

The hold-up time of the output often dominates any other consideration in output capacitor selection. Hold-up is the length of time that the output voltage remains within a specified range after input power has been turned off. Hold-up times of 15 to 50 milliseconds are typical. In off-line power supplies with a 400Vdc output the hold-up requirement generally works out to between 1 and 2μ F per watt of output. In our 250W example the output capacitor is **450µF**. If hold-up is not required the capacitor will be much smaller, perhaps **0.2µF** per watt, and then ripple current and ripple voltage are the major concern.

Hold-up time is a function of the amount of energy stored in the output capacitor, the load power, output voltage and the minimum voltage the load will operate at. This can be expressed in an equation to define the capacitance value in terms of the holdup time.

$$Co = \frac{2 \times Pout \times \Delta t}{Vo^2 - Vo (min)^2}$$

Where Co is the output capacitor, Pout is the load power, Δt is the hold-up time, Vo is the output voltage and Vo(min) is the minimum voltage the load will operate at. For the example converter Pout is 250W, Δt is 64msec, Vo is 400V and Vo(min) is 300V so Co is **450µF**.

Switch and Diode

The switch and diode must have ratings which are sufficient to insure reliable operation. The choice of these components is beyond the scope of this Application Note. The switch must have a current rating at least equal to the maximum peak current in



Figure 8. Current Transformers Used with Negative Output



the inductor and a voltage rating at least equal to the output voltage. The same is true for the output diode. The output diode must also be very fast to reduce the switch turn-on power dissipation and to keep its own losses low. The switch and diode must have some level of derating and this will vary depending on the application.

For the example circuit the diode is a high speed, high voltage type with 35ns reverse recovery. 600Vdc breakdown, and 8A forward current ratings. The power MOSFET in the example circuit has a 500Vdc breakdown and 23Adc current rating. A major portion of the losses in the switch are due to the turn-off current in the diode. The peak power dissipation in the switch is high since it must carry full load current plus the diode reverse recovery current at full output voltage from the time it turns on until the diode turns off. The diode in the example circuit was chosen for its fast turn off and the switch was oversized to handle the high peak power dissipation. A turn on snubber for the switch would have allowed a smaller switch and a slightly slower diode.

Current Sensing

There are two general methods for current sensing, a sense resistor in the ground return of the converter or two current transformers. The sense resistor is the least expensive method and is most appropriate at low power or current levels. The power dissipation in the resistor may become quite large at higher current levels and in that case the current transformers are more appropriate. Two current transformers are required, one for the switch current and one for the diode current, to produce an analog of the inductor current as is required for average current mode control. The current transformers must operate over a very wide duty factor range and this can be difficult to achieve without saturating them. Current transformer operation is outside the scope of this paper but Unitrode has Design Note DN-41 which discusses the problem in some detail.

The current transformers may be configured for either a positive output voltage or a negative output voltage. In the negative output configuration, shown in Figure 8, the peak current limit on pin 2 of the UC3854 is easy to implement. In the positive output configuration, shown in Figure 9, this feature may be lost. It can be added back by putting another resistor in series with the ground leg of the current transformer which senses the switch current.

The configuration of the multiplier output and the current error amplifier are different depending on whether a resistor is used for current sensing or whether current transformers with positive output voltages are used for current sensing. Both work equally well and the configurations of the current error amplifier are shown in Figures 8 and 9 respectively. The positive output current transformer configuration requires the inverting input to the integrator be connected to the sense resistor and the resistor at the output of the multiplier be connected to ground. (see Figure 9) The voltage at the output of the multiplier is not zero but is the programming voltage for the current loop and it will have the half sine wave shape which is necessary for the current loop.

The resistor current sense configuration is used in the example converter (Figure 6) so the inverting input to the current error amplifier (pin 4) is connected to ground through Rci. The current error amplifier is configured as an integrator at low frequencies for average current mode control so the average voltage at the non-inverting input of the current error amplifier (pin 5, which it shares with the multiplier output) must be zero. The non-inverting input to the current error amplifier acts like a summing junction for the current control loop and adds the multiplier output current to the current from the sense resistor (which flows through the programming resistor Rmo). The difference controls the boost regulator. The voltage at the inverting input of the current error amplifier (pin 4) will be small at low frequencies because the gain at low frequencies is large. The gain at high frequencies is small so relatively large voltages at the switching frequency may be present. But, the average voltage on pin 4 must be zero because it is connected through Rci to ground.

The voltage across Rs, the current sense resistor in the example converter, goes negative with respect to ground so it is important to be sure that the pins of the UC3854 do not go below ground. The voltage across the sense resistor should be kept small and pins 2 and 5 should be clamped to prevent their going negative. A peak value of 1 volt or so across the sense resistor provides a signal large enough to have good noise margin but which is small enough to have low power dissipation. There is a great deal of flexibility in choosing the value of the sense resistor. A 0.25 ohm resistor was chosen for Rs in the example converter and at the worst case peak current of 5.6 amps gives a maximum voltage of 1.40V peak.

Peak Current Limit

The peak current limit on the UC3854 turns the switch off when the instantaneous current through it exceeds the maximum value and is activated when pin 2 is pulled below ground. The current limit value is set by a simple voltage divider from the reference voltage to the current sense resistor.

The equation for the voltage divider is given below:

$Rpk2 = \frac{Vrs \times Rpk1}{Vref}$

Where Rpk1 and Rpk2 are the resistors of the voltage divider, Vref is 7.5 volts on the UC3854, and Vrs is the voltage across the sense resistor Rs at the current limit point. The current through Rpk2 should be around 1 mA. The peak current limit in the example circuit is set at 5.4 amps with an Rpk1 of 10K and Rpk2 of 1.8K. A small capacitor, Cpk, has been added to give extra noise immunity when operating at low line and this also increases the current limit slightly.

Multiplier Set-up

The multiplier/divider is the heart of the power factor corrector. The output of the multiplier programs the current loop to control the input current to give a high power factor. The output of the multiplier is therefore a signal which represents the input line current.

Unlike most design tasks where the design begins at the output and proceeds to the input the design of the multiplier circuits must begin with the inputs. There are three inputs to the multiplier circuits: the programming current lac (pin 6) the feedforward voltage Vff from the input (pin 8) and the voltage error amplifier output voltage Vvea (pin 7). The multiplier output current is Imo (pin 5) and it is related to the three inputs by the following equation:

$$Imo = \frac{Km \times lac \times (Vvea - 1)}{Vff^2}$$

Where Km is a constant in the multiplier and is equal to 1.0, lac is the programming current from the rectified input voltage, Vvea is the output of the voltage error amplifier and Vff is the feedforward voltage.

Feedforward Voltage

Vff is the input to the squaring circuit and the UC3854 squaring circuit generally operates with a Vff range of 1.4 to 4.5 volts. The UC3854 has an internal clamp which limits the effective value of Vff to 4.5 volts even if the input goes above that value. The voltage divider for the Vff input has three resistors (Rff1, Rff2 and Rff3 - see Figure 6) and two capacitors (Cff1 and Cff2) and so it filters as well as providing two outputs. The resistors and capacitors of the divider form a second order low pass filter so the DC output is proportional to the average value of the input half sine wave. If the RMS value of the AC input voltage is 270Vac

the average value of a half sine will be 243Vdc and the peak will be 382V.

The Vff voltage divider has two DC conditions to meet. At high- input line voltage Vff should not be greater than 4.5 volts. At this voltage the Vff input clamps so the feedforward function is lost. The voltage divider should be set up so that Vff is equal to 1.414 volts when Vin is at its low line value and the upper node of the voltage divider, Vffc, should be about 7.5 volts. This allows Vff to be clamped as described in Unitrode Design Note DN-39B. There is an internal current limit which holds the multiplier output constant if the Vff input goes below 1.414 volts. The Vff input should always be set up so that Vff is equal to 1.414 volts at the minimum input voltage. This may cause Vff to clip on the high end of the input voltage range if there is an extremely wide AC line voltage input range. However, it is preferable to have Vff clip at the high end rather than to have the multiplier output clip on the low end of the range. If Vff clips the voltage loop gain will change but the effect on the overall system will be small whereas the multiplier clipping will cause large amounts of distortion in the input current waveform.

The example circuit uses the UC3854 so the maximum value of Vff is 4.5 volts. If Rff1, the top resistor of the divider, is 910K and Rff2, the middle resistor, is 91 K and Rff3, the bottom resistor, is 20K the maximum value of Vff will be 4.76 volts when the input voltage is 270Vac RMS and the DC average value will be 243 volts. When the input voltage is 80Vac RMS the average value is 72 volts and Vff is 1.41Vdc. Also at Vin=80Vac the voltage at the upper node on the voltage divider, Vffc, will be 7.83 volts. Note that the high end of the range goes above 4.5 volts so that the low end of the range will not go below 1.41 volts.

The output of the voltage error amplifier is the next piece of the multiplier setup. The output of the voltage error amplifier, Vvea, is clamped inside the UC3854 at 5.6 volts. The output of the voltage error amplifier corresponds to the input power of the converter. The feedforward voltage causes the power input to remain constant at given Vvea voltage regardless of line voltage changes. If 5.0V is established as the maximum normal operating level then 5.6V gives an overload power limit which is 12% higher.

The clamp on the output of the voltage error amplifier is what sets the minimum value of Vff at 1.414 volts. This can be seen by plugging these values into the equation for the multiplier output current given above. When Vff is large the inherent errors of the multiplier are magnified because Vvea/Vff becomes small. If the application has a wide input voltage range and if a very low harmonic distortion

is required then Vff may be changed to the range of 0.7 to 3.5 volts. To do this an external clamp MUST be added to the voltage error amplifier to hold its output below 2.00 volts. In general, however, this is not a recommended practice.

Multiplier Input Current

The operating current for the multiplier comes from the input voltage through Rvac. The multiplier has the best linearity at relatively high currents, but the recommended maximum current is 0.6mA. At high line the peak voltage for the example circuit is 382Vdc and the voltage on pin 6 of the UC3854 is 6.0Vdc. A 620K value for Rvac will give an lac of 0.6mA maximum. For proper operation near the cusp of the input waveform when Vin=0 a bias current is needed because pin 6 is at 6.0Vdc. A resistor, Rb1, is connected from Vref to pin 6 to provide the small amount of bias current needed. Rb1 is equal to Rvac/4. In the example circuit a value of 150K for Rb1 will provide the correct bias.

The maximum output of the multiplier occurs at the peak of the input sine wave at low line. The maximum output current from the multiplier can be calculated from the equation for Imo, given above, for this condition. The peak value of lac will be 182 microamps when Vin is at low line. Vvea will be 5.0 volts and Vff will be 2.0. Imo will then be 365 microamps maximum. Imo may not be greater than twice lac so this represents the maximum current available at this input voltage and the peak input current to the power factor corrector will be limited accordingly.

The lset current places another limitation on the multiplier output current. Imo may not be larger than 3.75 / Rset. For the example circuit this gives Rset = 10.27K maximum so a value of 10K is chosen.

The current out of the multiplier, Imo, must be summed with a current proportional to the inductor current to close the voltage feedback loop. Rmo. a resistor from the output of the multiplier to the current sense resistor, performs the function and the multiplier output pin becomes the summing junction. The average voltage on pin 5 will be zero under normal operation but there will be switching frequency ripple voltage which is amplitude modulated at twice the line frequency. The peak current in the boost inductor is to be limited to 5.6 amps in the example circuit and the current sense resistor is 0.25 ohms so the peak voltage across the sense resistor is 1.4 volts. The maximum multiplier output current is 365 microamps so the summing resistor, Rmo, must be 3.84K and a 3.9K resistor is chosen.

Oscillator Frequency

The oscillator charging current is lset and is determined by the value of Rset and the oscillator frequency is set by the timing capacitor and the charging current. The timing capacitor is determined from:

$$Ct = \frac{1.25}{Rset \times fs}$$

Where Ct is the value of the timing capacitor and fs is the switching frequency in Hertz. For the example converter fs is 100KHz and Rset is 10K so Ct is 0.00125μ F.

Current Error Amplifier Compensation

The current loop must be compensated for stable operation. The boost converter control to input current transfer function has a single pole response at high frequencies which is due to the impedance of the boost inductor and the sense resistor (Rs) forming a low pass filter. The equation for the control to input current transfer function is:

Vrs	Vout × Rs	;
Vcea	Vs×sL	

Where Vrs is the voltage across the input current sense resistor and Vcea is the output of the current error amplifier. Vout is the DC output voltage, Vs is the peak-to-peak amplitude of the oscillator ramp, sL is the impedance of the boost inductor (also jwL), and Rs is the sense resistor (with a current transformer it will be Rs/N). This equation is only valid for the region of interest between the resonant frequency of the filter (LCo) and the switching frequency. Below resonance the output capacitor dominates and the equation is different.

The compensation of the current error amplifier provides flat gain near the switching frequency and uses the natural roll off of the boost power stage to give the correct compensation for the total loop. A zero at low frequency in the amplifier response gives the high gain which makes average current mode control work. The gain of the error amplifier near the switching frequency is determined by matching the down slope of the inductor current when the switch is off with the slope of the ramp generated by the oscillator. These two signals are the inputs of the PWM comparator in the UC3854.

The downslope of the inductor current has the units of amps per second and has a maximum value when the input voltage is zero. In other words, when the voltage differential between the input and output of the boost converter is greatest. At this point (Vin=0) the inductor current is given by the ratio of the converter output voltage and the inductance (Vo/L). This current flows through the current sense resistor Rs and produces a voltage

with the slope VoRs/L (with current sense transformers it will be VoRs/NL). This slope, multiplied by the gain of the current error amplifier at the switching frequency, must be equal to the slope of the oscillator ramp (also in volts per second) for proper compensation of the current loop. If the gain is too high the slope of the inductor current will be greater than the ramp and the loop can go unstable. The instability will occur near the cusp of the input waveform and will disappear as the input voltage increases.

The loop crossover frequency can be found from the above equation if the gain of the current error amplifier is multiplied with it and it is set equal to one. Then rearrange the equation and solve for the crossover frequency. The equation becomes:

 $fci = \frac{Vout \times Rs \times Rcz}{Vs \times 2\pi L \times Rci}$

Where fci is the current loop crossover frequency and Rcz/Rci is the gain of the current error amplifier. This procedure will give the best possible response for the current loop.

In the example converter the output voltage is 400Vdc and the inductor is 1.0mH so the down slope of inductor current is 400mA per microsecond. The current sense resistor is 0.25 ohms so the input to the current error amplifier is 100mV per microsecond. The oscillator ramp of the UC3854 has a peak to peak value of 5.2V and the switching frequency is 100KHz so the ramp has a slope of 0.52 volts per microsecond. The current error amplifier must have a gain of 5.2 at the switching frequency to make the slopes equal. With an input resistor (Rci) value of 3.9K the feedback resistance (Rcz) is 20K to give the amplifier a gain of 5.2. The current loop crossover frequency is 15.9KHz.

The placement of the zero in the current error amplifier response must be at or below the crossover frequency. If it is at the crossover frequency the phase margin will be 45 degrees. If the zero is lower in frequency the phase margin is very stable, has low overshoot and has good tolerance for component variations. The zero must be placed at the crossover frequency so the impedance of the value of Rcz. The equation is: $Ccz = 1 / (2\pi x \text{ fci x Rcz})$. The example converter has Rcz=20K and fci=I5.9KHz so Ccz=500pF. A value of 620pF was chosen to give a little more phase margin.

A pole is normally added to the current error amplifier response near the switching frequency to reduce noise sensitivity. If the pole is above half the switching frequency the pole will not affect the frequency response of the control loop. The example converter uses a 62pF capacitor for Ccp which gives a pole at 128KHz. This is actually above the switching frequency so a larger value of capacitor could have been used but 62pF is adequate in this case.

Voltage Error Amplifier Compensation

The voltage control loop must be compensated for stability but because the bandwidth of the voltage loop is so small compared to the switching frequency the requirements for the voltage control loop are really driven by the need to keep the input distortion to a minimum rather than by stability. The loop bandwidth must be low enough to attenuate the second harmonic of the line frequency on the output capacitor to keep the modulation of the input current small. The voltage error amplifier must also have enough phase shift so that what modulation remains will be in phase with the input line to keep the power factor high.

The basic low frequency model of the output stage is a current source driving a capacitor. The power stage and the current feedback loop compose the current source and the capacitor is the output capacitor. This forms an integrator and it has a gain characteristic which rolls off at a constant 20dB per decade rate with increasing frequency. If the voltage feedback loop is closed around this it will be stable with constant gain in the voltage error amplifier. This is the technique which is used to stabilize the voltage loop. However, its performance at reducing distortion due to the second harmonic output ripple is miserable. A pole in the amplifier response is needed to reduce the amplitude of the ripple voltage and to shift the phase by 90 degrees. The distortion criteria is used to define the gain of the voltage error amplifier at the second harmonic of the line frequency and then the unity gain crossover frequency is found and is used to determine the pole location in the voltage error amplifier frequency response.

The first step in designing the voltage error amplifier compensation is to determine the amount of ripple voltage present on the output capacitor. The peak value of the second harmonic voltage is given by:

$$Vopk = \frac{Pin}{2\pi fr \times Co \times Vo}$$

Where Vopk is the peak value of the output ripple voltage (the peak to peak value will be twice this), fr is the ripple frequency which is the second harmonic of the input line frequency, Co is the value of the output capacitance and Vo is the DC output voltage. The example converter has a peak ripple voltage of 1.84Vpk.

The amount of distortion which the ripple contributes to the input must be decided next. This deci-

sion is based on the specification for the converter. The example converter is specified for 3% THD so 0.75% THD is allocated to this component. This means that the ripple voltage at the output of the voltage error amplifier is limited to 1.5%. The voltage error amplifier has an effective output range (Δ Vvea) of 1.0 to 5.0 volts so the peak ripple voltage at the output of the voltage error amplifier is give by Vvea(pk) = %Ripple x Δ Vvea. The example converter has a peak ripple voltage at the output of the voltage error amplifier of 60mVpk.

The gain of the voltage error amplifier, Gva, at the second harmonic ripple frequency is the ratio of the two values given above. The peak ripple voltage allowed on the output of the voltage error amplifier is divided by the peak ripple voltage on the output capacitor. For the example converter Gva is 0.0326.

The criteria for the choice of Rvi, the next step in the design process, are reasonably vague. The value must be low enough so that the opamp bias currents will not have a large effect on the output and it must be high enough so that the power dissipation is small. In the example converter a 511 K resistor was chosen for Rvi and it will have power dissipation of about 300mW.

Cvf, the feedback capacitor sets the gain at the second harmonic ripple frequency and is chosen to give the voltage error amplifier the correct gain at the second harmonic of the line frequency. The equation is simply:

$Cvf = \frac{1}{2\pi fr \times Rvi \times Gva}$

1

The example converter has a Cvf value of 0.08μ F. If this value is rounded down to Cvf= 0.047μ F the phase margin will be a little better with only a little more distortion so this value was chosen.

The output voltage is set by the voltage divider Rvi and Rvd. The value of Rvi is already determined so Rvd is found from the desired output voltage and the reference voltage which is 7.50Vdc. In the example Rvd=I0K will give an output voltage of 390Vdc. This could be trimmed up to 400VDC with a 414K resistor in parallel with Rvd but for this application 390Vdc is acceptable. Rvd has no effect on the AC performance of the active power factor corrector. Its only effect is to set the DC output voltage.

The frequency of the pole in the voltage error amplifier can be found from setting the gain of the loop equation equal to one and solving for the frequency. The voltage loop gain is the product of the error amplifier gain and the boost stage gain, which can be expressed in terms of the input power. The multiplier, divider and squarer terms can all be lumped into the power stage gain and their effect is to transform the output of the voltage error amplifier into a power control signal as was noted earlier. This allows us to express the transfer function of the boost stage simply in terms of power. The equation is:

$$Gbst = \frac{Pin \times Xco}{\Delta Vvea \times Vo}$$

Where Gbst is the gain of the boost stage including the multiplier, divider and squarer, Pin is the average input power, Xco is the impedance of the output capacitor, $\Delta V vea$ is the range of the voltage error amplifier output voltage (4 volts on the UC3854) and Vo is the DC output voltage.

The gain of the error amplifier above the pole in its frequency response is given by:

Where Gva is the gain of the voltage error amplifier, Xcf is the impedance of the feedback capacitance and Rvi is the input resistance.

The gain of the total voltage loop is the product of Gbst and Gva and is given by the this equation:

$$Gv = \frac{Pin \times Xco \times Xcf}{\Delta Vvea \times Vo \times Rvi}$$

Note that there are two terms which are dependent on f, Xco and Xcf. This function has a second order slope (-40dB per decade) so it must be a function of frequency squared. To solve for the unity gain frequency set Gv equal to one and rearrange the equation to solve for fvi. Xco is replaced with $1/(2\pi fCo)$ and Xcf is replaced with $1/(2\pi fCvf)$.

The equation becomes:

fvi ² =
$$\frac{\text{Pin}}{\Delta \text{Vvea} \times \text{Vo} \times \text{Rvi} \times \text{Co} \times \text{Cvf} \times (2\pi)^2}$$

Solving for fvi in the example converter gives fvi=19.14Hz. The value of Rvf can now be found by setting it equal to the impedance of Cvf at fvi. The equation is: $Rvf=1/(2\pi fviCvf)$.

In the example converter a value of 177K is calculated and 174K is used.

Feedforward Voltage Divider Filter Capacitors

The percentage of second harmonic ripple voltage on the feedforward input to the multiplier results in the same percentage of third harmonic ripple current on the AC line. The capacitors in the feedforward voltage divider (Cff1 and Cff2) attenuate the ripple voltage from the rectified input voltage. The

second harmonic ripple is 66.2% of the input AC line voltage. The amount of attenuation required, or the "gain" of the filter, is simply the amount of third harmonic distortion allocated to this distortion source divided by 66.2% which is the input to the divider. The example circuit has an allocation of 1.5% total harmonic distortion from this input so the required attenuation is Gff = 1.5 / 66.2 = 0.0227.

The recommended divider string impliments a second order filter because this gives a much faster response to changes in the RMS line voltage. Typically, it is about six times faster. The two poles of the filter are placed at the same frequency for the widest bandwidth. The total gain of the filter is the product of the gain of the two filter section so the gain of each section is the square root of the total gain. The two sections of the filter do not interact much because the impedances are different so they can be treated separately. In the example converter the gain of each filter section at the second harmonic frequency is 0.0227 or 0.15 for each section. This same relationship holds for the cutoff frequency which is needed to find the capacitor values. These are simple real poles so the cutoff frequency is the section gain times the ripple frequency or:

 $fc = \sqrt{Gff} \times fr$

The example converter has a filter gain of 0.0227 and a section gain of 0.15 and a ripple frequency of 120Hz so the cutoff frequency is fc=0.15x120=18Hz.

The cutoff frequency is used to calculate the values for the filter capacitors since, in this applivation, the impedance of the capacitor will equal the impedance of the load resistance at the cutoff frequency. The two equations given below are used to calculate the two capacitor values.

$$Cff1 = \frac{1}{2\pi \times fp \times Rff2}$$
$$Cff2 = \frac{1}{2\pi \times fp \times Rff3}$$

In the example converter Rff2 is 91K and Rff3 is 20K; so,

Cff1=1/2xx18x91K=0.1µF;

Cff2=1/2\pi x18x20K=0.44\pi F;

so choose Cff2=0.47µF.

This completes the design of the major circuits of an active power factor corrector.

DESIGN PROCEDURE SUMMARY

This section contains a brief, step-by-step summary of the design procedure for an active power factor corrector. The example circuit used above is repeated here.

1. Specifications: Determine the operating requirements for the active power factor corrector.

Example:

Pout (max): 250W Vin range: 80-270Vac Line frequency range: 47-65Hz Output voltage: 400Vdc

2. Select switching frequency:

Example:

100KHz

3. Inductor selection:

A. Maximum peak line current. Pin = Pout(max)

$$lpk = \frac{\sqrt{2 \times Pin}}{Vin (min)}$$

lpk=1.41 x250/80=4.42 amps

B. Ripple current.

Example:

∆I=0.2x4.42= = 0.2x4.42 = 0.9 amps peak to peak

C. Determine the duty factor at lpk where Vin(peak) is the peak of the rectified line voltage at low line.

$$\mathsf{D} = \frac{\mathsf{Vo} - \mathsf{Vin}\,(\,\mathsf{peak}\,)}{\mathsf{Vo}}$$

Example:

D=(400-113)/400=0.71

D. Calculate the inductance. fs is the switching frequency.

$$\mathbf{L} = \frac{\operatorname{Vin} \times \mathrm{D}}{\operatorname{fs} \times \Delta \mathrm{I}}$$

Example:

L=(113x.71)/(100,000x0.9)=0.89mH

Round up to 1.0mH.

4. Select output capacitor. With hold-up time, use the equation below. Typical values for Co are $1 \,\mu$ F to 2μ F per watt. If hold-up is not required use the second harmonic ripple voltage and total capacitor power dissipation to determine minimum size of the capacitor. Δ t is the hold-up time in seconds and V1 is the minimum output
capacitor voltage.

$$Co = \frac{2 \times Pout \times \Delta t}{Vo^2 - V1^2}$$

Example: Co=(2x250x34msec)/(400-350)=450µF

 Select current sensing resistor. If current transformers are used then include the turns ratio and decide whether the output will be positive or negative relative to circuit common. Keep the peak voltage across the resistor low. 1.0V is a typical value for Vrs.

A. Find tlpk (max) = lpk +
$$\frac{\Delta l}{2}$$

Example:

lpk(max)=4.42+0.45 = 5.0amps peak

B. Calculate sense resistor value.

$$Rs = \frac{Vrs}{Ipk(max)}$$

Example:

Rs=1.0/5.0=0.20 ohms. Choose 0.25ohms

C. Calculate the actual peak sense voltage.

Vrs(pk)=lpk(max)xRs

Example:

Vrs(pk)=5.0x0.25=1.25V

 Set independent peak current limit. Rpk1 and Rpk2 are the resistors in the voltage divider. Choose a peak current overload value, lpk(ovld). A typical value for Rpk1 is 10K.

Vrs (ovld) = lpk (olvd) x Rs

Example:

Vrs(ovid)=5.6x0.25=1.4V

Example:

Rpk2=(1.4x10K)/7.5=1.87K. Choose 1.8K

 Multiplier setup. The operation of the multiplier is given by the following equation. Imo is the multiplier output current, Km=1, lac is the multiplier input current, Vff is the feedforward voltage and Vvea is the output of the voltage error amplifier.

Imo=Kmxlacx(Vvea-1)

∨ff ²

A. Feedforward voltage divider. Change Vin from RMS voltage to average voltage of the rectified input voltage. At Vin(min) the voltage at Vff should be 1.414 volts and the voltage at Vffc, the other divider node, should be about 7.5 volts. The average value of Vin is given by the following equation where Vin(min) is the RMS value of the AC input voltage:

Vin (av) = Vin (min) x 0.9

The following two equations are used to find the values for the Vff divider string. A value of 1 Megohm is usually chosen for the divider input impedance. The two equations must be solved together to get the resistor values.

$$vff = 1.414v = \frac{Vin(av) \times Rff3}{Rff1 + Rff2 + Rff3}$$

$$Vnode = 7.5V = \frac{Vin(av)x(Rff2+Rff3)}{Rff1 + Rff2 + Rff3}$$

Example:

Rff1=910K, Rff2=91K, and Rff3=20K

B. Rvac selection. Find the maximum peak line voltage.

 $Vpk(max) = \sqrt{2} \times Vin(max)$

Example:

Vpk(max)=1.414x270=382Vpk

Divide by 600 microamps, the maximum multiplier input current.

Rvac = <u>Vpk(max)</u> 600E-6

Example:

Rvac=(382)/6E-4=637K. Choose 620K

C. Rb1 selection. This is the bias resistor. Treat this as a voltage divider with Vref and Rvac and then solve for Rb1. The equation becomes:

Rb1 = 0.25 Rvac

Example:

Rb1=0.25Rvac=155K. Choose 150K

D. Rset selection. Imo cannot be greater than twice the current through Rset. Find the multiplier input current, lac, with Vin(min). Then calculate the value for Rset based on the value of lac just calculated.

lac (min) = $\frac{Vin(pk)}{Rvac}$

Example:

lac(

Rset =
$$\frac{3.75}{2 \times \text{lac} (\text{min})}$$

Example:

Rset=3.75V/(2x182µA)=10.3Kohms.

Choose 10 Kohms

E. Rmo selection. The voltage across Rmo must be equal to the voltage across Rs at the peak current limit at low line input voltage.

 $Rmo = \frac{Vrs(pk) \times 1.12}{2 \times lac(min)}$

Example:

Rmo=(I.25x1.12)/(2x182E-6)=3.84K.

Choose 3.9Kohms

Oscillator frequency. Calculate Ct to give the desired switching frequency.

Ct=<u>1.25</u> Rset x fs

Example:

Ct=1.25/(10K x 100K)=1.25nF.

- 9. Current error amplifier compensation.
 - A. Amplifier gain at the switching frequency. Calculate the voltage across the sense resistor due to the inductor current downslope and then divide by the switching frequency. With current transformers substitute (Rs/N) for Rs. The equation is:

 $\Delta Vrs = \frac{Vo \times Rs}{L \times fs}$

Example:

∆Vrs=(400x0.25)/(0.001 =(400x0.25)/(0.001 x100,000)=1.0Vpk

This voltage must equal the peak to peak amplitude of Vs, the voltage on the timing capacitor (5.2 volts). The gain of the error amplifier is therefore given by:

 $Gca = \frac{Vs}{\Delta Vrs}$

Example:

Gca=5.2/1.0=5.2

B. Feedback resistors. Set Rci equal to Rmo.

Rci = Rmo

Rcz = Gca x Rci

Example:

Rcz=5.2x3.9K=20Kohms

C. Current loop crossover frequency.

$$fci = \frac{Vout \times Rs \times Rcz}{Vout \times Rs \times Rcz}$$

 $V_{\rm S} \times 2\pi L \times Rci$

Example:

fci=(400x0.25x20K)/(5.2x2nx0.001

=15.7KHz

D. Ccz selection. Choose a 45 degree phase margin. Set the zero at the loop crossover frequency.

$$cz = \frac{1}{2\pi \times fci \times Bcz}$$

Example:

C

Ccz=1/(2xx15.7Kx20K)=507pF.

Choose 620pF

E. Ccp selection. The pole must be above fs/2.

$$Ccp = \frac{1}{2\pi \times fs \times Rcz}$$

Example:

Ccp=1/(2nx1 100Kx20K)=80pf.

Choose 62pF

10. Harmonic distortion budget. Decide on a maximum THD level. Allocate THD sources as necessary. The predominant AC line harmonic is third. Output voltage ripple contributes 1/2% third harmonic to the input current for each 1% ripple at the second harmonic on the output of the error amplifier. The feedforward voltage, Vff, contributes 1% third harmonic to the input current for each 1% second harmonic at the Vff input to the UC3854.

Example:

3% third harmonic AC input current is chosen as the specification. 1.5% is allocated to the Vff input and 0.75% is allocated to the output ripple voltage or 1.5% to Vvao. The remaining 0.75% is allocated to miscellaneous nonlinearities.

11. Voltage error amplifier compensation.

A. Output ripple voltage. The output ripple is given by the following equation where fr is the second harmonic ripple frequency:

Vo (pk) =
$$\frac{Pin}{2\pi fr \times Co \times Vo}$$

Example:

Vo(pk)=250/(2π120x450E-6x400)=1.84Vac

B. Amplifier output ripple voltage and gain. Vo(pk) must be reduced to the ripple voltage allowed at the output of the voltage error amplifier. This sets the gain of the voltage error amplifier at the second harmonic frequency. The equation is:

$$Gva = \frac{\Delta Vvao \times \%Ripple}{Vo(pk)}$$

For the UC3854 Vvao is 5-1=4V

Example:

Gva=(4x0.015)/1.84=0.0326

C. Feedback network values. Find the component values to set the gain of the voltage error amplifier. The value of Rvi is reasonably arbitrary.

Example:

Choose Rvi=511K

 $Cvf = \frac{1}{2\pi \times fr \times Rvi \times Gva}$

Example: Cvf=1/(2nx120x511Kx0.0326)=0.08µF.

Choose 0.047µF

D. Set DC output voltage.

$$Rvd = \frac{Rvi \times Vref}{Vo - Vref}$$

Example:

Rvd=(511Kx7.5)/(400-7.5)=9.76K.

Choose 10.0K

E. Find pole frequency. fvi = unity gain frequency of voltage loop.

$$fvi^{2} = \frac{Pln}{\Delta V vao \times Vo \times Rvi \times Co \times Cvf \times (2\pi)^{2}}$$

Example:

 $fvi = \sqrt{(250/(4x400x511Kx450E-6x47E-9x39.5))} =$

19.1 Hz

REFERENCES

L. H. Dixon, "High Power Factor Preregulator for Off -Line Supplies," Unitrode Power Supply Design Seminar Manual SEM600, 1988 (Reprinted in subsequent editions of the Manual.)

L. H. Dixon, "High Power Factor Switching Preregulator Design Optimization," Unitrode Power Supply Design Seminar Manual SEM700, 1990 (Reprinted in subsequent editions of the Manual.) F. Find Rvf.

$$Rvf = \frac{1}{2\pi \times fvi \times Cvf}$$

Example:

Rvf=1/(2nx19.1 x47E-9)=177K. Choose 174K

12. Feedforward voltage divider capacitors. These capacitors determine the contribution of Vff to the third harmonic distortion on the AC input current. Determine the amount of attenuation needed. The second harmonic content of the rectified line voltage is 66.2%. %THD is the allowed percentage of harmonic distortion budgeted to this input from step 10 above.

<u>%THD</u> Gff = 66.2%

011 - 00.2

Example:

Gff=1.5/66.2=0.0227

Use two equal cascaded poles. Find the pole frequencies. fr is the second harmonic ripple frequency.

 $fp = \sqrt{Gff} \times fr$

Example:

fp=0.15xl20=18Hz Select Cff1 and Cff2.

$$Cff1 = \frac{1}{2\pi \times fp \times Rff2}$$

$$Cff2 = \frac{1}{2\pi \times \text{fp} \times \text{Rff3}}$$

Example:

Cff1=1/(2xx18x91K)=0.097µF. Choose 0.10µF

Cff2=1/(2xx18x20K)=0.44µF. Choose 0.47µF

L. H. Dixon, "Average Current Mode Control of Switching Power Supplies," Unitrode Power Supply Design Seminar Manual SEM700,1990 (Reprinted in subsequent editions of the Manual.)

S. Freeland, "Input-Current Shaping for Single-Phase Ac-Dc Power Converters," Ph.D. Thesis, California Institute of Technology, 1988

7 CONTINENTAL BLVD I MERRIMACK, NH 03054 TEL. (603) 424-2410 I FAX (603) 424-3460

APPENDIX B: Datasheets Of The Components





UC1854B UC2854A, UC2854B UC3854A, UC3854B

SLUS329A - JUNE 1998 - REVISED AUGUST 2003

ADVANCED HIGH-POWER FACTOR PREREGULATOR

FEATURES

- Controls Boost PWM to Near-Unity Power Factor
- Limits Line Current Distortion To < 3% .
- World-Wide Operation Without Switches
- Accurate Power Limiting
- Fixed-Frequency Average Current-Mode • Control
- High Bandwidth (5 MHz), Low-Offset Current • Amplifier
- Integrated Current- and Voltage Amplifier Output Clamps
- Multiplier Improvements: Linearity, 500 mV V_{AC} Offset (Eliminates External Resistor), 0 V to 5 V Multout Common-Mode Range
- V_{REF} GOOD Comparator
- Faster and Improved Accuracy ENABLE Comparator
- UVLO Options (16 V/10 V or 10.5 V/10 V)
- 300-µA Start-Up Supply Current

BLOCK DIAGRAM

DESCRIPTION

The UC3854A/B products are pin compatible enhanced versions of the UC3854. Like the UC3854, these products provide all of the functions necessary for active power factor corrected preregulators. The controller achieves near unity power factor by shaping the AC input line current waveform to correspond to the AC input line voltage. To do this the UC3854A/B uses average current mode control. Average current mode control maintains stable, low distortion sinusoidal line current without the need for slope compensation, unlike peak current mode control.

A 1% 7.5 V reference, fixed frequency oscillator, PWM, voltage amplifier with soft-start, line voltage feedforward (V_{RMS} squarer), input supply voltage clamp, and over current comparator round out the list of features.

Available in the 16-pin N (PDIP), DW (SOIC-Wide), and J (CDIP) and 20-pin Q (PLCC) package. See ordering information on page 3 for availability by temperature range.



ation is current as of publication ations per the terms of Texas Inst

TEXAS INSTRUMENTS www.ti.com

Copyright @ 2003, Texas Instruments Incorporated

UC1854B UC2854A, UC2854B UC3854A, UC3854B ISED AUGUST 2003 - JUN

DESCRIPTION (continued)

The UC3854A/B products improve upon the UC3854 by offering a wide bandwidth, low offset current amplifier, a faster responding and improved accuracy enable comparator, a VREF GOOD comparator, UVLO threshold options (16 V/10 V for offline, 10.5 V/10 V for startup from an auxiliary 12 V regulator), lower startup supply current, and an enhanced multiply/divide circuit. New features like the amplifier output clamps, improved amplifier current sinking capability, and low offset VAC pin reduce the external component count while improving performance. Improved common mode input range of the multiplier output/current amplifier input allow the designer greater flexibility in choosing a method for current sensing. Unlike its predecessor, RSET controls only oscillator charging current and has no effect on clamping the maximum multiplier output current. This current is now clamped to a maximum of 2 × I_{AC} at all times which simplifies the design process and provides foldback power limiting during brownout and extreme low line conditions.



These devices have limited built-in ESD protection. The leads should be shorted together or the device placed in conductive foam during storage or handling to prevent electrostatic damage to the MOS gates.

ABSOLUTE MAXIMUM RATINGS

over operating free-air temperature range unless otherwise noted⁽¹⁾

		UCX854A, UCX854B	UNIT
Supply voltage, V _{CC}		22	V
GTDRV current, IGTDRV	Continuous	0.5	A
GTDRV Current, IGTDRV	50% duty cycle	1.5	A
land on the set	VSENSE, VRMS, ISENSE MOUT	11	V
Input voltage	PKLMT	5	V
Input current	RSET, IAC, PKLMT, ENA	10	mA
Power dissipation		1	W
Junction temperature, TJ		-55 to 150	
Storage temperature, Tstg		-65 to 150	°C
Lead temperature, T _{SOI} , 1,6 mm (1/16 inch)	from case for 10 seconds	300]

(1) Stresses beyond those listed under "absolute maximum ratings" may cause permanent damage to the device. These are stress ratings only, and functional operation of the device at these or any other conditions beyond those indicated under "recommended operating conditions" is not implied. Exposure to absolute-maximum-rated conditions for extended periods may affect device reliability. All voltages are with respect to GND. Currents are positive into and negative out of, the specified terminal. ENA input is internally clamped to approximately 10 V.

RECOMMENDED OPERATING CONDITIONS

		MIN	MAX	UNIT
Supply voltage, V _{CC}		10	20	V
	UC1854X	-55	125	
Operating junction temperature, TJ	UC2854X	-40	85	°C
	UC3854X	0	70	1

3



ORDERING INFORMATION

	UVLO	UVLO	PART NUMBERS					
TA	TURN-ON (V)	TURN-OFF (V)	CDIP-16 (J)	PDIP-16 (N)	SOIC-16 (DW)	PLCC-20 (Q)		
FF00 to 40500	16	10	-	-	-	-		
-55°C to 125°C	10.5	10	UC1854BJ	-	-	-		
10:0 1- 85:0	16	10	UC2854AJ	UC2854AN	UC2854ADW	UC2854AQ		
-40°C to 85°C	10.5	10	UC2854BJ	UC2854BN	UC2854BDW	UC2854BQ		
000 4- 7000	16	10	-	UC3854AN	UC3854ADW	-		
0°C to 70°C	10.5	10	-	UC3854BN	UC3854BDW	-		

(1) The DW and Q packages are available taped and reeled. Add TR suffix to device type (e.g. UC2854ADWTR) to order quantities of 2,000 devices per reel for the DW package and 1,000 devices per reel for the Q package.

THERMAL RESISTANCE

	PACKAGED DEVICES			
RESISTANCES	CDIP-16	PDIP-16	SOP-16 (DW)	PLCC-20
	(0)	(14)	(011)	(%)
9JC (°C/W)	28(2)	45	27	34
ө јд (°C/W)	80-120	90(3)	50-130(3)	43-75(3)
(0)				

 (2) egC data values stated are derived from MIL-STD-1835B which states "the baseline values shown are worst case (mean +2s) for a 60 × 60 mil microcircuit device silicon die and applicable for devices with die sizes up to 14,400 square mils. For device die sizes greater than 14,400 square mils use the following values, dual-in-line, 11°C/W, flat pack and pin grid array, 10°C/W.
 (3) egA (junction-to-ambient) applies to devices mounted to five square inch FR4 PC board with one ounce copper where noted. When resitance range is given, lower values are for five square inch aluminum PC board. Test PWB is 0.062 inches thick and typically uses 0.635 nm trace widths for power packages and 1.3 mm trace widths for non-power packages with a 100 × 100 mil probe land are at the end of each trace. end of each trace.

TEXAS INSTRUMENTS www.ti.com

199

UC1854B UC2854A, UC2854B UC3854A, UC3854B SLUS329A - JUNE 1998 - REVISED AUGUST 2003

ELECTRICAL CHARACTERISTICS $V_{CC} = 18 V$, $R_T = 8.2 k_{2}$, $C_T = 1.5 nF$, $V_{PKL,MT} = 1 V$, $V_{VRMS} = 1.5 V$, $I_{IAC} = 100 \mu$ A, $I_{ISENSE} = 0 V$, $V_{CAO} = 3.5 V$, $V_{VAO} = 5 V$, $V_{VSENSE} = 3 V$, $-40^{\circ}C < T_A < 85^{\circ}C$ for the UC2854A and UC2854B, and $0^{\circ}C < T_A < 70^{\circ}C$ for the UC3854A and UC2854B, and $T_A = T_J$ (unless otherwise noted)

	PARAMETER	TEST CONDITIONS	MIN	TYP	MAX	UNIT
OVERA	LL	÷	-			
	Supply current, off	CAO = 0 V, VAO = 0 V, VCC = VUVLO-0.3 V		250	400	μA
	Supply current, on			12	18	mA
		UCx854A	15.0	16.0	17.5	
	VCC turn-on threshold voltage	UCx854B	8.0	10.5	11.2	v
	V _{CC} turn-off threshold voltage		9	10	12	v
	VCC clamp	IVCC = IVCC(on) + 5 mA	18	20	22	
VOLTA	GE AMPLIFIER					
	Input voltage		2.9	3.0	3.1	V
	VSENSE bias current		-500	-25	500	nA
	Open loop gain	$2 V \le V_{OUT} \le 5 V$	70	100		dB
VOH	High-level output voltage	$I_{LOAD} = -500 \mu A$		6		V
VOL	Low-level output voltage	I _{LOAD} = 500 μA		0.3	0.5	V
Isc	Output short-circuit current	V _{OUT} = 0 V		1.5	4.5	mA
	Gain bandwidth product ⁽¹⁾	f _{IN} = 100 kHz, 10 mVp_p		1		MHz
CURRE	NT AMPLIFIER	÷				
	long to final college	$V_{CM} = 0 V$, $T_A = 25^{\circ}C$	-4		0	
	input onset voitage	V _{CM} = 0 V, overtemperature	-5.5		0	mv
	Input bias current, ISENSE	∨ _{CM} = 0 ∨	-500		500	nA
	Open loop gain	$2 V \le V_{OUT} \le 6 V$	80	110		dB
VOH	High-level output voltage	$I_{LOAD} = -500 \mu A$		8		V
VOL	Low-level output voltage	I _{LOAD} = 500 μA		0.3	0.5	v
Isc	Output short-circuit current	V _{OUT} = 0 V		1.5	4.5	mA
CMRR	Common mode rejection range		-0.3		5.0	V
	Gain bandwidth product ⁽¹⁾	f _{IN} = 100 kHz, 10 mVp-p	3	5		MHz
REFER	ENCE					
	Output wellings	IREF = 0 mA, T _A = 25°C	7.4	7.5	7.6	V
	Output voitage	IREF = 0 mA	7.35	7.50	7.65	v
	Load regulation	$1 \text{ mA} \leq I_{REF} \leq 10 \text{ mA}$	0	8	20	2014
	Line regulation	$12 \text{ V} \le \text{V}_{CC} \le 18 \text{ V}$	0	14	25	mv
Isc	Short circuit current	VREF = 0 V	25	35	60	mA

(1) Ensured by design. Not production tested.

4

(2) Gain constant. (K) = $\frac{I_{AC} \times \left(V_{VAO} - 1.5 V\right)}{\left[\left(V_{VRMS}\right)^2 \times I_{MOUT}\right]}$

ELECTRICAL CHARACTERISTICS V_{CC} = 18 V, R_T = 8.2 kΩ, C_T = 1.5 nF, V_{PKLMT} = 1 V, V_{VRMS} = 1.5 V, I_{JAC} = 100 µA, I_{JSENSE} = 0 V, V_{CAO} = 3.5 V, V_{VAO} = 5 V, V_{VSENSE} = 3 V, -40°C < T_A < 85°C for the UC2854A and UC2854B, and 0°C < T_A <70°C for the UC3854A and UC3854B, and T_A = T_J (unless otherwise noted)

PARAMETER	TEST CONDITIONS	MIN	TYP	MAX	UNIT		
OSCILLATOR	-						
Initial accuracy	T _A = 25°C	85	100	115	kHz		
Voltage stability	Voltage stability 12 V ≤ V _{CC} ≤ 18 V						
Total variation	Line, temperature	80		120	kHz		
Ramp amplitude (peak-to-peak)		4.9		5.9			
Ramp valley voltage		0.8		1.3	V V		
ENABLE/SOFT-START/CURRENT LIMIT	· · · · ·						
Enable threshold voltage		2.35	2.55	2.80	V		
Enable hysteresis	VFAULT = 2.5 V		500	600	mV		
Enable input bias current	V _{ENA} = 0 V		-2	-5	μA		
Propagation delay to disable time	 Enable overdrive = 100 mV 		300		ns		
Soft-start charge current	V _{SS} = 2.5 V	10	14	24			
Peak limit offset voltage		-15		15	mV		
Peak limit input current	VPKLMT = -0.1 V	-200	-100		μA		
Peak limit propagation delay time	(1)		150		ns		
MULTIPLIER	•						
Output current, I _{AC} limited	I _{AC} = 100 μA, V _{RMS} = 1 V, R _{SET} = 10 kΩ	-220	-200	-170			
Output current, zero	I _{AC} = 0 μA, R _{SET} = 10 kΩ	-2.0	-0.2	2.0	μA		
Output current, power limited	V _{RMS} = 1.5 V, Va = 6 V	-230	-200	-170			
	V _{RMS} = 1.5 V, Va = 2 V		-22		μΑ		
	V _{RMS} = 1.5 V, Va = 5 V		-156				
Output current	V _{RMS} = 5 V, Va = 2 V		-2		μA		
	V _{RMS} = 5 V, Va = 5 V		-14				
Gain constant ⁽²⁾	V _{RMS} = 1.5 V, Va = 6V, T _A = 25°C	-1.1	-1.0	-0.9	A/A		
GATE DRIVER							
VOH High-level output voltage	IOUT = -200 mA, VCC = 15 V	12.0	12.8		V		
VOL Low-level output voltage	IOUT = 200 mA		1.0	2.2	v		
	I _{OUT} = 10 mA		300	500	mV		
Low-level UVLO voltage	I _{OUT} = 50 mA, V _{CC} = 0 V		0.9	1.5	V		
Output rise time(1)	CLOAD = 1 nF		35				
Output fall time(1)	C _{LOAD} = 1 nF		35		ns		
Output peak current(1)	CLOAD = 10 nF		1.0		A		

(1) Ensured by design. Not production tested.

(2) Gain constant. (K) = $\frac{I_{IAC} \times (V_{VAO} - 1.5 V)}{\left[(V_{VRMS})^2 \times I_{MOUT} \right]}$

UC1854B UC2854A, UC2854B UC3854A, UC3854B SLUS329A – JUNE 1998 – REVISED AUGUST 2003

TERMINAL FUNCTIONS

· ·	DACK	ACE6		DE CODIDITION
NAME J/N/DW Q/L		10	DESCRIPTION	
	J/N/DW	Q/L		
CAO	3	4	0	Output of the wide bandwidth current amplifier and one of the inputs to the PWM duty-cycle comparator. The output signal generated by this amplifier commands the PWM to force the correct input current. The output can swing from 0.1 V to 7.5 V.
CT	14	18	1	Capacitor from CT to GND sets the PWM oscillator frequency
ENA	10	13	Т	A nominal voltage above 2.65 V on this pin allows the device to begin operating. Once operating, the device shuts off if this pin goes below 2.15 V nominal.
GND	1	2	-	All bypass and timing capacitors connected to GND should have leads as short and direct as possible. All voltages are measured with respect GND.
GTDRV	16	20	0	Output of the PVMI is a 1.5-A peak totem-pole MOSFET gate driver on GTDRV. This output is internally clamped to 15 V so that the device can be operated with VCC as high as 35 V. Use a series gate resistor of at least 5 Ω to prevent interaction between the gate impedance and the GTDRV output driver that might cause the GTDRV output to overshoot excessively. Some overshoot the GTDRV output output as
IAC	6	8	I	Current input to the multiplier, proportional to the instantaneous line voltage. This input to the analog multiplier is a current. The multiplier is tailored for very low distortion from this current input (IAC) to MOUT, so this is the only multiplier input that should be used for sensing instantaneous line voltage. The nominal voltage on IAC is 6 V, so in addition to a resistor from IAC to rectified 60 Hz, connect a resistor from IAC to VREF. If the resistor to VREF is one-fourth of the value of the resistor to the rectifier, then the 6-V offset is cancelled, and the line current has minimal cross-over distortion.
ISENSE	4	5	I	Switch current sensing input. This is the inverting input to the current amplifier. This input and the non-inverting input MOUT remain functional down to and below GND. Care should be taken to avoid taking these inputs below -0.5V, because they are protected with diodes to GND.
MOUT	5	7	I/O	Multiplier output and current sense plus. The output of the analog multiplier and the non-inverting input of the current amplifier are connected together at MOUT. The cautions about taking ISENSE below –0.5V also apply to MOUT. As the multiplier output is a current, this is a high-impedance input similar to ISENSE, so the current amplifier can be configured as a differential amplifier to reject GND noise.
PKLMT	2	3	I	Peak limit. The threshold for PKLMT is 0.0 V. Connect this input to the negative voltage on the current sense resistor. Use a resistor to REF to offset the negative current sense signal up to GND.
RSET	12	15	I	Oscillator charging current and multiplier limit set. A resistor from RSET to ground programs oscillator charging current and maximum multiplier output. Multiplier output current does not exceed 3.75V divided by the resistor from RSET to ground.
SS	13	17	I	Soft-start. SS remains at GND as long as the device is disabled or V _{CC} is too low. SS pulls up to over 8 V by an internal 14-mA current source when both V _{CC} becomes valid and the device is enabled. SS acts as the reference input to the voltage amplifier if SS is below VREF. With a large capacitor from SS to GND, the reference to the voltage regulating amplifier rises slowly, and increase the PWM duty cycle slowly. In the event of a disable command or a supply dropout, SS will quickly discharge to ground and disable the PWM.
VAO	7	9	- I	Voltage amplifier output
VCC	15	19	I	Positive supply rail
VREF	9	12	0	Used to set the peak limit point and as an internal reference for various device functions. This voltage must be present for the device to operate.
VRMS	8	10	I	One of the inputs into the multiplier. This pin provides the input RMS voltage to the multiplier circuitry.
VSENSE	11	14	Ι	This pin provides the feedback from the output. This input goes into the voltage error amplifier and the output of the error amplifier is another of the inputs into the multiplier circuit.

FUNCTIONAL DESCRIPTION

The UC3854A and UC3854B family of products are designed as pin compatible upgrades to the industry standard UC3854 active power factor correction circuits. The circuit enhancements allow the user to eliminate in most cases several external components currently required to successfully apply the UC3854. In addition, linearity improvements to the multiply, square and divide circuitry optimizes overall system performance. Detailed descriptions of the circuit enhancements are provided below. For in-depth design applications reference data refer to the application notes, UC3854 Controlled Power Factor Correction Circuit Design (SLUA144) and UC3854A and UC3854B Advanced Power Factor Correction Control ICs (SLUA177).

Multiply/Square and Divide

The UC3854A/B multiplier design maintains the same gain constant (K=-1) as the UC3854. The relationship between the inputs and output current is given as:

$$I_{MOUT} = I_{iAC} \times \frac{\left(V_{VAO} - 1.5 V\right)}{K \times \left(V_{VRMS}\right)^2}$$
(1)

This is nearly the same as the UC3854, but circuit differences have improved the performance and application.

The first difference is with the IAC input. The UC3854A/B regulated this pin voltage to the nominal 500 mV over the full operating temperature range, rather than the 6.0 V used on the UC3854. The low offset voltage eliminates the need for a line zero crossing compensating resistor to VREF from IAC that UC3854 designs require. The maximum current at high line into IAC should be limited to 250 μ A for best performance.

Therefore, if V_{VAC(max)} = 270 V,

$$R_{IAC} = \frac{270 \times 1.414}{250 \,\mu\text{A}} = 1.53 \,\text{M}\Omega \tag{2}$$

The V_{RMS} pin linear operating range is improved with the UC3854A/B as well. The input range for VRMS extends from 0 V to 5.5 V. Since the UC3854A squaring circuit employs an analog multiplier, rather than a linear approximation, accuracy is improved, and discontinuities are eliminated. The external divider network connected to VRMS should produce 1.5 V at low line (85 VAC). This puts 4.77 V on VRMS at high line (27 VAC) which is well within its operating range.

The voltage amplifier output forms the third input to the multiplier and is internally clamped to 6.0 V. This eliminated an external zener clamp often used in UC3854 designs. The offset voltage at this input to the multiplier has been raised on the UC3854A/B to 1.5 V.

The multiplier output pin, which is also common to the current amplifier non-inverting input, has a -0.3 V to 5.0 V output range, compared to the -0.3 V to 2.5 V range of the UC3854. This improvement allows the UC3854A/B to be used in applications where the current sense signal amplitude is very large.

Voltage Amplifier

The UC3854A/B voltage amplifier design is essentially similar to the UC3854 with two exceptions. The first is with the internal connection. The lower voltage reduces the amount of charge on the compensation capacitors, which provides improved recovery form large signal events, such as line dropouts, or power interruption. It also minimizes the dc current flowing through the feedback. The output of the voltage amplifier is also changes. In addition to a 6.0 V temperature compensated clamp, the output short circuit current has been lowered to 2 mA typical, and an active pull down has replaced the passive pull down of the UC3854.



UC1854B UC2854A, UC2854B UC3854A, UC3854B SLUS329A - JUNE 1998 - REVISED AUGUST 2003

Current Amplifier

The current amplifier for an average current PFC controller needs a low offset voltage in order to minimize AC line current distortion. With this in mind, the UC3854A/B current amplifier has improved the input offset voltage from ± 4 mV to 0 V to ± 3 mV. The negative offset of the UC3854A/B guarantees that the PWM circuit will not drive the MOSFET is the current command is zero (both current amplifier inputs zero.) Previous designs required an external offset cancellation network to implement this key feature. The bandwidth of the current amplifier has been improved as well to 5 MHz typical. While this is not generally an issue at 50 Hz or 60 Hz inputs, it is essential for 400 Hz input avionics applications

Miscellaneous

Several other important enhancements have been implemented in the UC3854A/B. A V_{CC} supply voltage clamp at 20 V allows the controller to be current fed if desired. The lower startup supply current (250 mA typical), substantially reduces the power requirements of an offline startup resistor. The 10.5 V/10 V UVLO option (UC3854B) enables the controller to be powered off of an auxiliary 12 V supply.

The VREF GOOD comparator guarantees that the MOSFET driver output remains low if the supply of the 7.5 V reference are not yet up. This improvement eliminates the need for external Schottky diodes on the PKLMT and Mult Out pins that some UC3854 designs require. The propagation delay of the disable feature has been improved to 300 ns typical. This delay was proportional to the size of the VREF capacitor on the UC3854, and is typically several orders of magnitude slower.



UC1854B UC2854A, UC2854B UC3854A, UC3854B SLUS329A - JUNE 1998 - REVISED AUGUST 2003



OSCILLATOR FREQUENCY





Metglas®

POWERLITE® High Frequency Distributed Gap Inductor Cores

www.metglas.com

recifical bulleti

POWERLITE® C-Cores are manufactured with iron based Metglas[®] amorphous Alloy 2605SA1. Their unique combination of low loss and high saturation flux density provide for size reduction and improvements in energy efficiency making them an ideal solution for automotive inductor applications



Benefits

Manufactured in a variety of ultra-efficient core configurations, POWERLITE C-Cores provide significant cost, design and performance benefits over ordinary Si-Fe, ferrite and MPP cores such as:

- High Saturation Flux Density (1.56 T)
- Low Profile enables weight and volume reductions of up to 50%
- Low Temperature Rise enabling smaller compact designs
- Low Loss resulting from micro-thin Metglas ribbon (25 μm)

Physical Properties METGLAS Alloy 2605SA1

Ribbon Thickness (µm)	.25
Density (g/cm3)	7.18
Thermal Expansion (ppm/°C)	7.6
Crystallization Temperature (°C)	505
Curie Temperature (°C)	392
Continuous Service Temperature (°C)	150
Tensile Strength (MN/m2)	1k-1.7k
Elastic Modulus (GN/m2)	100-110
Vicker's Hardness (50g load)	860

Magnetic Properties METGLAS Powerlite Cores

Saturation Flux Density (Tesla)	1.56
Permeability (depending on gap size)	VARIABLE
Saturation Magnetostriction (ppm)	27
Electrical Resistivity (μΩcm)	137

e-mail: metglas@metglas.com

1-800-581-7654

Copyright © 2004-2005, Metglas, Inc. All Rights Reserved



Technical Bulletin

Standard Core Specifications



	Core Dimensions Performance Parameters									ers					
	a(mm)	±	b(mm)	c(mm)	d(mm)	±	e(mm)	±	f(mm)	±	(lm) cm	ac (cm ²)	Wa (cm ²)	Ар (cm ⁴)	(gms)
AMCC 4	9.00	0.50	10.50	32.75	15.25	0.25	28.50	0.50	51.00	1.00	12.70	1.11	3.44	3.82	102
AMCC 6.3	10.0	0.50	11.00	33.0	20.00	0.50	31.00	1.00	53.00	2.00	13.10	1.60	3.60	5.80	150
AMCC 10	11.0	0.80	13.00	40.0	20.00	0.50	35.00	1.00	62.00	2.00	15.40	1.80	5.20	9.40	200
AMCC 8	11.0	0.80	13.00	30.0	20.00	0.50	35.00	1.00	52.00	2.00	13.20	1.80	3.90	7.00	170
AMCC 16B	11.0	0.80	13.00	50.0	25.00	0.50	35.00	1.00	72.00	2.00	16.90	2.30	6.50	15.0	280
AMCC 16A	11.0	0.80	13.00	40.0	25.00	0.50	35.00	1.00	62.00	2.00	15.10	2.30	5.20	12.0	250
AMCC 20	11.0	0.80	13.00	50.0	30.00	0.50	35.00	1.00	72.00	2.00	17.50	2.70	6.50	17.6	340
AMCC 40	13.0	0.80	15.00	56.0	35.00	0.50	41.00	1.00	82.00	2.00	19.90	3.70	8.40	31.1	530
AMCC 25	13.0	0.80	15.00	56.0	25.00	0.50	41.00	1.00	82.00	2.00	19.60	2.70	8.40	22.7	380
AMCC 32	13.0	0.80	15.00	56.0	30.00	0.50	41.00	1.00	82.00	2.00	20.00	3.20	8.40	26.9	460
AMCC 50	16.0	1.00	20.00	70.0	25.00	0.50	52.00	1.00	102.0	3.00	24.90	3.30	14.0	46.2	590
AMCC 63	16.0	1.00	20.00	70.0	30.00	0.50	52.00	1.00	102.0	3.00	25.30	3.90	14.0	54.6	710
AMCC 80	16.0	1.00	20.00	70.0	40.00	1.00	52.00	1.00	102.0	3.00	25.40	5.20	14.0	72.8	950
AMCC 100	16.0	1.00	20.00	70.0	45.00	1.00	52.00	1.00	102.0	3.00	25.00	5.90	14.0	82.6	1,060
AMCC 160	19.0	1.00	25.00	83.0	40.00	1.00	63.00	1.00	121.0	3.00	28.50	6.50	20.8	135.2	1,330
AMCC 125	19.0	1.00	25.00	83.0	35.00	1.00	63.00	1.00	121.0	3.00	30.20	5.40	20.8	112.1	1,170
AMCC 250	19.0	1.00	25.00	90.0	60.00	1.00	63.00	1.00	128.0	3.00	31.40	9.30	22.5	209.3	2,100
AMCC 200	19.0	1.00	25.00	83.0	50.00	1.00	63.00	1.00	121.0	3.00	29.80	7.80	20.8	162.2	1,670
AMCC 168S	20.4	0.50	30.20	155.2	20.00	0.50	71.00	0.75	196.0	2.00	45.40	3.35	45.7	153.0	1,090
AMCC 320	22.0	1.00	35.00	85.0	50.00	1.00	79.00	1.00	129.0	4.00	32.50	9.00	29.8	267.8	2,170
AMCC 400	22.0	1.00	35.00	85.0	65.00	1.00	79.00	1.00	129.0	4.00	33.60	11.7	29.8	348.1	2,820
AMCC 500	25.0	1.00	40.00	85.0	55.00	1.00	90.00	1.00	135.0	4.00	35.60	11.3	34.0	384.2	2,900
AMCC 630	25.0	1.00	40.00	85.0	70.00	1.00	90.00	1.00	135.0	4.00	35.60	14.3	34.0	486.2	3,670
AMCC 800A	25.0	1.00	40.00	85.0	85.00	1.50	90.00	1.00	135.0	4.00	35.60	17.4	34.0	591.6	4,450
AMCC 367S	25.8	1.00	66.00	97.8	25.00	0.70	117.6	1.50	149.4	1.50	43.78	5.29	63.8	338.0	1,662
AMCC 800B	30.0	1.00	40.00	95.0	85.00	1.50	100.0	1.00	155.0	4.00	39.30	21.0	38.0	798.0	5,930
AMCC 1000	33.0	1.00	40.00	105.0	85.00	1.50	106.0	1.00	171.0	5.00	42.70	23.0	42.0	966.0	7,060

e-mail: metglas@metglas.com

1-800-581-7654

Copyright © 2004-2005, Metglas, Inc. All Rights Reserved

Bulletin PD-20766 rev. B 02/04

Г

International **TOR** Rectifier

8ETX06 8ETX06S 8ETX06-1 8ETX06FP

Hyperfast Rectifier

Fe

.

eatures	$t_{rr} = 15$ ns typ.
Hyperfast Recovery Time	$l_{r(a)0} = 8Amp$
Low Leakage Current	
175°C Operating Junction Temperature UL E78996 approved	$V_R = 600V$

Description/Applications

State of the art Hyperfast recovery rectifiers designed with optimized performance of forward voltage drop, Hyperfast recover time, and soft recovery. The planar structure and the platinum doped life time control guarantee the best overall performance, ruggedness and reliability characteristics. These devices are intended for use in PFC Boost stage in the AC-DC section of SMPS, inverters or as freewheeling diodes. The IR extremely optimized stored charge and low recovery current minimize the switching losses and reduce over dissipation in the switching element and snubbers.

Absolute Maximum Ratings

	Parameters	Max	Units
VRRM	Peak Repetitive Reverse Voltage	600	v
I _{F(AV)}	Average Rectified Forward Current @ $T_c = 143^{\circ}C$ @ $T_c = 106^{\circ}C$ (FULLPACK)	8	A
IFSM	Non Repetitive Peak Surge Current @ T_ = 25°C	110	
IFM	Peak Repetitive Forward Current	18	
TJ, TSTG	Operating Junction and Storage Temperatures	- 65 to 175	°C



www.irf.com

Current Cransducer LA 55-P/SP1 For the electronic measurement of ourrents : DC, AC, pulsed, with a gaivanic isolation between the primary circuit (high power) with a gaivanic isolation between the primary circuit (high power) Next	_				<u>LEM</u>
Current Transducer LA 55-P/SP1 For the electronic measurement of currents : DC, AC, puised, with a gaivanic isolation between the primary circuit (high power) and the secondary circuit (electronic circuit):	_				
For the electronic measurement of currents : DC, AC, pulsed, with a gaivanic isolation between the primary obtaul (high power) and the secondary circuit (lectronic diruct).	Cu	rrent Transducer LA 55-P	/SP1		I _{PN} = 50 A
and the secondary circuit (electronic bircuit). We will secondary contain tends to be the second of the second o	For the	he electronic measurement of currents : DC, a galvanic isolation between the primary circu	AC, puised, It (high power	1	-
Vertice Vertice Vertice Vertice k_{1} Primary nominal rms. current k_{2} k_{1} k_{2}	and t	he secondary circuit (electronic circuit).		·	
Electrical data $L_{n} = Primary nominal runs, current50AR_{n} = Primary nominal runs, current50AR_{n} = Primary nominal runs, current50AR_{n} = Primary nominal runs, currentR_{nerr} R_{n$					
L_{i} Primary nominal r.m.s. currentS0AFeatures L_{i} Primary current, measuring range $0 = 100$ AClosed loco (compensated) current R_{i} Measuring resistance (0 $T_{i} + 70^{\circ}$ ($T_{i} + 85^{\circ}$ ($T_{i} + 1200$)Special features L_{i} Substry valage ($t = 5.5$ ($t = 5.5$) $t = 12.5 \times 5.5 \times$	EI	ectrical data			•
$ \begin{array}{c} \begin{array}{c} \\ R_{e} \\ R_{e} \\ \end{array} \\ \begin{array}{c} \\ \mbox{Measuring resistance (2) \\ R_{e} \\ Measuring resistance (2) \\ \mbox{Measuring $	Ļ.,	Primary nominal r.m.s. current	50	A	Features
$ \begin{array}{c} R_{k} & \text{Measuring resistance is R_{k} = N^{k} = N^{k} = R_{k} = R = $	Ļ	Primary current, measuring range	0±100	A	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	R.	Measuring resistance @ T _A =	70°C T _A = 85'	·C	 Crosed loop (compensated) current transducer using the Hall effect
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		1944 1947 - 1946 - 1946 1947 - 1946 - 1946	245 0 246		Printed circuit board mounting
according to UL 94-VC.with $\pm 15 V$ $\frac{2}{6} \pm 50 A_{max}$ 0 335 30 300 1122000 <td></td> <td>0 ± 100 A 0</td> <td>35 0 30</td> <td>- u</td> <td> insulated plastic case recognized </td>		0 ± 100 A 0	35 0 30	- u	 insulated plastic case recognized
Line Secondary nominal r.m.s. current25 mAmASpecial featuresKi Conversion ratio25 mAmA1, 2000 ± 12, 15+ I, = 0, ± 100 A + K_x = 11 2000.Vi Supply valtage (± 5%)± 12, 15 mA+ K_x = 11 2000.+ K_x = 11 2000.Vi Current consumption25 mAMVAdvantagesAccuracy - Dynamic performance data X Accuracy = Up, m, T_x = 25°C $@ \pm 15 \vee (\pm 5\%) \pm 0.65$ $@ \pm 12 - 15 \vee (\pm 5\%) \pm 0.05$ + Excellent accuracy $+ Very pool inearityXCurrent @ U_x = 0, T_x = 25°CU_x = Restolual current @ U_x = 0, T_x = 25°CU_x = Restolual current @ U_x = 0, after an overload of 3 x I_{M_x} = 0.15 mA\pm 0.05 \pm 0.25 mA- Excellent accuracy\pm 0.05 \pm 0.25 mAI_bU_bCurrent @ U_x = 0, W of I_{M_x} = 0°C + 85°CU_x = Restolual current @ U_x = 0, after an overload of 3 x I_{M_x} = 0.15 mA\pm 0.05 \pm 0.25 mA- Ourrent overload capability.I_bCurrent @ 10 % of I_M_x = 0°C + 85°CUidet accuracely folowedf> 200 A/Us\pm 0.05 \pm 0.25 mA- Ourrent overload capability.I_bClick accuracely folowedfStatic converters for DO motor drives\pm 0.05 \pm 0.05- Alus\pm 0.05 \pm 0.05- Alus\pm 0.05 \pm 0.05I_bClick accuracely folowedfT_x = 70°C145\pm 0- Current overload capability.I_bClick accuracely folowedfT_x = 70°C145\pm 0- Current overload capability.I_bClick accuracely folowedfT_x = 70°C145\pm 0- Current overload capability.I_bClick accuracely folowedfT_x = 70°C145\pm 0$		w1th ± 15 ∨ @ ± 50 A _{ma} 0 @ ± 100 A 0	335 30 330 95 30 90	0 0	according to UL 94-V0.
$ \begin{array}{c} \begin{array}{c} & & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\ $		Secondary nominal r.m.s. current	25	må	Special features
$ \begin{array}{c} V_{k}^{\circ} & \text{Supply voltage (s 5 %)} \\ V_{k}^{\circ} & \text{Current consumption} \\ V_{k}^{\circ} & \text{R.m.s. voltage for AC isolation test, 50 Hz, 1 mn} \\ \hline & 10(g_{2} \pm 15 \vee) = I_{k}^{\circ} \text{MA} \\ \hline & \text{Accuracy - Dynamic performance data} \\ \hline & \text{Accuracy - Dynamic performance data} \\ \hline & \text{Accuracy (b, L_{k}, T_{k} - 25^{\circ}\text{C}) \\ & (b \pm 12 - 15 \vee (\pm 5 \%) \pm 0.65 \\ \hline & (b \pm 21 - 15 \vee (\pm 5 \%) \pm 0.35 \\ \hline & (b \pm 21 - 15 \times (\pm 5 \%) \pm 0.35 \\ \hline & (b \pm 21 - 15 \times (\pm 5 \%$	Ř.	Conversion ratio	1:2000		• L = 0±100 A
	v, E	Supply voltage (± 5 %)	± 1215	V	 K = 1:2000.
V. R.m.s. voltage for AC isolation test, 50 Hz, 1 mn 2.5 kV AdVantages Accuracy - Dynamic performance data -	e	Current consumption	10(Q±16V)+	l, mA	
Accuracy - Dynamic performance data • Excellent accuracy X Accuracy @ $I_{w_1}, T_a = 25^{\circ}C$ @ ± 15 V (± 5 %) ± 0.55 % $@$ ± 12 - 15 V (± 5 %) ± 0.30 % • Common fill the common fill the fil	v,	R.m.s. voltage for AC isolation test, 90 Hz, 1 mn	2.5	kν	Advantages
XAccuracy (2) $I_{\mu_{n}}, T_{a} = 25^{\circ}$ C(2) $\pm 15 \vee (\pm 5\%)$ ± 0.55 % ± 0.55 % \mathcal{E}_{i} Linearity error(2) $\pm 12 - 15 \vee (\pm 5\%)$ ± 0.30 % $\oplus 0.55$ % \mathcal{E}_{i} Linearity error(3) $12 - 15 \vee (\pm 5\%)$ ± 0.30 % $\oplus 0.55$ % I_{0} Offset current (2) $I_{i} = 0, T_{a} - 25^{\circ}$ C $= 0.15 \text{ mA}$ $\oplus 115 \text{ mA}$ $\oplus 115 \text{ mA}$ $\oplus 115 \text{ mA}$ I_{0} Residual current (2) $I_{0} = 0, \text{ after an overload of 3 x I_{\mu_{n}}\pm 0.15 \text{ mA}\oplus 115 \text{ mA}\oplus 115 \text{ mA}I_{0}Residual current (2) I_{0} = 0, \text{ after an overload of 3 x I_{\mu_{n}}\pm 0.15 \text{ mA}\oplus 115 \text{ mA}\oplus 115 \text{ mA}I_{0}Residual current (2) I_{0} = 0, \text{ after an overload of 3 x I_{\mu_{n}}\pm 0.15 \text{ mA}\oplus 0.15 \text{ mA}\oplus 115 \text{ mainly to external interference}I_{0}Residual current (2) I_{0} = 0, \text{ after an overload of 3 x I_{\mu_{n}}\pm 0.05 \pm 0.25 \text{ mA}\oplus 0.05 \pm 0.25 \text{ mA}\oplus 0.05 \pm 0.25 \text{ mA}I_{0}Reschon time (2) 10 % of I_{\mu_{n}}< 1 \text{ us}= 0.25 \text{ mA}\oplus 0.05 \pm 0.25 \text{ mA}\oplus 0.05 \pm 0.25 \text{ mA}I_{0}Response time (2) 10 % of I_{\mu_{n}}< 1 \text{ us}= 0.20 \text{ Mas}ApplicationsApplicationsI_{1}Response time (2) 50 % of I_{\mu_{n}}< 1 \text{ us}= 0.55 \text{ mC}\odot 0.55 \text{ mB}\odot 0.55 \text{ mB}I_{1}Amblent operating temperature-25 \text{ mB}10 \text{ mB}\odot 0.55 \text{ mB}\odot 0.55 \text{ mB}I_{1}$	A	ccuracy - Dynamic performance data			Excellent accuracy
$ \begin{array}{c} \begin{array}{c} \text{Controls} (0, y_{1}, Y_{1}, y_{2}, y_{3}, y_{4}, y_{5}, y_{4}, y_{5}, y$	x	Accuracy @ L., T. = 25°C @ ± 15 V (± 5 %)	± 0.65	%	 Very good linearity
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		@ ± 12 15 V (± 5 %)	± 0.90	%	Low temperature drift Optimized response time
I_0 Offset ourrent (Q) I_s = 0, T_s = 25°C Typ Max No insertion losses I_0 Residual current (Q) I_s = 0, after an overload of 3 x I_m = 0.10 mA = 110 mA = High immunity to external interference I_0 Thermal drift of I_0 0°C + 70°C = 25°C + 85°C ± 0.05 = 0.30 mA Current overload capability. I_1 Thermal drift of I_0 0°C + 70°C = 25°C + 85°C ± 0.05 = 0.30 mA Current overload capability. I_2 Reaction time (Q) 10 % of I_m < \$200 ns	ε,	Linearity error	< 0.15	%	Wide frequency bandwidth
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $			Typ Max		 No insertion losses
I_{ox} Residual current "@ $I_{a} = 0$, after an overload of 2 x I_{bx} Thermal diff of $I_{a} = 0^{\circ}C_{-} + 70^{\circ}C_{-} + 25^{\circ}C_{-} + 85^{\circ}C_{-} = 0.05 = 0.30 \text{ mA}$ t Reaction time @ 10 % of I_{a} • Current overload capability. t Response time "@ 90 % of I_{a} < 300 ms	l.,	Offset current @ I, = 0, T, = 25°C	± 0.10	mA	 High immunity to external
I_{arr} Thermal dift of I_{arr} 0°C+70°C $\ge 0.05 \ge 0.30$ mA + Current overload capability. $-25^{\circ}C+85^{\circ}C$ $\ge 0.05 \ge 0.30$ mA + Current overload capability. I_{arr} Response time $\frac{1}{2}$ (§ 50 % of I_{arr} < 500	l _{ee}	Residual current ¹⁰ @ I, = 0, after an overload of 3 x I,	± 0.15	mA	Interference
t Reaction time (2) 10 % of I_m < 500	G1	Thermal drift of I ₀ 0°C + 70°C - 25°C + 85°C	± 0.05 ± 0.25 ± 0.05 ± 0.30	mA mA	 Current overload capability.
Image: Standards <1	t	Beaction time @ 10 % of I	< 500	05	Applications
divid divid accurately followed ************************************	ř.	Response time * @ 90 % of L	< 1	μs.	
f Frequency bandwidth (~ 1 dB) DC 200 kHz motor drives General data	ál∕at	di/dt accurately followed	> 200	A/µs	 AC variable speed drives and servo
General data • state converters for DC motor drives T_A Ambient operating temperature • 25 - + 85 • C T_A Ambient storage temperature • 25 - + 85 • C T_A Ambient storage temperature • 00 - + 90 - * 00 • Satisfied Mode Power Supplies R_B Secondary coll resistance @ T_A = 70°C 145 0 Mass T_A = 85°C 150 0 • Fower supplies for weiding applications. Mass 18 9 • Satisfied Mode Power Supplies for weiding applications. • Satisfied Mode Power Supplies for weiding applications. Notes 18 9 • Noter supplies for weiding applications. • Fower supplies for weiding applications. Notes 18 9 • 90178 : 1997 • 90178 : 1997 Notes • 18 • 90178 : 1997 • 90178 : 1997 • 90178 : 1997 Notes • 18 • 90178 : 1997 • 90178 : 1997 • 90178 : 1997 Notes • 190 Applications on the operation of the magnetic circuit • 9000000000000000000000000000000000000	f	Frequency bandwidth (- 1 dB)	DC 200	kHz	motor drives
General data -25 + 85 C T _A Ambient operating temperature -25 + 85 C T _A Ambient storage temperature -25 + 85 C T _A Ambient storage temperature -20 + 90 C R _A Secondary coll resistance (training temperature) -25 + 85 C T _A -70°C 145 C Mass T _A = 85°C 150 C Mass 18 0 Fouer supplies for weiding applications. Notes: IN 50178 : 1997 Power supplies for weiding applications. Notes: ************************************	0	anoral data			Battery supplied applications
T_A Ambient storage temperature -12 - * so * (UFS) T_A Ambient storage temperature -40 - * 90 * C (UFS) R_B Secondary coll resistance @ T_A = 70°C 145 C (SMPS) Mass 18 0 Fower supplies for weiding applications. Mass 18 0 applications. Notes: 18 0 applications.		deblect constant terrorities			 Uninterruptible Power Supplies
T _A Forbeth statute to the statute	÷	Ambient operating temperature	-40 +90	-0	(UPS)
T _A = 85°C 150 0 (SMME) m Mass 18 9 Standards EN 50178 : 1997 applications. Notes 1 1 Notes 1 1 Notes 1 1 Index to the coercive field of the magnetic circuit Notes 1 1 Index to the coercive field of the magnetic circuit Notes 1 1 Index to the coercive field of the magnetic circuit Notes 060000/11 Index to the coercive field of the magnetic circuit Notes 060000/11	R.	Secondary coll resistance @ T. = 70°C	145	ă	 Switched Mode Power Supplies
m Mass 18 0 Foreign duplications. Standards EN 50178 : 1997 applications. Notes : " Result of the coercive field of the magnetic circuit > With a divid of 100 Alus. Utbroketering to any street field of the magnetic circuit		T, = 85°C	150	a	(SMPS) • Enver supplies for welding
Standards EN 50178 : 1997 Notes : 1 Result of the coercive field of the magnetic circuit 3 With a direct of 100 A/µs. 660003/11 Lible convertinging carry extendification of the response field processes for a second proceses for a second processes for a second processes for a second p	m	Mass	18	9	applications.
Notes : 11 Result of the coercive field of the magnetic circuit 8 Notes : 11 Result of 100 Alps. 660000/11 LEM : Comparison the process of the coercive intercoded in scenario powerses. 660000/11 LEM : Comparison to the coercive intercoded in scenario powerses. 660000/11		Standards	EN 50178 : 19	97	
Life many schedights carry schedifications as between schedure we then with supervise ratio. 0800003/11 LFM water large V2	Notes	: $^{\rm H}$ Result of the coercive field of the magnetic circul $^{\rm H}$ With a didt of 100 Alps.	t		
LEM www.lem.com		Little receives the right to carry out an difficultons on the	tanadadera, in externali	ngersve finer	osocia/11
	LEM				www.lern.com



Mechanical characteristics

· General tolerance

· Primary through-hole

· Fastening & connection of secondary

Recommended PC8 hole

3 pins 0.63 x 0.56 mm 0.9 mm

± 0.2 mm

12.7 x 7 mm

Remarks

- I is positive when I flows in the direction of the arrow.
 Temperature of the primary conductor should not exceed
- 90°C. · Dynamic performances (di/dt and response time) are best
- with a single bar completely filling the primary hole. In order to achieve the best magnetic coupling, the primary
- windings have to be wound over the top edge of the device.

LEM reserves the right is carry estimatification on its transdocers, in order to improve them, without previous notice.

050203/11 page 37 www.lem.com

LEM



1N4001/L - 1N4007/L

1.0A RECTIFIER

Features

- Diffused Junction
- High Current Capability and Low Forward Voltage Drop Surge Overload Rating to 30A Peak Low Reverse Leakage Current ٠
- •
- ٠ Plastic Material: UL Flammability ٠
- Classification Rating 94V-0

Mechanical Data

- Case: Molded Plastic
- Terminals: Plated Leads Solderable per MIL-STD-202, Method 208 •

- Polarity: Cathode Band
 Weight: DO-41 0.30 grams (approx) A-405 0.20 grams (approx)
 Mounting Position: Any
 Marking: Type Number



	DO-41	Plastic	A-405			
Dim	Min Max		Min	Max		
Α	25.40	_	25.40	_		
В	4.06	5.21	4.10	5.20		
C	0.71	0.864	0.53	0.64		
D	2.00	2.72	2.00	2.70		
All Dimensions in mm						

"L" Suffix Designates A-405 Package No Suffix Designates DO-41 Package

Maximum Ratings and Electrical Characteristics @ T_A = 25°C unless otherwise specified

Single phase, half wave, 60Hz, resistive or inductive load. For capacitive load, derate current by 20%.

Characteristic	Symbol	1N 4001/L	1N 4002/L	1N 4003/L	1N 4004/L	1N 4005/L	1N 4006/L	1N 4007/L	Unit
Peak Repetitive Reverse Voltage Working Peak Reverse Voltage DC Blocking Voltage	V _{RRM} V _{RWM} V _R	50	100	200	400	600	800	1000	٧
RMS Reverse Voltage	VR(RMS)	35	70	140	280	420	560	700	V
Average Rectified Output Current (Note 1) @ T _A = 75°C	lo				1.0				A
Non-Repetitive Peak Forward Surge Current 8.3ms single half sine-wave superimposed on rated load (JEDEC Method)		30					A		
Forward Voltage @ IF = 1.0A	VEM				1.0				V
Peak Reverse Current @ T _A = 25°C at Rated DC Blocking Voltage @ T _A = 100°C		5.0 50					μA		
Typical Junction Capacitance (Note 2)		15 8				рF			
Typical Thermal Resistance Junction to Ambient		100						K/W	
Maximum DC Blocking Voltage Temperature		+150					°C		
Operating and Storage Temperature Range (Note 3)	Tj. TSTG			-	65 to +17	5			°C

Notes: 1. Leads maintained at ambient temperature at a distance of 9.5mm from the case. 2. Measured at 1. MHz and applied reverse voltage of 4.0V DC. 3. JEDEC Value

DS28002 Rev. E-2

1 of 2

1N4001/L-1N4007/L



7. SILICON BRIDGE RECTIFIERS

carries U/L r apacitance Lo dual Technico	ecognition 94V-O. bad Derate Current by al Data Sheets Giving	20%. Rating				OPERA	TIONAL T	EMPERATURE RANG	i€ =55° ⊮.
haracteristic	Curves are Available					S	TORAGE 1	EMPERATURE RANG	E -55° to
уура Тура	PRV	Max A Cur @ Hal Res 60	vq Rect rent 1-Wave Load IHz	Max Fwd Peak Surge Current 1 ~ 60Hz Superimposed	Max Reverse Current © PRV Voltage © 25°C TA	Mex Vol 25°	Fwd Itage P C T A	Avalanche Breakdown Voltage	Max Re Recuver Time
		10 0	P TA	IFM (Surge)	IR	IFM	VFM	MIN MAX	111
	VPK	AAV	*c	АРК	Adc	APK	VPK	VPK	ns
7-1	1 TO 2 AMPE	RE/WB	DUTLINE	E					
WB100 WB101 WB102 WB104 WB106 WB108 WB108	50 100 200 400 600 800 1000	1.0 1.0 1.0 1.0 1.0 1.0 1.0	25 25 25 25 25 25 25 25	50 50 50 50 50 50 50 50	10 10 10 10 10 10 10	1.0 1.0 1.0 1.0 1.0 1.0 1.0	1.0 1.0 1.0 1.0 1.0 1.0		
W8150 W8151 W8152 W8154 W8156 W8158 W8158 W8158	50 100 200 400 600 800 1000	1.5 1.5 1.5 1.5 1.5 1.5 1.5	25 25 25 25 25 25 25	50 50 50 50 50 50 50 50	10 10 10 10 10 10	1.0 1.0 1.0 1.0 1.0 1.0 1.0	1.0 1.0 1.0 1.0 1.0 1.0 1.0		
WB200 WB201 WB202 WB204 WB206 WB208 WB208 WB210	50 100 200 400 600 800 1000	2.0 2.0 2.0 2.0 2.0 2.0 2.0 2.0	25 25 25 25 25 25 25 25	50 50 50 50 50 50 50	10 10 10 10 10 10	1.0 1.0 1.0 1.0 1.0 1.0 1.0	1.0 1.0 1.0 1.0 1.0 1.0 1.0		4
7.2	1 TO 2 AMPER	E/SB200	OUTLIN	E	1				
3N246 3N247 3N248 3N249 3N250 3N250 3N251 3N252	50 100 200 400 600 800 1000	1.0 1.0 1.0 1.0 1.0 1.0 1.0	75 75 75 75 75 75 75 75	30 30 30 30 30 30 30 30	10 10 10 10 10 10	3.14 3.14 3.14 3.14 3.14 3.14 3.14 3.14	1.3 1.3 1.3 1.3 1.3 1.3 1.3		
3N253 3N254 3N255 3N256 3N256 3N257 3N258 3N259	50 100 200 400 600 800 1000	2.0 2.0 2.0 2.0 2.0 2.0 2.0 2.0	55 55 55 55 55 55 55	60 60 60 60 60 60 60	10 10 10 10 10 10	3.14 3.14 3.14 3.14 3.14 3.14 3.14 3.14	1.1 1.1 1.1 1.1 1.1 1.1 1.1 1.1	Ó	
SB200 SB201 SB202 SB204 SB206 SB208 SB210	50 100 200 400 600 800 1000	2.0 2.0 2.0 2.0 2.0 2.0 2.0 2.0	50 50 50 50 50 50 50 50 50	60 60 60 60 60 60 60	10 10 10 10 10 10	1.0 1.0 1.0 1.0 1.0 1.0 1.0	1.0 1.0 1.0 1.0 1.0 1.0 1.0	č	
7-3	4 AMPERE/SB40	00 OUTL	INE					chee	
SB400 SB401 SB402 SB404 SB406 SB408 SB408	50 100 200 400 600 800	4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	50 50 50 50 50 50 50	200 200 200 200 200 200	10 10 10 10 10	3.0 3.0 3.0 3.0 3.0 3.0 3.0	1.2 1.2 1.2 1.2 1.2 1.2 1.2	ð a	

FAIRCHILD SEMICONDUCTOR

LM78XX/LM78XXA 3-Terminal 1A Positive Voltage Regulator

Features

- Output Current up to 1A
- Output Voltages of 5, 6, 8, 9, 10, 12, 15, 18, 24
- Thermal Overload Protection
- Short Circuit Protection
- Output Transistor Safe Operating Area Protection

May 2006

LM78XX/LM78XXA 3-Terminal 1A Positive Voltage Regulator

Ordering Information

Product Number	Output Voltage Tolerance	Package	Operating Temperature
LM7805CT	±4%	TO-220	-40°C to +125°C
LM7806CT	1		
LM7808CT			
LM7809CT	1		
LM7810CT			
LM7812CT	1		
LM7815CT	1		
LM7818CT			
LM7824CT			
LM7805ACT	±2%		0°C to +125°C
LM7806ACT			
LM7808ACT			
LM7809ACT			
LM7810ACT			
LM7812ACT			
LM7815ACT			
LM7818ACT	1		
LM7824ACT			

General Description

The LM78XX series of three terminal positive regulators are available in the TO-220 package and with several fixed output voltages, making them useful in a wide

range of applications. Each type employs internal current

limiting, thermal shut down and safe operating area protection, making it essentially indestructible. If adequate heat sinking is provided, they can deliver over 1A output

current. Although designed primarily as fixed voltage regulators, these devices can be used with external components to obtain adjustable voltages and currents.

©2006 Fairchild Semiconductor Corporation LM78XX/LM78XXA Rev. 1.0.1

1

www.fairchildsemi.com

September 2001

National Semiconductor

LM79XX Series 3-Terminal Negative Regulators

General Description

The LM79XX series of 3-terminal regulators is available with fixed output voltages of -5V, -12V, and -15V. These devices need only one external component—a compensation capacitor at the output. The LM79XX series is packaged in the TO-220 power package and is capable of supplying 1.5A of output current.

These regulators employ internal current limiting safe area protection and thermal shutdown for protection against virtually all overload conditions.

Low ground pin current of the LM79XX series allows output voltage to be easily boosted above the preset value with a

Connection Diagrams



Order Number LM7905CT, LM7912CT or LM7915CT See NS Package Number TO3B

resistor divider. The low quiescent current drain of these devices with a specified maximum change with line and load ensures good regulation in the voltage boosted mode. For applications requiring other voltages, see LM137 datasheet

Features

- Thermal, short circuit and safe area protection
- High ripple rejection
 1.5A output current
- 4% tolerance on preset output voltage

Typical Applications



*Required if regulator is separated from filter capacitor by more than 3". For value given, capacitor must be solid tantalum. 25µF aluminum electrolytic may be substituted. †Required for stability. For value given, capacitor must be solid tantalum. 25µF aluminum electrolytic may be substituted. Values given may be increased without limit.

For output capacitance in excess of $100\mu F,$ a high current diode from input to output (1N4001, etc.) will protect the regulator from momentary input shorts.

© 2001 National Semiconductor Corporation DS007340

www.national.com



FAIRCHILD

SEMICONDUCTOR

Data Sheet

RURG5060

50A, 600V Ultrafast Diode

The RURG5060 is an ultrafast diode with soft recovery characteristics ($t_{\rm rr}$ < 65ns). It has low forward voltage drop and is of silicon nitride passivated ion-implanted epitaxial planar construction.

This device is intended for use as a freewheeling/clamping diode and rectifier in a variety of switching power supplies and other power switching applications. Its low stored charge and ultrafast recovery with soft recovery characteristic minimizes ringing and electrical noise in many power switching circuits, thus reducing power loss in the switching transistors.

Formerly developmental type TA09909.

Ordering Information

PART NUMBER	PACKAGE	BRAND			
RURG5060	TO-247	RURG5060			
NOTE: When ordering, use the entire part number.					

Symbol



January 2002

Features

- Avalanche Energy Rated
- Planar Construction

Applications

- Switching Power Supplies
- Power Switching Circuits
- General Purpose

Packaging

JEDEC STYLE 2 LEAD TO-247



Absolute Maximum Ratings T_C = 25°C, Unless Otherwise Specified

	RURG5060	UNITS
Peak Repetitive Reverse Voltage	600	V
Working Peak Reverse Voltage	600	V
DC Blocking Voltage	600	V
Average Rectified Forward Current	50	А
Repetitive Peak Surge CurrentIFRM (Square Wave, 20kHz)	100	А
Nonrepetitive Peak Surge CurrentIFSM (Halfwave, 1 Phase, 60Hz)	500	А
Maximum Power Dissipation	150	W
Avalanche Energy (See Figures 7 and 8) E _{AVL}	40	mJ
Operating and Storage Temperature	-65 to 175	°C

©2002 Fairchild Semiconductor Corporation

RURG5060 Rev. B

RURG5060

Electrical Specifications T_C = 25°C, Unless Otherwise Specified

SYMBOL	TEST CONDITION	MIN	ТҮР	MAX	UNITS
VF	I _F = 50A	-	-	1.6	V
	I _F = 50A, T _C = 150 ^o C	-	-	1.4	V
IR	V _R = 600V	-	-	250	μA
	$V_{R} = 600V, T_{C} = 150^{\circ}C$	-	-	1.5	mA
t _{rr}	$I_F = 1A$, $dI_F/dt = 100A/\mu s$	-	-	65	ns
	$I_F = 50A$, $dI_F/dt = 100A/\mu s$	-	-	75	ns
ta	$I_{F} = 50A, dI_{F}/dt = 100A/\mu s$	-	30	-	ns
tb	$I_F = 50A$, $dI_F/dt = 100A/\mu s$	-	20	-	ns
R _{0JC}		-	-	1	°C/W

DEFINITIONS

 $V_{\mbox{F}}$ = Instantaneous forward voltage (pw = 300 $\mu s,$ D = 2%).

I_R = Instantaneous reverse current.

 t_{rr} = Reverse recovery time at dI_F/dt = 100A/µs (See Figure 6), summation of $t_a + t_b$.

 t_a = Time to reach peak reverse current at dIp/dt = 100A/µs (See Figure 6).

tb = Time from peak IRM to projected zero crossing of IRM based on a straight line from peak IRM through 25% of IRM (See Figure 6).

 $R_{\theta JC}$ = Thermal resistance junction to case.

pw = Pulse width.

D = Duty cycle.

Typical Performance Curves









©2002 Fairchild Semiconductor Corporation

RURG5060 Rev. B

Single Phase Rectifier Bridge

V_{RSM}	VRRM	Туре
V	v	
800	800	VBO 30-08NO7
1200	1200	VBO 30-12NO7
1400	1400	VBO 30-14NO7
1600	1600	VBO 30-16NO7
1800	1800	VBO 30-18NO7*

* delivery time on request

Symbol	Conditions		Maximum R	atings
I _{dAVM}	T _c = 85°C, module	1	35	A
IFSM	T _{vJ} = 45°C;	t = 10 ms (50 Hz), sine	400	A
	$V_{R} = 0$	t = 8.3 ms (60 Hz), sine	440	A
	$T_{VJ} = T_{VJM}$	t = 10 ms (50 Hz), sine	360	Α
	$V_{R} = 0$	t = 8.3 ms (60 Hz), sine	400	A
l²t	$T_{VJ} = 45^{\circ}C$	t = 10 ms (50 Hz), sine	800	A ² s
	$V_{R} = 0$	t = 8.3 ms (60 Hz), sine	810	A²s
	$T_{VJ} = T_{VJM}$	t = 10 ms (50 Hz), sine	650	A ² s
	$V_R = 0$	t = 8.3 ms (60 Hz), sine	670	A²s
T _{VJ}			-40+150	°C
T _{VJM}			150	°C
T _{stg}			-40+150	°C
VISOL	50/60 Hz, RMS	t = 1 min	2500	V~
	l _{isoL} ≤ 1 mA	t = 1 s	3000	V~
Md	Mounting torque (M4)	1.5 ±15%	Nm
			13 ±15%	lb.in.
	Terminal connection	on torque (M4)	1.5 ±15%	Nm
			13 ±15%	lb.in.
Weight	typ.		135	g

Symbol	Conditions		Character	ristic	Values
I _R	$V_{B} = V_{BBM};$ $V_{B} = V_{BBM};$	$T_{VJ} = 25^{\circ}C$ $T_{VJ} = T_{VJM}$	≤ ≤	0.3 5.0	mA mA
V _F	I _F = 150 A;	$T_{vJ} = 25^{\circ}C$	\leq	2.2	V
V _{to} r _t	For power-loss o T _{VJ} = T _{VJM}	alculations only		0.85 12	V mΩ
R _{thJC} R _{thJK}	per diode; DC cu per module per diode; DC cu per module	irrent irrent		2.8 0.7 3.4 0.85	K/W K/W K/W

Data according to IEC 60747 refer to a single diode unless otherwise stated.

IXYS reserves the right to change limits, test conditions and dimensions.

© 2004 IXYS All rights reserved

VBO 30

I_{dAVM} = 35 A V_{RBM} = 1200-1800 V



Features

- Package with screw terminals
 Isolation voltage 3000 V~
 Planar passivated chips
 Blocking voltage up to 1800 V
 Low forward voltage drop
 UL registered E 72873

Applications

Supplies for DC power equipment
Input rectifiers for PWM inverter
Battery DC power supplies
Field supply for DC motors

Advantages

- Easy to mount with two screws
 Space and weight savings
 Improved temperature and power cycling

Dimensions in mm (1 mm = 0.0394")



1 - 2



SEMICONDUCTOR

Data Sheet

APRIL 2002

600V, SMPS Series N-Channel IGBT with Anti-Parallel Hyperfast Diode

This family of MOS gated high voltage switching devices combine the best features of MOSFETs and bipolar transistors. These devices have the high input impedance of a MOSFET and the low on-state conduction loss of a bipolar transistor. The much lower on-state voltage drop varies only moderately between 25°C and 150°C. The IGBT used is the development type TA49339. The diode used in anti-parallel is the development type TA49372.

These IGBT's are ideal for many high voltage switching applications operating at high frequencies where low conduction losses are essential. These devices have been optimized for high frequency switch mode power supplies.

Formerly Developmental Type TA49341.

Ordering Information

PART NUMBER	PACKAGE	BRAND		
HGTG20N60A4D	TO-247	20N60A4D		
HGT4E20N60A4DS	TO-268	20N60A4DS		
NOTE: When ordering, use the entire part number.				

Symbol



Features

- >100kHz Operation At 390V, 20A
- 200kHz Operation At 390V, 12A
- 600∨ Switching SOA Capability
- Low Conduction Loss
- Temperature Compensating SABER™ Model www.fairchildsemi.com

Packaging

JEDEC STYLE TO-247





FAIRC	HILD SEMICONDU	ICTOR IGBT PRO	DUCT IS COVER	ED BY ONE OR M	ORE OF THE FO	LLOWING U.S. PA	TENTS
4,364,073	4,417,385	4,430,792	4,443,931	4,466,176	4,516,143	4,532,534	4,587,713
4,598,461	4,605,948	4,620,211	4,631,564	4,639,754	4,639,762	4,641,162	4,644,637
4,682,195	4,684,413	4,694,313	4,717,679	4,743,952	4,783,690	4,794,432	4,801,986
4,803,533	4,809,045	4,809,047	4,810,665	4,823,176	4,837,606	4,860,080	4,883,767
4,888,627	4,890,143	4,901,127	4,904,609	4,933,740	4,963,951	4,969,027	

62002 Fairchild Semiconductor Corporation

HGTG20N60A4D, HGT4E20N60A4DS Rev. C

Absolute Maximum Ratings T_C = 25°C, Unless Otherwise Specified

	HGTG20N60A4D, HGT4E20N60A4DS	UNITS
Collector to Emitter Voltage	600	V
Collector Current Continuous		
At T _C = 25°CI _{C25}	70	A
At T _C = 110°C I _{C110}	40	A
Collector Current Pulsed (Note 1) I _{CM}	280	A
Gate to Emitter Voltage Continuous	±20	V
Gate to Emitter Voltage Pulsed	±30	V
Switching Safe Operating Area at TJ = 150°C (Figure 2) SSOA	100A at 600V	
Power Dissipation Total at $T_C = 25^{\circ}C$ P_D	290	W
Power Dissipation Derating T _C > 25°C	2.32	W/°C
Operating and Storage Junction Temperature Range	-55 to 150	°C
Maximum Lead Temperature for Soldering	260	°C

CAUTION: Stresses above those listed in "Absolute Maximum Ratings" may cause permanent damage to the device. This is a stress only rating and operation of the device at these or any other conditions above those indicated in the operational sections of this specification is not implied.

NOTE:

1. Pulse width limited by maximum junction temperature.

PARAMETER	SYMBOL	TEST CONDITIONS		MIN	TYP	MAX	UNITS
Collector to Emitter Breakdown Voltage	BV _{CES}			600	-	-	V
Collector to Emitter Leakage Current	ICES	V _{CE} = 600∨	T _J = 25°C	-	-	250	μA
			T _J = 125 ^o C	-	-	3.0	mA
Collector to Emitter Saturation Voltage	V _{CE(SAT)}	I _C = 20A, V _{GE} = 15V	T _J = 25°C	-	1.8	2.7	V
			T _J = 125°C	-	1.6	2.0	V
Gate to Emitter Threshold Voltage	VGE(TH)	I _C = 250μA, V _{CE} = 6	00V	4.5	5.5	7.0	V
Gate to Emitter Leakage Current	IGES	V _{GE} = ±20V		-	-	±250	nA
Switching SOA	SSOA	T _J = 150 ^o C, R _G = 3 L = 100μH, V _{CE} = 6	$T_J = 150^{\circ}C, R_G = 3\Omega, V_{GE} = 15V,$ L = 100µH, V _{CE} = 600V		-	-	A
Gate to Emitter Plateau Voltage	Emitter Plateau Voltage V _{GEP} I _C = 20A, V _{CE} = 300V		IV	-	8.6	-	V
On-State Gate Charge	Q _{g(ON)}	I _C = 20A, V _{CE} = 300V	V _{GE} = 15V	-	142	162	nC
			V _{GE} = 20V	-	182	210	nC
Current Turn-On Delay Time	t _{d(ON)}	IGBT and Diode at $T_J = 25^{\circ}C$,		-	15	-	ns
Current Rise Time	t _{ri}	I _{CE} = 20A, V _{CE} = 390V.		-	12	-	ns
Current Turn-Off Delay Time	td(OFF)I	V _{GE} = 15V,		-	73	-	ns
Current Fall Time	t _{fi}	$R_G = 312$ L = 500µH, Test Circuit Figure 24		-	32	-	ns
Turn-On Energy (Note 3)	E _{ON1}			-	105	-	μ
Turn-On Energy (Note 3)	E _{ON2}			-	280	350	μ
Turn-Off Energy (Note 2)	EOFF			-	150	200	μ
Current Turn-On Delay Time	t _{d(ON)}	IGBT and Diode at	Гј = 125°С,	-	15	21	ns
Current Rise Time	t _{ri}	ICE = 20A,		-	13	18	ns
Current Turn-Off Delay Time	t _{d(OFF)}	$R_G = 3\Omega$	$R_G = 3\Omega$,		105	135	ns
Current Fall Time	t _{fi}	L = 500µH, Test Circuit Figure 24		-	55	73	ns
urn-On Energy (Note 3) E _{ON1}					115	-	μ
Turn-On Energy (Note 3)	E _{ON2}	E _{ON2}		-	510	600	μ
Turn-Off Energy (Note 2)	EOFF			-	330	500	μJ

©2002 Fairchild Semiconductor Corporation

HGTG20N60A4D, HGT4E20N60A4DS Rev. C

Electrical Specifications T_J = 25°C, Unless Otherwise Specified (Continued)

PARAMETER	SYMBOL	TEST CONDITIONS	MIN	TYP	MAX	UNITS
Diode Forward Voltage	V _{EC}	I _{EC} = 20A	-	2.3	-	V
Diode Reverse Recovery Time	t _{rr}	I_{EC} = 20A, dI_{EC}/dt = 200A/µs	-	35	-	ns
		I _{EC} = 1A, dI _{EC} /dt = 200A/µs	-	26	-	ns
Thermal Resistance Junction To Case	R _{ØJC}	IGBT	-	-	0.43	°C/W
		Diode	-	-	1.9	°C/W

NOTE:

 Turn-Off Energy Loss (E_{OFF}) is defined as the integral of the instantaneous power loss starting at the trailing edge of the input pulse and ending at the point where the collector current equals zero (I_{CE} = 0A). All devices were tested per JEDEC Standard No. 24-1 Method for Measurement of Power Device Turn-Off Switching Loss. This test method produces the true total Turn-Off Energy Loss.

Values for two Turn-On loss conditions are shown for the convenience of the circuit designer. E_{ON1} is the turn-on loss of the IGBT only. E_{ON2} is the turn-on loss when a typical diode is used in the test circuit and the diode is at the same T_J as the IGBT. The diode type is specified in Figure 20.

Typical Performance Curves Unless Otherwise Specified



©2002 Fairchild Semiconductor Corporation

HGTG20N60A4D, HGT4E20N60A4DS Rev. C





National Semiconductor

November 1994

LM110/LM210/LM310 Voltage Follower

General Description

The LM110 series are monolithic operational amplifiers internally connected as unity-gain non-inverting amplifiers. They use super-gain transistors in the input stage to get low bias current without sacrificing speed. Directly interchange-able with 101, 741 and 709 in voltage follower applications, these devices have internal frequency compensation and provision for offset balancing.

The LM110 series are useful in fast sample and hold circuits, active filters, or as general-purpose buffers. Further, the frequency response is sufficiently better than standard IC amplifiers that the followers can be included in the feedback loop without introducing instability. They are plug-in replacements for the LM102 series voltage followers, offering lower offset voltage, drift, bias current and noise in addition to higher speed and wider operating voltage range. The LM110 is specified over a temperature range -55°C < $T_A \leq \pm 125^\circ\text{C},$ the LM210 from $-25^\circ\text{C} \leq T_A \leq \pm 85^\circ\text{C}$ and the LM310 from 0°C \leq T_A \leq +70°C.

Features

- Input current
- Small signal bandwidth
- Slew rate
- Supply voltage range

 $\pm 5V$ to $\pm 18V$

10 nA max over temperature

20 MHz

30 V/µs



©1995 National Semiconductor Corporation TL/H/7761

	Vene	I _C (in A	I _C (in	Ver				Digikey USD Price
IGBT	(min.)	25	110	v CE (sat.)	Rise	Fall		(Each/
Code	(V)	⁰ C)	⁰ C)	V	Time	Time	Frequency	1000)
			,			$I_{\rm C} = 24 \text{ A}, T_{\rm J} = 25$		/
					I _C =24 A, T _J =25	⁰ C 140 ns		
					⁰ C	typical, 300 ns		
IXSH30N60					$30ns; T_J=125 \ ^{0}C$	max, $T_J=125 \ ^{0}C$	good up to 20	
B2D1	600	48	30	2.0	50 ns	234 ns typical	kHz	3.29
					$I_{C}=27 \text{ A}, T_{J}=25$			
					25 ns,	$I_{\rm C}$ =27A, $T_{\rm J}$ =25		
					typical $T_J = 150$	^o C 74 ns typical		
IRG4PC50					^{0}C	110 ns max, T_J	Ultrafast 8-60	
UD	600	55	27	2.00	27 ns	$=150 {}^{0}\text{C} 130 \text{ns}$	kHz	7.01
					$I_{C}=33 \text{ A}, T_{J}=25$			
					°C	$I_{\rm C} = 33$ A, $T_{\rm J} = 25$		
					26 ns typical 36	°C 43 ns typical		
					ns max, $1_{J}=125$	56ns max,		
					26 mg	$I_{\rm J} = 125 ^{\circ}{\rm C} 50 \rm ns$	WADD 60	
INGESUD 60PD	600	75	42	22	20 IIS typical 36 ns max	typical 05 lis	150 kHz	7 54
001 D	000	15	72	2.2	$I_{c}=33 \text{ A } T_{r}=25$	$L_{a} = 33 \text{ A} \text{ T}_{a} = 25$	150 KHZ	7.54
					${}^{0}C$ rise time=10	${}^{0}C$ fall time=11		
					ns typical 15 ns	ns typical 15 ns.		
					max, $T_{I}=125 \ ^{0}C$	$T_1=125$ °C fall		
IRGP50B60					rise time 13 ns	time= 15 ns	WARP 60-	
PD1	600	75	45	2.35	typical 20 ns max	typical 20ns max	150 kHz	6.20
						$I_{\rm C} = 27 \text{ A} T_{\rm J} = 25$		
					$I_{\rm C}$ =27 A T _J =25	^o C fall time=57		
					^o C rise time=33	ns typical 86 ns		
					ns typical,	typical, $T_J=125$		
IRG4PC50				• •	$T_J = 150$ °C rise	^o C fall time 62 ns	WARP 60-	
W-ND	600	55	27	2.3	time 43ns typical	typical	150 kHz	3.91
				$1_{J}=25$	I	$I_{\rm C} = 20 \text{ A} T_{\rm J} = 25$	I-20 A	
				27 V	$1_{\rm C}=20 \text{ A } 1_{\rm J}=25$	c fall time=32	10-20 A max.	
				$\Delta . / V$	c rise time=12	T=125 0 C fall	at 100 KHZ	
				пал, Т.=125	$T_{r}=125$ ⁰ C rise	1J = 125 C fall time 55 ns	12 A may at	
HGTG20N60				${}^{0}C 20$	time=13 ns	typical 73 ns	200 kHz	
A4D	600	70	40	V max	typical 18 ns max	max	switching	2.93

APPENDIX C: IGBT And Ultrafast Diode Selection Tables

 Table C.1 IGBT selection table.

						Digikey
		I ₀ (rec.)				USD
	V _{RRM}	Max (A) at 125 9 C	T			Price
Diada Cada		$I_J = 125^{\circ}C$	I _{FSM} Max (A)	V May (V)	t (ns)	(Eacn/ 1000)
Dioue Coue	(*)	(nearry)	Max (A)	V_{FM} Wax (V)	$\iota_{\rm rr}$ (118)	1000)
				$1_{\rm F}$ = 13 Å 1=130 °C 1 2V max T=25 °C	$I_{\rm E}=1.0$ A di/dt=50A/us	
MUR3060PT	600	30	150	1.5 V max	60 ns max	1.40
				$I_{\rm F} = 15 \text{ A T} = 150 ^{0}\text{C}$		
				$1.4 \text{ V} \text{ max}, \text{ T}=25 ^{\circ}\text{C}$	I _F =1.0 A di/dt=50A/us	
MUR3060WT	600	30	150	1.7 V max	60 ns max	1.55
					$I_F=1.0 A$	
					di/dt=100A/us 55 ns	
				$I_F=30 \text{ A} T=25 \text{ °C} 1.5$	\max_{i} I _F =30 A	
	600	20	225	$v \max_{1=150}^{150} C$	d1/dt=100A/us 60 ns	1 2 5
RURG3000	000	30	323	1.5 v max 2.4 (I_=30A T=25 ^{0}C	IIIdX	1.55
				2.4 (IF 50A 1 25 C 2 1 V typical 2 4 V	$35ns$ ($I_{\rm F}=30A$ T=25 ^{0}C	
				max. $T=125$ °C 1.7 V	36 ns typical 110 ns	
ISL9R3060P2	600	30	325	typical 2.1 V max)	max)	1.58
					$I_F = 1A dI_F/dt = 100A/us$	
				$I_F = 30 \text{ A T} = 25 {}^{0}\text{C} 1.5$	55 ns max, I _F =30A	
				V max, T=150 °C 1.3	dIF/dt=100A/us 60 ns	
RURP3060	600	30	325	V max	max	1.11
					$I_F = 1A dI_F/dt = 200A/us$	
				$I_F = 30 \text{ A } I = 25 \text{ °C } 2.1$	40 ns max, $I_F=30A$	
RHRC3060	600	30	325	$V \max$, 1-150 C 1.7 V max	$u_{\rm F}/u_{\rm C}=200{\rm A}/u_{\rm S}$ 43 max	1.45
KIIKO5000	000	50	525	$2 1 (I_{\rm F}=30A T=25 {}^{0}C)$	Шах	1.75
				2.0 V typical. 2.6 V	40 ns (T=25 0 C I _F =30	
				max and $I_F=30A$,	A, trr= 31 ns typical	
			325	T=150 °C 1.34 V	and T=125 ^{0}C I _F =30 A,	
30EPH06	600	30	A(60Hz)	typical 1.75 V max)	trr=77 ns typical	1.34
DSEI30-06A	600	37		$1.4 (I_F = 37 \text{ A T} = 25)$	$I_F=1 \text{ A}, \text{di/dt}=100 \text{ A/us},$	
				^o C 1.6V max, T=150	$V_{R}=30 V, T_{VJ}=25 °C$	
			2(0	°C 1.4V max)	typical 35ns, Max 50	1.24
	600	220	260	1 4 (I - 27 A T - 25 °C)	ns	1.34
DSER00-00A	000	2x30	200	$1.4 (I_F - 37 A T - 23 C)$ 1 6 V max T=150 °C	$V_{\rm F}=1$ A, u/u(=100A/us V_{\rm F}=30 V T_{\rm H}=25 °C	
				1.0 v max, 1 100 C 1 4 V max)	v_R 50 v_1 v_2 25 C	
				i.i.v inux)	ns	2.99
				Version A 1.25		
				$(I_F=30 \text{ A T}=25 \ ^{0}\text{C}$		
				$1.6V \text{ max}, \text{T}=150 \ ^{0}\text{C}$		
				1.25V max), Version		
				$B 1.56 (I_F=30 T=25)$	$I_F=1$ A, di/dt=100A/us	
DCED20				$^{\circ}C 2.51 \text{ V max},$	$V_{R}=30 V, T_{VJ}=25 °C$	
	600	20	250	1=150°C 1.56 V	Version R 35 ns max,	1.60
UUA/D/DK	000	50	230	max)	version D 30 hs max	1.00

 Table C.2 Ultrafast diode selection table.

APPENDIX D: Classical UPF Rectifier And Proposed Rectifier's Waveforms Obtained Via Simulations

D.1 Waveforms of Classical Unity Power Factor Rectifier And Proposed Rectifier With 3 kW Purely Resistive Load

D.1.1 Waveforms of Classical Unity Power Factor Rectifier With 3 kW Purely Resistive Load



Figure D.1 Classical unity power factor rectifier's input power and power factor at purely resistive 3 kW load, (a) its input power, (b) its power factor graph.



Figure D.2 (a) FFT waveform of classical unity power factor rectifier's input current at 3 kW purely resistive load at steady state, (b) its zoomed version.



Figure D.3 THD of classical UPF rectifier's input current at 3 kW purely resistive load at steady state.



Figure D.4 (a) IGBT's current and voltage waveform of classical UPF rectifier with 3 kW purely resistive load at steady state, (b) their zoomed version.


Figure D.5 (a) One of the bridge diodes, D1's current and voltage waveform of classical UPF rectifier with 3 kW purely resistive load at steady state, (b) their zoomed version.



Figure D.6 (a) Boost diode, D5's current and voltage waveform of classical UPF rectifier with 3 kW purely resistive load at steady state, (b) their zoomed version.



D.1.2 Waveforms of Proposed Rectifier With 3 kW Purely Resistive Load

Figure D.7 Proposed rectifier's input power and power factor graph at purely resistive 3 kW load, (a) its input power, (b) its power factor graph.

Time (ms)



Figure D.8 (a) FFT waveform of proposed rectifier's input current at 3 kW purely resistive load at steady state, (b) its zoomed version.



Figure D.9 THD of proposed rectifier's input current at 3 kW purely resistive load at steady state.



Figure D.10 (a) IGBT's current and voltage waveform of proposed rectifier with 3 kW purely resistive load at steady state, (b) their zoomed versions.



Figure D.11 (a) One of the boost diodes, D3's current and voltage waveform of proposed UPF rectifier with 3 kW purely resistive load at steady state, (b) their zoomed versions.



Figure D.12 (a) One of the bridge diodes, D1's current and voltage waveform of proposed UPF rectifier with 3 kW purely resistive load at steady state, (b) their zoomed versions.



Figure D.13 (a) One of the bridge diodes, D7's current and voltage waveform of proposed UPF rectifier with 3 kW purely resistive load at steady state, (b) their zoomed versions.

D.2 Waveforms of Classical Unity Power Factor Rectifier And Proposed Rectifier With 3 kW Inductive-Resistive Load

D.2.1 Waveforms of Classical Unity Power Factor Rectifier With 3 kW Inductive-Resistive Load

D.2.1.1 Waveforms of Classical Unity Power Factor Rectifier With 3 kW Inductive-Resistive Load (100 uH)



Figure D.14 Classical unity power factor rectifier's input power and power factor at inductive-resistive 3 kW load (100 uH), (a) its input power, (b) its power factor.



Figure D.15 (a) FFT waveform of classical unity power factor rectifier's input current at 3 kW inductive-resistive load (100 uH) at steady state, (b) its zoomed version.



Figure D.16 THD of classical UPF rectifier's input current at 3 kW inductiveresistive load (100 uH) at steady state.



Figure D.17 (a) IGBT's current and voltage waveform of classical UPF rectifier with 3 kW inductive-resistive load (100 uH) at steady state, (b) their zoomed version.



Figure D.18 (a) One of the bridge diodes, D1's current and voltage waveform of classical UPF rectifier with 3 kW inductive-resistive load (100 uH) at steady state, (b) their zoomed version.



Figure D.19 (a) Boost diode, D5's current and voltage waveform of classical UPF rectifier with 3 kW inductive-resistive load (100 uH) at steady state, (b) their zoomed version.

D.2.1.2 Waveforms of Classical Unity Power Factor Rectifier With 3 kW Inductive-Resistive Load (10 uH)



Figure D.20 Classical unity power factor rectifier's input power and power factor at inductive-resistive 3 kW load (10 uH), (a) its input power, (b) its power factor.



Figure D.21 (a) FFT waveform of classical unity power factor rectifier's input current at 3 kW inductive-resistive load (10 uH) at steady state, (b) its zoomed version.



Figure D.22 THD of classical UPF rectifier's input current at 3 kW inductiveresistive load (10 uH) at steady state.

D.2.2 Waveforms of Proposed Rectifier With 3 kW Inductive-Resistive Load

D.2.2.1 Waveforms of Proposed Rectifier With 3 kW Inductive-Resistive Load (100 uH)



Figure D.23 Proposed rectifier's input power and power factor at inductive-resistive 3 kW load (100 uH), (a) its input power, (b) its power factor.



Figure D.24 (a) FFT waveform of proposed rectifier's input current at 3 kW inductive-resistive load (100 uH) at steady state, (b) its zoomed version.



Figure D.25 THD of proposed rectifier's input current at 3 kW inductive-resistive load (100 uH) at steady state.











Figure D.27 (a) One of the boost diodes, D3's current and voltage waveform of proposed UPF rectifier with 3 kW inductive-resistive load (100 uH) at steady state, (b) their zoomed versions.



Figure D.28 (a) One of the bridge diodes, D1's current and voltage waveform of proposed UPF rectifier with 3 kW inductive-resistive load (100 uH) at steady state, (b) their zoomed versions.



Figure D.29 One of the bridge diodes, D7's current and voltage waveform of proposed UPF rectifier with 3 kW inductive-resistive load (100 uH) at steady state.

D.2.2.2 Waveforms of Proposed Rectifier With 3 kW Inductive-Resistive Load (10 uH)



Figure D.30 Proposed rectifier's input power and power factor at inductive-resistive 3 kW load (10 uH), (a) its input power, (b) its power factor.



Figure D.31 (a) FFT waveform of proposed rectifier's input current at 3 kW inductive-resistive load (10 uH) at steady state, (b) its zoomed version.



Figure D.32 THD of proposed rectifier's input current at 3 kW inductive-resistive load (10 uH) at steady state.

D.3 Waveforms of Classical Unity Power Factor Rectifier With 3 kW Inductive-Resistive Load By Taking Reference Voltage From A Transformer Connected To Mains Supply



Figure D.33 The classical unity power factor rectifier's input power and power factor with 3 kW purely resistive load by taking reference voltage from a transformer connected to mains supply, (a) the rectifier's input power (b) its power factor.



Figure D.34 The classical unity power factor rectifier's input current and voltage waveform, and its output voltage waveform with 3 kW purely resistive load by taking reference voltage from a transformer connected to mains supply, (a) its input voltage and current waveform, (b) its output voltage.



Figure D.35 The classical unity power factor rectifier's input current FFT and THD graph obtained with 3 kW purely resistive load by taking reference voltage from a transformer connected to mains supply, (a) its FFT waveform (b) its THD graph.





Figure E.1 Classical unity power factor rectifier and proposed rectifier's printed circuit board.



Figure E.2 Classical unity power factor rectifier and proposed rectifier's printed circuit board's top layer.



Figure E.3 Classical unity power factor rectifier and proposed rectifier's printed circuit board's bottom layer.